EXPLORING PROFESSIONAL IDENTITY DEVELOPMENT IN HIGHER EDUCATION: A SOCIAL IDENTITY APPROACH

by

Bahijah Abas
Stirling Management School
University Of Stirling

THESIS SUBMITTED TO THE UNIVERSITY OF STIRLING FOR THE DEGREE OF DOCTOR OF PHILOSOPHY

March 2018

EXPLORING PROFESSIONAL IDENTITY DEVELOPMENT IN HIGHER EDUCATION: A SOCIAL IDENTITY APPROACH

Acknowledgement

There are many people for whom I owe a great deal of gratitude for supporting me, not only in the completion of this thesis but throughout my PhD journey.

Firstly, I would like to express my deepest sense of gratitude to my principal supervisor, Dr Juliette Summers for the continuous support of my PhD study, for her patience, motivation and immense knowledge. I am fortunate to have a wonderful supervisor who helped me through this process by encouraging and challenging me.

I would also like to express my sincere gratitude to my additional supervisor, Prof. William Webster, for the support to make this thesis available.

Finally, I take this opportunity to express my profound gratitude to my husband, Mohd Ikram, who always assured me I was capable of anything, who understood the nights and weekends of research and writing, and who always believes in me. I also would like to thank my beloved family and family in-law for their ongoing support. To my beloved daughters, Ajda and Amni, I would like to express my thanks for being such good girls and always cheering me up.

Abstract

Higher education provides a platform for students to prepare themselves for the world of work. In an effort to engage students with their intended profession, this study focuses on how students in higher education develop their professional identity. The social identity approach comprises of Tajfel's (1974) Social Identity Theory (SIT) and Turner's (1985) Self-Categorisation Theory (SCT) were adopted as a guiding framework. A qualitative paradigm was employed as it allows the students' professional identity development to be examined in detail and in depth. An interpretive approach was adopted to understand the reality of the students' identity as it is perceived and experienced by the students. Studies have been done on students undergoing teacher education programmes. A total of 80 student teachers undergoing a bachelor of education and postgraduate teacher education programme in the biggest teacher education institution in Malaysia were recruited in this study. A focus group was conducted to obtain detailed information of group perception and evaluation. The interview data were transcribed verbatim, and subsequently analysed to facilitate the development of themes. This study provides an important insight on how the students define themselves as a member of one occupational group, and how the membership in the group gave their behaviour a distinct meaning. The findings of this study confer implication for practice with a particular reference on how to motivate students to achieve and maintain a positive social identity of their intended profession, and develop a professional identity during their socialisation process at higher education institutions.

TABLE OF CONTENT

Ackn	owledgement	i	
Abstr	act	ii	
Table	of Content	iii	
List o	of Tables	vii	
	CHAPTER ONE: INTRODUCTION		
1.1	Introduction	1	
1.2	Background of study	3	
1.3	Problem statement	5	
1.4	Social Identity Framework	8	
1.5	Aims of study	13	
1.6	Objectives of study	13	
1.7	Research questions	14	
1.8	Significance of study	14	
1.9	Context of Study	15	
1.10	Framework of study	18	
1.11	Organisation of the thesis	19	
	CHAPTER TWO :LITERATURE REVIEW		
2.1	Introduction	21	
2.2	Professional identity development in higher education	21	
2.3	The relevance of social identity approach in interpreting the		
	professional identity development among students in higher education	22	
2.4	Students' motives to choose disciplinary field and occupational group	29	
2.5	Socialisation process that helps in identification of teaching		
	aspirations and identity development	32	
	2.5.1 Introduction to students' socialisation in higher education	33	
	2.5.2 Professional socialisation approach	35	
	2.5.3 Functionalist approach of professional socialisation	36	
	2.5.4 Interpretive approach of professional socialisation	37	
2.6	Interpretive approach of socialisation: Malaysian perspectives	41	
2.7	Malaysian students and the interpretive approach to learning.	44	
2.8	Conclusion		

CHAPTER THREE : METHODOLOGY

3.1	Introd	uction	48	
3.2	Metho	odology	48	
	3.2.1	Qualitative paradigm perspective of the methodology	49	
	3.2.2	Social constructionism	51	
	3.2.3	The interpretive framework	53	
3.3	Partic	ipants and methods of data collection	54	
	3.3.1	Participants	54	
	3.3.2	Methods of data collection	58	
	3.3.3	Research tools	59	
3.4	Data a	nnalysis	63	
3.5	Ethics	s, reliability and validity in qualitative research	66	
	3.5.1	Ethics	66	
	3.5.2	Reliability and validity in qualitative research	66	
3.6	Concl	usion	67	
		CHAPTER FOUR : DISCUSSION OF FINDINGS		
4.0	Introd	uction	68	
4.1	Impor	tant themes that describe student teachers' identity towards		
	teachi	ng	69	
4.2	Stude	nts' reason for entering teacher education programme	70	
	4.2.1	Compatibility between students' characteristics and teaching		
		attributes	74	
	4.2.2	Model influences	77	
	4.2.3	Formal life experiences	84	
	4.2.4	Vocational reason	88	
	4.2.5	Other reasons to embark on teacher education programme	91	
	4.2.6	Subsection summary	93	
4.3	Student's perception of teaching during commencement in teacher			
	education programme			
	4.3.1	Evaluation of teaching profession	96	
	4.3.2	Professional relationship	102	
	4.3.3	Understanding of learning	105	
	4.3.4	Subsection summary	107	

4.4	Socialising agents who influence students' beliefs about teaching			
	4.4.1	Environment	107	
	4.4.2	Former teachers	108	
	4.4.3	School pupils	108	
	4.4.4	Subsection summary	109	
4.5	Concl	usion of chapter	111	
		CHAPTER FIVE : DISCUSSION OF FINDINGS		
5.0	Introd	luction	113	
5.1	Stude	nts' self-definition to teaching	114	
5.2	Stude	nts' commitment	117	
	5.2.1	Commitment towards prospective students	118	
		5.2.1.1 I want to educate the younger generation	118	
		5.2.1.2 I am a second parent to the students	121	
		5.2.1.3 I want to attract students' attention to learn	122	
		5.2.1.4 Education for all	127	
		5.2.1.5 Conclusion for the subsection	129	
	5.2.2	Commitment towards society	130	
	5.2.3	Commitment towards the profession	132	
	5.2.4	Challenges for students' nurturing sense of commitment	135	
	5.2.5	Conclusion of the subsection	135	
5.3	Sense	of professional identity that fostered the students' commitment	137	
	5.3.1	Professional orientation	137	
	5.3.2	Personality	139	
	5.3.3	Self-efficacy	140	
	5.3.4	Autonomy	140	
	5.3.5	Emotion	141	
	5.3.6	Conclusion of the subsection	142	
5.4	Social	l identity intersectionality	142	
5.5	Concl	usion of chapter	143	
		CHAPTER SIX : CONCLUSION		
6.0	Sumn	narising the findings	145	
6.1	Theoretical implication		149	

	6.1.1 Professional identity formation and development	149
	6.1.2 Social Identity Approach	150
	6.1.3 Interpretive approach	153
6.2	Practical implication	153
6.3	Limitation of the study	156
6.4	Future studies	157
	REFERENCES	159

LIST OF TABLES

Table 3.1:	Number of research participants	56
Table 3.2:	Distribution of participants undergoing their teaching practice	57
Table 3.3:	Schedule of focus group sessions	62
Table 3.4:	An extract of an analysis process	65
Table 4.1:	Students' motives to enter teacher education programme	72

CHAPTER ONE INTRODUCTION

1.1 Introduction

Education is of paramount importance for personal, social and economic development of the nation. The education process may occur in both formal and informal settings where the former has been systematically executed and measured in classrooms or educational institutions. Without neglecting the importance of informal education, past research suggests that participation and quality of education in primary and secondary levels have helped provide a more progressive and higher cognitive level of human capital (Hanushek, 2013; Jones and Ramchand, 2013). Equally important, further and higher education is pivotal in providing a platform to further develop skilled and qualified human capital through a training and credentialing process in order to prepare students for the world of work (Trede, Macklin and Bridges, 2012). Well trained and skilled workforce are seen as important resources that help increase productivity, boost output, and propel growth in a wider economy.

In many countries around the world, there has been an increasing number of students enrolled in higher education due to the expansion of the higher education sector ("UNESCO Institute of Statistics", n.d.). As a consequence, new higher education institutions are established, existing institutions are expanded, and a diverse range of courses are offered to accommodate the increasing number of students (Klien, 2016; Liu, Green and Pensiero, 2016). The intertwining of higher education expansion and the aim to develop skilled and qualified human capital has heightened the need for higher education institutions to focus on important management functions; planning, organizing, leading and controlling the student recruitment, selection, and development process, and facilitate students' learning experience in higher education.

In this decade, higher education applicants must choose between many diversified alternative courses while deciding what to study. It is important to realise that the applicants' motive on disciplinary choices can be treated as an indicator on how students navigate their socialization process in higher education, as motives may antecede students' learning behaviour (Byrne and Flood, 2005; Hudson et. al., 2016). Higher education literatures have signalled different motives behind applicants' disciplinary choices, with the desire to work in their intended field as a leading cause for commencement in higher education (Byrne and Flood, 2005). Hence, students' recruitment and selection process to specific academic programmes become important for higher education administrators where they have to identify the appropriate candidates to enrol in different programmes of study.

It is common practice that higher education applicants will be assessed with selected recruitment and selection tools as some established disciplines such as law and medical education have certain prerequisites for those who wish to become a future professional in that field (Kyratsis, Atun, Phillips, Tracey and George, 2016). As higher education prepares students for the world of work, the prerequisites become important to enable the selection of right candidates for professional training in higher education and equip them with the essential knowledge and skills in order to maintain the social closure of the profession, thus uphold the professional status (Saks, 2010). Equally important is the different motives to enrol in higher education which indicates that higher education administrators and academics have a crucial task to plan and execute activities or programmes in order to enhance support for the new students' socialisation process in higher education, and manage their motivation and emotions throughout their studies so that they may engage in their academic programme and future profession thereafter.

Higher education becomes a place of importance as there is a requirement to develop and control students' intellectual abilities (Jones and Ramchand, 2013). Intellectual in this sense refers to the ability of students to utilise knowledge and skills for living including engagement in their future careers. Additionally, the need to obtain higher education credentials as a means to compete in the job market has become a trend in this decade. Higher education credentials also provide graduates with the right of self-government or autonomy as it should be "gathered from higher education rather than trade school or long apprenticeship" (Friedson 1937, p. 22, citied in Monteiro, 2015).

Hence, the credentials become evidence that the learning and development process has taken place at a higher education institution (Tomlinson, 2008), and it is the graduates' responsibility on how to apply the knowledge and skills obtained in higher education into their personal and professional life.

The two factors indicated above highlighted a need for higher education administrators and academics to help students navigate themselves for their future profession as research finds that higher education provides early socialisation for the students' intended professional choice (Bakar et al, 2008; Trede et al., 2012). The socialisation process includes managing students' learning experience on professional dispositions and roles as highlighted in higher education literature. Most of the studies focus on strategies to promote learning and construct a conducive learning environment in order to enhance the students' learning experience related to their future career choices (Fokken-Bruinsma and Canrinus, 2012; Fokken-Bruinsma and Canrinus, 2014; Struyven, Jacobs and Dochy, 2013).

Taking this into consideration, there are huge responsibilities for higher education administrators and academics to nurture the capacity and capabilities of the students for their future career through learning. This includes preparing the students with relevant knowledge, skills and attitude needed to become a member of their disciplinary field-related occupation. It follows that relevant knowledge, skills and attitude related to the students' intended occupational group will assist them to maintain or enhance a positive identity of the group (Saks, 2010).

1.2 Background of Study

One of the most significant discussions on higher education functions is to cultivate students' identity (Bunce, Baird and Jones, 2017; Meca et al., 2015). As higher education is regarded as an important platform to prepare students for the world of work, the development of the identity of future workers becomes a crucial task for the institution. The initial professional identity development starts when students decide to choose a specific academic programme, hence the motive to choose a specific disciplinary field has become an antecedent in the process of students' identity

development (Gibney, Moore, Murphy and O' Sullivan, 2011). The students' motives explain their perceived readiness to become a member of one disciplinary field, and relevant occupational groups thereafter.

As has been previously reported in the literature, the students' identity is further developed through the socialisation process during the commencement in higher education (Fokken-Bruinsma & Canrinus, 2012; Fokken-Bruinsma & Canrinus, 2014; Struyven, Jacobs & Dochy, 2013). For instance, the literature emphasizes on teaching and learning of professional roles, understanding organisational culture and behaviour, and commencing professional socialisation (Atadero, Paguyo, Rambo-Hernandez and Henderson, 2018; Jackson, 2016; Neve, Lloyds and Collett, 2017). These work preparations are seen as contributing to the development of students' professional identity. The literature also reported that professional identity is created through students' beliefs on the values and norms of their anticipated profession, and these values and norms are shared with others in the same occupational group (Van Maanen and Barley, 1984). Despite the consistent findings that professional identity explains the students' perceptions of future occupational and professional attributes, the literature also suggests that it may explain student readiness to perform their future work. Thus, occupational or professional identity is important as it becomes the way students assign meaning to themselves, and it shapes students' aspirations on work attitude and behaviour (Siebert and Siebert, 2005).

Recently, many occupations need a higher education credential as an entry requirement to specialised work; hence, the institution helps to facilitate the socialisation process that has a significant impact on the students' knowledge, skills, and attitude as well as the autonomy to enable access towards one occupational group. Preparing students for their intended profession is very important to individual students and society. Society places hope on higher education institutions to prepare the students so that they will come back, contribute and maintain order in the society (Friedson, 1970). Thus, the role of higher education becomes much more challenging in providing students with the requirements needed by specific professions. This includes designing curriculum, preparing activities for teaching and learning as well as designing programmes outside the classroom to enhance the capability and capacity of students so that they can

become a competent member of their intended profession. This involves the academics, the administrators as well as the students' affair department.

1.3 Problem Statement

As highlighted in the previous subsection, literature on higher education produced consistent evidence on the role of higher education institutions in cultivating students' professional identity towards their intended profession. However, there are two neglected areas which arise from these studies; firstly, how to develop students' professional identity for individuals who are not sure or convinced with their programme of study in higher education? This turns out to be a challenging area of inquiry on how to plan and execute the professional socialisation process and develop a professional identity for students who are alienated and excluded from their counterparts who are conscious about their intended profession. Secondly, how do students feel about their intended profession? Does the status of the intended profession in the social structure influence the way students define themselves as a member of one occupational group? These two areas need to be addressed in order to understand the students' professional identity formation and development in higher education.

In a Malaysian context, higher education¹ aims to achieve national needs in critical areas that are vital to the country, such as science, technology, agricultural, management and education, in the effort to form a developed country by year 2020 (Ministry of Higher Education Malaysia, 2006). Like most countries around the world, the higher education sector in Malaysia is now experiencing a significant increase in the number of students' enrolment. It has been published in the higher education statistics that in the past 10 years, students' enrolment in bachelor degree programmes has increased significantly from 333,403 to 696,899 students in year 2016 ("Ministry of Higher Education Malaysia", n.d.-a). The increasing number of student enrolment in higher education indicates that there is a concerted effort from the government through the higher education sector to enhance the capabilities of students to compete in the job market after graduation ("Ministry of Higher Education Malaysia", n.d.-b).

¹ Higher education in Malaysia is offered by universities, colleges and polytechnics.

Even though access to higher education and the number of graduates are increasing, it is reported in prior research that there were students who enrolled in their programme of study not because of their interest in the field of study nor the desire to work in the area related to their academic programme, but the decision had been influenced or advised by other significant individuals. In Malaysian literature, concern has been raised on students' enrolment in teacher education programmes (Mohd Dahan and Ghazali, 2007; Harun, 2006; Mohd Ilias, 2008). The findings highlighted students' disengagement with teaching both while they are in the first year of the teacher education programme, and while commencing professional socialisation in the second and third year of study.

Issues on teacher education students who are not convinced with the teacher education programme has also been reported since the previous decade. Heretofore, teacher education programmes experienced difficulties in attracting students when a country is in an economic boom and jobs are plentiful (Lee, 1996; Matrak, 1995). Lee (1996) also claims that when Malaysia experienced an economic downturn, and employment opportunities become restricted due to economic conditions, the demand for teacher education rose. One factor that might attract applicants to enrol in teacher education programmes is the guaranteed job opportunity as has been implemented in Malaysia for decades; teacher education students in public higher education institutions are guaranteed placements at government funded schools after their graduation², if they satisfy the minimum requirement of teacher selection (10th Malaysia Plan, 2010).

The findings remark on the challenges for teacher education programmes in two ways; firstly, higher education administrators and academics have to find ways to attract students in the teacher education programme to their academic programme of study and motivate them to remain in teaching profession thereafter. Secondly, it is a huge responsibility to develop a student's positive identity towards the profession, and therefore achieve a positive image of teaching.

_

² Starting from year 2011, the Malaysian Government's 10th Malaysia Plan (2011-2015) abolished the guaranteed job privilege for students who enrolled in teacher education programmes from 2011 (Malaysia Prime Minister Department, 2010). Guaranteed placements have been removed due to the government's intention to ensure that only the best teacher candidates will be recruited and become major players in Malaysia's education system.

The centralised recruitment system for teachers which is exercised in Malaysia is a compelling reason to give attention to the development of professional identity for teacher education students. The intake for teacher education programmes at government-funded higher education institutions is based on the Ministry of Education's projection of the number of teachers needed in Malaysian public schools. It has been recorded that every year, the applications for teacher education programmes exceed the number of places offered by the Ministry of Education (Ingvarson et al., 2013). Ministry of Higher Education Malaysia (n.d.-a) report that teacher education programmes are among the five most popular programmes of study in demand by higher education candidates, followed by business and administration, engineering, and mathematics and science courses. In previous years, student intake³ for the educational bachelor degree level in Malaysian public higher education institutions was 12,910 students in 2009, 11,826 in 2010, 19,883 in 2011, 17,824 in 2012, and decreased to 9,877 students in 2013. In year 2017, intake for this programme was only 9,007 students ("Ministry of Higher Education Malaysia", n.d.-a).

As the admission to teacher education programmes in Malaysia is intensely competitive and tightly controlled by the government, it is important for the higher education academics and administrators in teacher education programme to fully utilize the scarce resources and minimise students' tendency to exit the profession before the end of training or shortly after service due to not being able to value the work, or have a positive orientation towards teaching. Indeed, past studies suggest that one of the factors contribute to beginning teachers attrition is inability to value teaching work (Cladinin et al., 2015, De Neve and Devos, 2017; Struyve, Daly, Vandecandelaere, Meredith, Hannes, and De Fraine, 2016).

In addition, over many years, this programme receives new intakes with higher academic achievements in their secondary school (Admission and Registration Unit, 2010). Definitely, teacher education programmes do not want to lose 'cream of the crop students' to other professions because society needs teachers that have outstanding academic backgrounds as role models for future generations. Thus, it is important to

-

³ The enrolment is based on the projection by the Ministry of Education, not determined by the institutions.

understand what motivates teacher education students to teach, what values and norms that aspire them to negotiate a positive image of teachers and what is the distinctive behaviour as a result of this positive image. The inputs are very useful to strengthening teacher education programme by catering for students' needs, taking into consideration their positive aspirations towards teaching profession.

1.4 Social Identity Framework

The theoretical framework underpinning the research on professional identity in higher education mainly focuses on the learning and teaching of work preparation that will cultivate the development of professional identity (Atadero, Paguyo, Rambo-Hernandez and Henderson, 2018; Jackson, 2016; Neve, Lloyds and Collett, 2017). There are two challenging issues which arise in regard to effective teaching and learning for future professionals. Firstly, it is important to determine the learning style or approach appropriate for students to learn about their profession. Thus, it is crucial to understand the students' self-definition on their future profession prior to suggesting appropriate learning styles to suit them. This is consistent with past research that revealed people's self-definition on a specific social category can strongly influence their learning approach (Biggs, Kember and Leung, 2001; Bunce et al., 2017). In this sense, the students' definition on who they are in one occupational group will help to identify the type of learning approach that suits them to learn about the profession.

Secondly, focusing on the learning and teaching of work characteristics and dispositions have limitations in explaining the students' sense of belonging in one occupational group. The empirical evidence highlighted in Subsection 1.3 signalled that there is evidence that students disengaged with their future profession, thus it is important to identify the students' self-perception on the uniqueness of their intended profession that might attract and encourage them to become a group member.

Furthermore, in searching for the students sense of belonging to one particular group, it is important to focus on why they are attracted to one group instead of another. It will help to explain if the group will bring significant changes to the students' attitude and behaviour. It is assumed that if the group provides students with a sense of meaning,

purpose and belonging and that will build a positive sense of social identity, then the students have a tendency to have positive psychological and behavioural consequences towards the group. Therefore, the current study will examine the students' feelings when they belong to the group and the consequences of that feeling on their behaviour. Students' feelings will explain the current view of the social identity of teaching.

The status of the profession also contributes to the chosen theoretical framework. Many researches related to higher education use the term professional identity to describe a way of being, evaluate, learn and make sense of occupational practice (Hallier and Summers, 2011; Trede et al., 2012) regardless the status of the occupational group. The sociology of profession, however has demonstrated that occupational groups can be understood in a variety of ways; the most studied is the notion of what constitutes a profession and non-profession. Profession is a group of occupation that has been given a better position in the social structure compared to non-profession (Hall, 1968).

Arguably, it is necessary to differentiate profession with non-profession as both groups have different aims, principles and aspirations that explain their social identity. The professional group aims to ensure their group members maintain the professional status while the latter aspires their group members to acquire the professional status. Consequently, it is predicted that to become a member of different occupational groups (professional status and non-professional status) requires the members to act according to their group' values and norms and give their behaviour a distinct meaning compared to before they became a member of a specific group (Haslam, 2004).

The process by which a group of occupations incorporates the structural aspects and criteria required of a profession is known as professionalization (Abott, 1998). Professionalization has been debated in a variety of ways, as sociologists studied the professions from diverse theoretical viewpoints in the past decade. The traditional professional model from a sociologist's perspective, where the taxonomic approach focuses on the special traits of the occupational group to enable them to become an established profession. Barber (1963) and Hall (1968) among others, defined profession as a group of occupation with extended and rigorous training, high prestige, high levels of authority and autonomy, a professional organisation and community, good salary and serves the community interest.

However, classifying professional work based on traits have been criticised, as the definitions are not stable (Gorman and Sandefur, 2011; Klegon, 1978), and it varied over time even though it was proposed by the same scholar (Ruth, cited in Klegon, 1978). As many groups of new occupations emerged, Klegon (1978) also argued that defining the profession by using occupational traits is insufficient due to the "inability to apply the list of criteria to concrete situations" (p.4). Despite the criticism, the taxonomic approach has provided grounds to differentiate profession and non-profession work, and this has been used until recently as a combination with other sociological perspectives on profession (Gorman and Sandefur, 2011).

Besides the taxonomic approach, another traditional model, the functionalist approach has been established and argued that profession is a group of occupations that should help to maintain the order of society (Friedson, 1970). Thus, functionalists further explain that to become a profession, one occupational group with specific traits has to contribute to the society, particularly those served by the profession. Later, the new movement in sociology of profession emphasises the profession as a special group of occupation that has social closure where the expertise knowledge of the profession becomes the boundary for other occupational groups to perform similar duties (Krejsler, 2005; Saks, 2010). Hence, the study of the profession presents the social identity of one occupational group together with the significant engagement needed to form a positive social identity.

The study of professional identity development of one profession mostly focuses on how students negotiate their identity content as prescribed by the related professional standard (Monteiro, 2015). However, far too little attention has been paid on how students in a non-profession occupational group are attracted to the group, and how the occupational group can contribute to the enhancement of students' knowledge, skills and attitudes. The latter issue brings to another question; although only occupations that are classified as an established profession has the professional standard and practice to be learnt and followed, what will be the guideline for occupations that are not rewarded the professional status? How will their future occupational group guide their behaviour?

A possible explanation for this might be where members of non-profession occupational groups might search for their occupational group norms and values that will differentiate themselves from other occupational groups. Monteiro (2015, p. 67) suggested the 'difference approach' to identify the identity of non-profession groups whereby "(the group) identify the characteristics peculiar to an occupation to cultivate them and advocate for improvement for its professional and social status".

Thus, the norms and values have to be something that will become the groups' aspiration, something to be followed and increase the groups' self-esteem (Mehra, Kilduff and Brass, 1998), and most importantly, such norms and values are built on trust to acquire a high status, if not the full-fledged profession. Gaining a high status is workable, provided that the members of the occupational groups demonstrate high commitment in order to change the occupational status (Gorman and Sandefur, 2011, Klegon, 1978; Monteiro, 2015). Significantly, working towards the groups' aspirations will enhance the groups' self-esteem and consequently help in upgrading the social position of a specific occupation. This is consistent with past researches that contended a positive group performance is the outcome of integration between the group and group members' goal (Hall, Schneider and Nygren, 1970).

This study focuses on teaching. Teaching is regarded as an old occupation and teacher education programmes have been offered in higher education for a very long time (Elliot, 1972; Monteiro, 2015). Teaching gives great influence to the development of the nation, however the teaching occupation has been regarded as a semi-profession because it meets some, but not all, of the specifications of a professional occupation (Etzioni, 1969; Krejsler, 2005; Ingersoll, 2003; Monteiro, 2015).

Even though the paradigm on what constitutes profession has changed from the taxonomic approach that highlights the traits of a profession to the capacity of social closure in the neo-Weberian approach, teaching has never been regarded as a full-fledged profession. The research conducted to date looks for evidence that justifies why the status is upheld regardless of the theoretical framework defining professions. It has been recognized from the literature that teachers' knowledge to perform their function becomes a leading cause of teaching status (Barber, 1963; Hall, 1968; Kerchner and Caufman, 1995; Saks, 2010). It has been argued that teaching does not involve the

development of specific knowledge, rather teachers are said as performing the function of manage and disseminate existing knowledge (Monteiro, 2015). The emergence of knowledge-based work (Gorman and Sandefur, 2011) and the revolution of technologies of information and communication have impacted teachers' knowledge and skills considerably, and these issues have negatively impacted the effort to recognize teaching as a profession. Additionally, teachers have not been given the autonomy to make decisions on educational policies (Toren, 1975), instead they are just implementing; thus making teaching powerless (Monteiro, 2015).

The social status of the profession might affect teachers' behaviour compared to the previous decade as many more challenges have emerged as a result of the new economic development. In upholding the semi-professional status, it is important for the students to identify the strength of the teaching to be identified with, and have significant values and norms that are able to guide their behaviour, thus enabling the teachers to present their positive social identity.

The proceeding paragraphs states the remarkable points for this study. Thus far, little is known on how this group of students try to understand their programme of study and future profession, particularly the type of professional identity students try to develop throughout their studies in higher education.

With these assumptions, the social identity approach will underpin the exploration of student teachers' professional identity development as this approach can explain what makes students define themselves in terms of one group membership (i.e. future teacher) rather than another group of profession. In addition, the social identity approach also provides a mechanism to identify what makes social identity salient; in other words why students choose teacher education programmes and what drive their engagement with teaching. The social identity approach also helps in determining the pride and self-esteem the members have even though they are a member of a less-privileged group. In this context, teaching is regarded as a semi-professional occupation and this might impact their sense of belonging to the group.

The social category (i.e. teaching profession) becomes salient when the student teachers can understand and assess the values and norms of teachers and the teaching profession.

The students are predicted to form and learn values and norms of teaching during their socialisation process in the teacher preparation programme. The process involves self-categorisation towards the group, how they learn the aspiration of the profession and how they assign the aspiration of teaching to themselves. Consequently, a social identity approach helps to explain the distinctive behaviour as a result of becoming a member of one specific group. This point will be discussed further in Chapter Two.

1.5 Aims of study

The aims of this study are to identify the development process of teacher professional identity among students undergoing teacher education programmes in higher education institutions, explain the perception of students on teaching, and discover the process involved in the development of this identity. This study will explore the identity towards teaching that the students bring to the teacher education programme and during commencement of this programme. Additionally, this study will identify the factors that influence the development of students' identity during the commencement of the teacher education programmes offered by higher education institutions.

Firstly, the study explores the factors that influence the decision to choose teaching. It is assumed that when students choose teaching, they are influenced by the positive aspects offered by teaching. This will be their initial perception of teaching, which will influence their cognitive redefinition about themselves and guide their future behaviour. Secondly, based on the premise that the group aspirations may shape and reshape identity, the study also explores the identity that these students acquire while they are in the process of completing their preparation programme at a higher education institution, and how social factors facilitate the process of identity acquisition.

1.6 Objectives of study

From the point of view of the teacher education students, this study intends to:

- 1. Discover what influences the students' decisions to choose teaching and the initial identity these students bring to the teacher education programmes.
- Explore how the teacher education students acquire a professional identity in their preparation programme and the significant factors that influence the acquisition process.

1.7 Research questions

This study will examine four main research questions:

- 1. What factors have influenced the decision to choose teaching?
- 2. What identity do the students bring to the teacher education programme?
- 3. How does group aspiration affect the professional identity development, or decline during the commencement of the teacher education programme?
- 4. How do the students explain their professional identity nearing to the professional practise?

1.8 Significance of study

The findings of this study will redound to society by considering teacher education students who will educate the future generations of the country. The greater demand to enrol in teacher education programmes justifies the need to train students with sufficient knowledge regarding the profession. Apart from the necessary knowledge and skills to become a teacher, students must be able to adhere to the aspiration of the profession as a basis for shaping their professional identity. For parents, this study gives input on the type of future teachers that will be available to teach the young and future generations.

For higher education administrators and academics, this study will enhance the understanding of the process of the students' professional development towards the teaching profession. This understanding is useful and can be utilised to design

curriculum and plan more effective teaching and learning activities, enhance students support programme and provide a more conducive learning environment. For the researcher, this study foregrounds the present study using literature from the disciplines of management, sociology, psychology, and education, thus providing cross-disciplinary research to explain the professional identity development process among student teachers.

Additionally, this study presents scientific evidence for local authorities such as the Ministry of Education in regards to student teachers' professional identity development for future planning on teacher preparation curriculum. Also, the current study will add to the Malaysian literature on the uses of interpretive framework, rather than rely on the currently available survey instruments in exploring and understanding student teachers' professional identity development. An interpretive framework explains in depth the process of identity acquisition among students. Furthermore, the study will help researchers to uncover the role of social factors and individual self-definition (i.e. using a social identity approach) in determining the professional identity of student teachers and the process contained therein.

1.9 Context of the Study

This study was conducted in Malaysia. Secondary school teachers in Malaysia must have at least a degree in education to qualify them to teach. There are multiple routes into the teaching profession in Malaysia, particularly for teachers in secondary schools. These are:

- 1. A Bachelor's Degree in Education or Integrated Bachelor's Degree with Education (ISMP in local acronym). This four-year programme is the dominant route into the teaching profession, conducted at public universities.
- 2. A Post Graduate Diploma (DPLI and KPLI in local acronym). This is a one-year full time programme (or 38 weeks, including a teaching practise at school), which prepares trainees with a degree in specialised areas. It was first introduced to prepare teachers to cater for the urgent need created by the rapid expansion of secondary education in the past decades, and still continues today

("Ministry of Education Malaysia", n.d.-a; Mokshein, Ahmad and Vongalis-Macrow, 2009). However, the intake for this programme is very limited.

In this study, teacher education students or student teachers refer to students who are undertaking a teacher education programme in the biggest secondary teacher training institution in Malaysia. At the time this study was conducted, this institution only offered a Bachelor's of Education, with different areas of concentration. Among the specialisations offered are languages, special education, moral education, geography, history, Islamic education, management education, economics education, and many others. The minimum duration is four years and the curriculum consists of classroom teaching and learning, teaching practice in public secondary schools, and an undergraduate dissertation. The Ministry of Education controls the admission to this programme. Students graduating with a minimum cumulative average grade of 2.75 points over 4.00 points are qualified to apply for the post of secondary school teacher. Teacher candidates who are successful in the teacher selection process may have their posting in selected public secondary schools throughout Malaysia ("Ministry of Education Malaysia", n.d.-b). In the past, almost all graduates that passed the minimum academic criteria from this institution, as well as other teacher education programmes, were employed after graduation (Mokhsein et al., 2009). Therefore, at the time this study was conducted, students who enrolled before 2011 (i.e. groups of students undergoing teaching practise and group of career change students) are guaranteed teaching placements, thus teaching continuously becomes a highly secure job for them. On the other hand, students who enrolled after the 2011 academic year (i.e. group of first year students) did not have privileges on teacher placements in government funded schools; however, there is always a demand for teachers in both private and public schools as the number of young people entering the education system in Malaysia keeps increasing.

Currently, there are 12 education faculties or departments in public and private universities in Malaysia. Teacher education graduates from universities will be assigned to teach at secondary schools. Additionally, there are 27 Teacher Education Institutes that offer only degrees or diplomas in education and prepare students for teaching in primary schools. The basic programme offered in teacher education faculties or departments and universities follows a common curriculum, as outlined by the Teacher

Education Division of the Ministry of Education. The six basic components are teacher dynamics, knowledge and professional competence, subject option and specialisation, self-enrichment, co-curricular activities and practicum at school (Ingvarson et al., 2013).

In Malaysia, the mainstream entry into a public higher education institution is controlled by the Ministry of Education via a centralised selection process, regardless of the new policy of teacher placement. The selection process initially uses a meritocratic screening that is based on students' academic achievements in the high school leaving exam, particularly the public examinations, Sijil Pelajaran Malaysia (Malaysia Education Certificate) and Sijil Tinggi Pelajaran Malaysia (Malaysia Higher Education Certificate). In general, public higher education institutions in Malaysia have the autonomy to set the admission requirements, as well as the number of enrolments for their institution. Despite the new regulation on teacher placements, the number of enrolments on teacher education programmes is still controlled by the Ministry of Education. Furthermore, there are only limited places offered in every academic calendar year. Previously, admission to teacher education programmes at universities was set in accordance to the forecasted number of teachers needed in Malaysian secondary schools ("School Division, Ministry of Education Malaysia", n.d.). The Ministry of Education's forecast is based on the number of secondary pupils' educational enrolment. The Ministry aims for a teacher-class ratio in typical schools that does not exceed 1.5 teachers per class. Currently, the number of pupils in Malaysian secondary school classes is 30-40. Despite the changes in the teacher placement policy, the Ministry of Education still controls the number of teacher education applicants, so as to avoid a huge surplus of teacher education graduates in the job market due to the overwhelming number of applications every year.

Despite the new regulations on teachers' recruitment, the number of applicants for teacher education programmes remains the same, regardless of the new policy on teacher placements. There are two possibilities that may explain this situation: firstly, the teaching profession is attractive as students are aware of the contribution of teaching to the development of the country and the nation. Secondly, teacher education applicants still expect job opportunities in teaching as statistics showed that the number of young people who enter education increases every year ("Ministry of Higher

Education Malaysia", n.d.- a); thus, there will always be a demand for teachers in the Malaysian education system.

For the past 13 years, the number of applications to the teacher education programmes has increased and the Ministry has implemented more objective control mechanisms, such as interviews and psychometric tests for teacher education applicants, in order to screen the best candidates ("Malaysia Examination Council", n.d.). This is to ensure that only those who meet the requirements and have the interest and inclination to be teachers are selected for admission into teacher education programmes. The psychometric test, known as the Malaysian Educators' Selection Inventory (MEdSI), was introduced in year 2007, consisting 300 questions and should be answered within an hour. The candidates sit for the MEdSI tests in selected higher education institutions all over Malaysia. The items in the instrument consist of four sections: personality (110 items), career interest (60 items), value integrity (70 items), and emotional intelligence (60 items). These instruments, developed by a group of experts from Malaysia's public higher education institutions, are based on the Personality-Programme Fit (P-P Fit). The 10th Malaysia Plan reports that among the huge number of applicants every year, many do not have the attitude and aptitude to become teachers (Prime Minister Department, 2010), which are measured in the four sections as indicated above.

Applicants who successfully pass the minimum test score will be called for an interview. Academic staff from the teacher education institutions will conduct the interviews. Successful candidates, thereafter, will undergo their teacher preparation programmes in selected higher education institutions, depending on students' preference list of higher education institutions. Explicitly, the Ministry of Education has created a proactive system in order to produce teachers that desire to become a teacher by conducting control mechanisms towards teacher education applicants.

Until recently, there have been few studies that investigated the MEdSI test (e.g. Mat Kasim et al., 2012; Noah, Karim and Othman, 2012; Hashim, Damio and Hussin, 2013). However, the studies focus more towards instrument development rather than the effectiveness of the test that may shed a light on the students' attitude towards teaching. Although the MEdSI test has become a tool to select the best candidates for teacher education programmes, the nature of the test is limited to explain the type of

conformity that students have with the teaching profession. Research on conformity suggests that students may conform to the profession because they are pressured to comply with the professions' aspirations, or they have information about teaching, but, to a certain extent, they may not conform to the aspirations of the profession (Abrams and Hogg, 1990).

1.10 Framework of Study

Most studies in determining the students' identity towards the teaching profession have only focused on quantitative approaches (Bakar et al., 2008; Zamri, 2011) There is a considerable amount of research that focuses on positivist approaches and tends to follow the mainstream research that uses the same, or almost the same, research instruments. Furthermore, the quantitative studies on identity development process have measured the effect of teacher preparation programmes on students' identity, such as the learning process, peer influence and social communication. As the learning process in higher education involves the learner, teachers, the environment and what is to be learned, the quantitative studies were inadequate to explain in depth the process of becoming a professional (Mohd Dahan and Ghazali, 2007; Sinclair, 2008). Recognising that the prospective teachers' live experiences are not an objective reality, in that there are multiple influences in the societal contexts shaping their identity, thus, it is inaccurate to adopt a deterministic approach when attempting to understand their experience.

This study will adapt the interpretive stance so that the social factors and their interplay can be discovered and identified. This will contribute to the literature on the teacher education students' engagement with teaching. Moreover, this study also contributes to teacher education and higher education literature, specifically as to how to explain the development of students' identity using the social identity approach. This study attempts to get inside and understand from within the students' experience on teacher education programmes. This is consistent with the findings that some higher education students, through early socialisation in preparation programmes at higher education institutions, are able to turn themselves into the kind of person that suits the situation

that they are in (Becker, 1964). Details of the methodology of the study are reported in the methodology chapter.

1.11 Organisation of the thesis

The thesis is organised as follows; this introductory chapter introduces the background, aims and research questions being addressed, as well as the significance of the study. This is followed by a literature review in Chapter Two. Chapter Two analyses the factors that influence students' enrolment in teacher education programmes, and the development of the students' identity during the commencement of their academic programme at higher education institutions. This chapter focuses on the role of social identity approach as an organising and unifying approach that accounts for the students' identity development towards teaching. Following this, Chapter Three discusses the method, detailing the study approach, data collection strategy, sampling process and the management of study. Chapters Four and Five present the findings of the study. Finally, Chapter Six concludes the results and the limitations are discussed.

CHAPTER TWO LITERATURE REVIEW

2.1 Introduction

This chapter begins with the introduction of the concept of professional identity, and how the concept will be used in this study. Firstly, the process of professional identity development among students in higher education is highlighted in order to get the commonality of the process. Secondly, reviews focus on the significance of the social identity approach underpinning this study and the categorisation process that students undergo when they decide to enrol in specific academic programmes in higher education. Finally, the review encompasses the students' learning experience that may explain the identity development process.

2.2 Professional identity development in higher education

Besides the definition by Trede et al. (2012), and Clouder (2008) on professional identity as citied in subsection 1.4, Patterson, Higgs, Wilcox and Villeneuve (2002, p.6) directly uses the term 'professional' to define professional identity as 'the sense of being a professional'. Past studies on professional identity development among students in higher education revealed how the students represent themselves as the member of one occupational group. This includes the narration about the knowledge, skills, and attitude that they get through the socialisation process in higher education (Jackson, 2016). The socialisation process happens in many forms that enables students to acquire the professional identity; this mostly comes from the interaction between students in specific communities (Jackson and Temperley, 2007; Reid, Dahlgren, Petocz, and Abrandt Dahlgren, 2008; Su and Chung, 2015), transformational learning (Bramming, 2007), and reflective learning (Cherry, 2005; Klenowski, Askew and Carnell, 2006).

Studies of students' professional identity development support the statement highlighted in subsection 1.4, that the concept of professional identity has been used interchangeably in the literature. This includes using the term 'professional' identity to represent how students see themselves as members of one occupational group (e.g. Su and Chung, 2015) even though many studies have not illustrated the elements of professionalism that must be included to ensure that professional development occurs. Arguably, previous researches have contributed to explain how students in higher education represent their readiness or unreadiness to become a member of their intended occupational group, however there is little explanation on whether their behaviour is consistent with the group norms and values. Nevertheless, the results from previous researches have given inputs to the higher education administrators and academics on the way to plan, organise, lead and control students' learning experience in higher education. Additionally, researching professional identity helps to identify students' readiness to work, as this narrates the students' perception on how they see themselves as individual workers in one occupational group.

All things considered, the current study defines professional identity as a sense of being a professional according to the professional model attribute (for professional group of occupation) and a sense of being a professional according to their perception of the groups' aspiration (for non-professional group of occupation). The term 'profession' will be used in this study to indicate one occupational group, not limited to the established profession, as defined by Hall (1968), which comprises the field of law, medicine and priest.

2.3 The relevance of social identity approach in interpreting the professional identity development among students in higher education

Subsection 2.2 shows empirical evidence over the years that has provided support for the validity of the identity development process among students commencing in higher education and has demonstrated their relevance to explain who they are as a member of their intended profession. However, despite the different focus of each of these approaches to identity development, one commonality is that they have been used mainly to understand the processes underlying the intended behaviour of students as

individuals. Put it another way, research on identity development has focused on the individual process to become a member of their intended profession including their motives, personality traits and anticipation in their future professional working life (Bramming, 2007).

Hence, the focus on the previous studies is more towards defining professional identity when personal identity is salient, as it is "associated with needs to self-actualised and enhance personal self-esteem by means of personal advancement and growth" (Haslam, 2004, p. 67). Evidence from the literature also suggests that different approaches have been applied to identify the professional identity formation among students in higher education, involving both group effort (e.g. the communities of practise) and individual effort (e.g. reflective learning) to present who they are as a member of their intended profession.

However, as society will see the profession as one entity, the members of the profession have to function in concert and cannot always be seen as representing independent bodies. As a result, students are not necessarily driven by personal motive, aspiration and professional anticipation only, but informed by the norms and values of the occupational group (Haslam, 2004). To put it differently, the occupational group will act as an important drive to regulate the behaviour of group members, marked by the capacity for groups to determine and change the cognitive process of individuals (Tajfel and Turner, 1979).

It is commonly accepted that students who will become a member of established professional groups such as law and medicine, the professional standard and practice for the profession is something that needs to be learnt and followed. On the contrary, members of non-professional groups, such as teaching, their intended behaviour should support the occupational group norms and values in order to enhance the social position of teaching. As the existing traits and functions of teaching are not sufficient to bring teaching to a full-fledged profession, there should be outstanding or differentiable values and norms in teaching that will guide their members' behaviour particularly to upgrade the teaching status in the society. In this sense, the ability of one occupational group to acquire a high status depends on how well the group members strive and share their responsibility to enhance their occupational group performances (Gunlach,

Zivnuska, and Stoner, 2006). Thus, it is predicted that the group members' behaviours are shaped and reshaped by the group's norms and values that will differentiate teaching with other occupational groups.

It has been emphasised that groups of professional occupation and groups of occupation acquiring for professional status have different aims which suit their social positions. As stated in subsection 1.4, the former aims to maintain their privilege as a professional occupation while the latter struggles to upgrade their status to acquire professional status, or at least get the recognition as a special type of occupation based on their contribution to the society. Hence, in the context of this study, the social identity approach will help to explain the students' intended behaviour, or their assumed professional identity and the process of identity development towards the intended profession. With the argument that the psychology of the individual is a product of group life, this study seeks to identify how students are bound together through their joint sense of belonging to the same social category which is their intended profession. The social identity approach suggested that a social identity associated with membership of a given organisation will make organisational behaviour possible (Haslam, 2004). In the current study, the 'organisation' refers to one's occupational group.

The social identity approach comprises of Tajfel's (1974) Social Identity Theory (SIT) and Turner's (1985) Self-Categorisation Theory (SCT). SIT was developed from Tajfel's work on intergroup processes, and Turner extended the work on group processes starting from the formation and categorisation of the group. Because these two approaches shared many key assumptions regarding group processes and intergroup relations, they are often discussed under the label of social identity approach in literature related to students and their future profession (Burford, 2012; Clouder, 2003; Hallier and Summers, 2011). These prior works deduce that the social identity approach and the model therein help to explain why students are attracted to different types of profession, and how the profession guides their behaviour accordingly. For non-established professions, it is a need to identify social identity salience among the members or future members of the profession. This in turn will help to explain how students use their occupational group for self-enhancement.

Different types of professions have different history that need to be considered in order to understand its unique norms and values (Klegon, 1978). In spite of the non-professional status accepted by most scholars, in a Malaysian context, teaching is regarded as having a high status even though the theoretical perspective on sociology of profession has excluded teaching as a type of established profession. Malaysia is a multiracial country consisting of Malays (the majority), Chinese, Indians and other small groups such as indigenous Bumiputras. Islam is the national religion and all Malays are Muslims⁴, but other major religions are practiced, including Buddhism, Hinduism, Christianity, Sikh and Taoism.

As a country that has a majority Muslim population, the religious and cultural values of Islam have a significant impact on social policy and practice, including education (Rassool, 2000). Following this, the position of teachers in Malaysia is influenced by the religious and cultural values of Islam. Teachers are recognised as knowledgeable people and Islam gives a high rank to a knowledgeable person, as stated in the Al-Quran⁵; a knowledgeable person is assumed to be able to disseminate knowledge and positive values and have a positive impact on other people through their efforts in teaching. Teachers are afforded high esteem and are viewed as role models and respected community leaders (Mogra, 2010). By being a role model, teachers are also expected to exemplify in their life the content of what they teach (Minnis, 1999). Therefore, in Islam, no other profession can compete with the teaching profession in terms of virtue (Mogra, 2010).

By taking the status of teaching into consideration and in the context of this study, two assumptions arise on how the social identity approach can help to explain students' identity development towards the teaching profession. Firstly, there is a need to identify the norms and values of the professional group that will make the social identity salient. Sachs (2003) illustrated:

"It is paramount that whatever meaning of professionalism is circulating, its meaning is generated and owned by teachers themselves in order that it

-

⁴According to Article 160(2) Malaysia Federal Constitution, all Malays are Muslims. Malays who convert to another religion or do not have a religion are not considered as Malays (by definition).

⁵ Central religious text of Islam.

should have currency among teachers and be useful in improving their public image and social importance." (p.3)

Therefore, it is feasible to argue that the status of teaching perceived by the local community is treated as value owned by the teaching profession. Even though teaching is not awarded with an established professional status, there is no doubt of the contribution to the society in facilitating learning. Thus, in Malaysia, students who identify as a future Malaysian teacher must be proud because they have a great opportunity to become a member of the teaching profession, due to the higher status given by the local community (i.e. noble and respectable). The motivational attachment to the group has been defined as one of the basis of the group's positive performance (Hogg and Abrams, 1998).

Another possible event that possibly guided the process of identifying the values and norms of the teaching profession is The National Philosophy of Education statement, which has been used in all educational activities. In a Malaysian context:

Education in Malaysia is a continuous effort towards enhancing the potential of individuals in a holistic and integrated manner in order to create individuals who are well-equipped intellectually, spiritually and emotionally. This effort aims to produce knowledgeable, ethical and responsible Malaysian citizens who can contribute towards the harmony and prosperity of the community and nation ("Ministry of Education Malaysia", n.d-a).

This statement consequently highlighted the important factors that teachers must adhere to, thus explaining the norms and values of teaching that might guide a teacher's behaviour.

Secondly, the identity development towards teaching occurs in intergroup and intragroup contexts. When students decide to enter teacher education programmes, they must understand that they want to be different from other students who choose different programmes of study. This is the process where members of the group define their own social identity through a discrimination process between their group and others (Turner, 2010; Haslam, Powell and Turner, 2000).

At the same time, their decision will give the impression that in order to become a member of the profession, they not only want to be seen as physically attached to the group, but they want to become part of the group which carries all the attributes of the profession. Tajfel (1978) explains this process as "the part of individual's self-concept which derives from knowledge of membership of a certain group together with the value and emotional significance attached to that membership" (p. 63). Thus, the professional identity towards the teaching profession will be framed both from the students' motive to enter a teacher preparation programme and their socialisation process in higher education.

The initial identity development towards the profession suggests that early engagement with the occupational group starts with the student's motive in choosing the disciplinary field as a means to become a member of a certain profession related to the field of study (Gibney et al., 2010). Students are assumed to be involved in the cognitive redefinition process of the social group, i.e. the profession (Turner, 1985), when they choose their academic programme. The choice stimulates their thinking of which group of occupation they will be in, thus provides an early conception of their future image.

Potentially, students may be attracted to a group of occupation with a positive social identity and will continue to imagine themselves as continuing the esteem of the group. However, the perception of positive social identity is varying; ranging from better positions in the society, rewards offered, opportunities for employability, the nature of work, and work autonomy. To some extent, student perception on one group's positive social identity does not correspond with the norms and values of the group. For example, students may be attracted to one occupational group as it offers greater employment opportunity (thus, indicates positive social identity) while the group's aspiration is to serve society to the best interest.

The reviews in the preceding paragraph implicitly suggests that the strong emotional ties between the individual and the group is commitment. According to the Social Identity Theory, identification with a social group influences the individual behaviour, and can serve as a motivator and leads the effort to achieve the group's aspiration (Meyer and Allen, 1991). As stated in Subsection 1.3, teaching now has been critiqued

due to the unintended behaviour of its members, and this brings a negative perception about teachers, hence downgrading the status of the profession.

The downgrading or deprofessionalisation challenged the positive social identity that teaching inspires to brings towards society. Thus, it is feasible to argue that the positive social identity will be realised, and subsequently help to uphold the teaching profession's status, and enhance the self-esteem of its members if teachers as well as future teachers present their commitment in carrying out the mandate or function in society (Geer, 1966; Hord and Tobia, 2012). Thus, it is important to identify the level and types of commitment student teachers have in order to achieve a positive social identity.

Commitment is commonly defined in the literature as emotional ties between an individual and the organisation they are in (Meyer and Allen, 1991). This includes one's occupational group. Previous researches have demonstrated that commitment can predict behaviour, and engagement with the group can be a predictor of group-oriented effort (Ellemers, Gikder and Haslam, 2004), hence become a predictor of professionalism (Geer, 1966).

In the literature, teacher commitment has been found to be a crucial predictor of teachers' work performance, absenteeism, retention, burnout and turnover, as well as having an important influence on school pupils' motivation, achievement, attitudes towards learning and being in school (Day, 2002). The level of teachers' commitment is considered a key factor as it heavily influences the teachers' willingness to engage in teaching and perform their duties.

Teacher commitment also becomes one of the most critical factor for the future success of education and schools (Crosswell and Elliot, 2004). Crosswell and Elliot (2004) suggested that "while teachers do articulate a commitment to external centres (such as students) they also make significant links to personal passions which include ideology, values and beliefs" (p. 7). They suggested that there are different ways that teachers perceive, understand and conceptualise teacher commitment; one is towards maintaining professional knowledge. The importance of commitment is also highlighted by Geer (1966, p.31), "particularly in occupations aspiring to be a profession or in

doubt of their status, there may be great concern because it is considered a mark of the real profession". Thus, commitment towards acquiring and maintaining professional knowledge becomes important in determining the teachers' commitment in teaching.

One type of commitment, termed as affective commitment, explains the desire to maintain membership in an organisation (Meyer and Allen, 1991), and also understanding the members' willingness to invest effort and knowledge to achieve the organisational goal (Allen and Meyer, 1990). Arguably, affective commitment may explain further the emotional aspect of under-privileged or under-represented groups such as teaching which the social identity has been challenged with the phenomena affecting teachers nowadays.

2.4 Students' motives to choose disciplinary field and occupational group

Knowing the students' motives when they choose a disciplinary field in higher education is important for the current study as it can identify what led them to become a member of one occupational group; therefore, giving an initial perception on the social identity of one occupational group. It is also feasible to argue that motives serve as an early predictor of the students' orientation towards maintaining a positive social identity. However, when the perceived social identity is not congruent with the occupational group's norms and values, the suitability of their early identification may need to be revisited in order to identify how similar they see themselves in relation with the group's aspiration. The need to revisit is also applied in the situation where students enrol in specific academic programmes without any concrete motives, but later realise that they are becoming a member of one occupational group.

This revisitation is workable as the social categorisation theory also suggests that identification is viewed as a perceptual cognitive construct that is not necessarily associated with any specific behaviours or affective states (Ashforth and Mael, 1989, p.21). Reference to the social identity approach provides a systematic explanation of the various cognitive and behavioural strategy in which students may cope if there are any differences between early identification and recent situation (Haslam, 2004). In one such enhancement strategy, individual mobility is associated with individual belief in

the possibility of advancement by joining a higher status (occupational) group (Tajfel and Turner, 1979). Thus, students may think of changing their academic programme if they think there exists a dissimilarity between their motive and the group norms and values. Students might exit the group and search for a more higher status group, in order to become a member of a group with a more positive social identity.

However, when the chance to change the academic programme or chosen disciplinary field is impossible because of insufficient academic background or other valid reasons, students might apply a different strategy to achieve a positive social identity. The social identity theory predicts that when an existing occupational group status is perceived to be low, in order to maintain the social identity salience, students might pursue a social creativity strategy (Haslam, 2004). Social creativity enables students to create a more positive image to achieve a positive social identity. According to Haslam (2004), there are three ways students might respond to social creativity; find a new positive dimension to compare an existing group with an out-group, evaluate the existing group with new positive dimensions, or redefine the meaning of in-group membership to enhance the status. Pursuing a social creativity strategy enables the students to remain in the existing group; however, there is a need for the students to show affective commitment in order to achieve a positive social identity.

Hence, it is important to identify students' motive to enrol in a specific disciplinary field in order to assist the students to achieve a positive social identity, particularly when their motive to enrol in the field is not related to social identity salience but rather personal identity salience. Previous studies in higher education have reported the three main motives students choose a specific disciplinary field; namely altruistic, intrinsic and extrinsic. In a teacher education context, altruistic motives refer to a feeling that shows a desire to help people (i.e. young people entering the education system). It can be inferred that altruistic motives relate to the students' perception on teachers' function and status of the profession that enable them to contribute to the society while intrinsic and extrinsic motives are viewed as rewards (internal and external rewards respectively) gained from the profession.

Studies in Malaysia suggest that student teachers favour teaching because of the nature of teaching work and appreciate the status of the teaching profession (Harun, 2006;

Azman, 2013). Azman (2013), for example, finds that student teachers realise that teaching enables them to work with children, and this helps to develop the younger generations. Harun (2006) added to the Malaysian literature on student teachers' motives by reporting that students in her study were attentive and thoughtful towards teaching young children. The findings, thus far, are insufficient to explain whether students evaluate teaching work when they choose to enrol in teacher education programmes.

Subsequently, it is also reported that some students do not favour teaching work at all as they would rather approach teaching as a means to seek a permanent job (Seker, Deniz and Gorgen, 2015). Additionally, most literature focuses on the status of the teaching profession and suggests that students will show engagement with teaching when they feel proud to belong to the group and feel esteemed with the profession. In addition, students with a positive perception on teaching are expected to preserve the status of the profession by following the norms and values held by the profession. Studies on teacher education in Malaysia provide more evidence that the status of the teaching profession has stimulated their thinking and feelings towards teaching as a profession.

The positive social identity of teaching where they perceive that teaching is a noble profession becomes part of the students' evaluation of the profession (Azman, 2013; Harun, 2006; Wellington, 2012), rather than their understanding of how teaching works. Nonetheless, literature that focuses on the rewards of teaching, particularly studies conducted in Malaysia, provides ample evidence of the students' attitudes derived from their beliefs and emotional reactions towards teaching. It has been reported that students possess positive attitudes towards teaching as a result of the benefits embodied in the profession such as long holidays, monetary incentives as civil servants, job security, and the short duration of working time (Azman, 2013; Harun, 2006; Wellington, 2012). The understanding of rewards also influences the student teachers in mainland China who chose teaching as a way to work in a vibrant place such as Hong Kong (Gao and Trent, 2009).

Despite the findings, the majority of respondents in the studies conducted in Malaysia report a positive evaluation of teaching in a way that they state their willingness to approach the teaching profession after graduation. This condition is predictable as

teacher education programmes provide professional training and it can be assumed that most students choose these programmes with the intention of becoming teachers. This is consistent with the work of Berger (1988) and Teng (2008), cited earlier in this chapter. What differentiates the individual students' evaluations about teaching is their level of engagement with the work, the status they hold in the profession and the type of rewards they expect from it.

2.5 Socialisation process that helps in identification of teaching aspirations and identity development

As highlighted in Subsection 1.1, higher education provides early socialisation for students towards the profession (Bakar et al., 2008; Trede et al., 2012). Prior studies reveal that the students' motive give insight into how students socialise in their professional preparation programmes (Fokken-Bruinsma and Canrinus, 2012, 2014; Struyven, Jacobs and Dochy, 2013). This subsection further explores the process of socialisation in higher education with regards to the identification process towards a specific profession. It is predicted that students will learn the professional roles and dispositions of teaching through professional socialisation, as primarily this process enables students to acquire knowledge, skills and values (Cohen, 1981) related to a specific profession. Also, through the socialisation process, students will be able to determine how similar they are to the groups that they find themselves occupying (Tajfel and Turner, 1979).

The aim for this subsection is to explore how different approaches of socialisation affect the students' process of learning professional attributes. Two approaches of the socialisation process that are most probably undertaken by the students while they are at higher education institutions will be highlighted, together with the reviews on students' perceptions towards the teaching profession in Western countries, Asian countries and Malaysia. This is to identify whether there are differences in socialising approaches among the different countries and the effect of the identification process. Arguably, the differences have taken into account identifying the occupational status which also involves the historical aspect on how one country sees the profession (Klegon, 1978). Finally, this subsection will suggest the suitability of the socialisation approach,

considering which approach will result in strong identified members that adhere to the teaching aspirations, which in this context is commitment.

2.5.1 Introduction to students' socialisation in higher education

This subsection explores what higher education offers students during the socialisation process that might influence their commitment and develop a positive social identity towards teaching. Certainly, in higher education, students have an opportunity to explore the epistemological aspects or nature of knowledge in their specific discipline (Dahlgren, Hult, Dahlgren, Segerstad and Johansson, 2006; Johansson, Segerstad, Hult and Dahlgren, 2008). Dahlgren et al. (2006) further suggested that this arrangement frames the students as members of specific academic and professional cultures. It is important to realise that knowledge is treated as an important criteria for occupational groups to be regarded as professions (Barber, 1963; Hall, 1968).

The socialisation process exposes students to a different level of knowledge during their learning process in higher education. Certain commonality in the literature suggests that students in their early years of studentship are exposed to descriptive knowledge about their academic major, including the root of the discipline and related theories (Dahgren et al., 2006). Descriptive knowledge is received by students mostly from the memorisation of facts derived from reading materials and seminars. In certain circumstances, students do not envisage a typical target job related to their discipline at this stage (Johansson et al., 2008) and their approach to learning is more oriented towards the memorisation of tacit information that focuses more on the intention of reproducing information without any further analysis (Huy, 2008).

When students progress in their studies, they become more exposed to specific knowledge related to their programmes of study (Dahlgren et al., 2006; Jackling and de Large, 2009). For example, an accounting major must have the ability to execute the technical aspects of accounting such as key accounting skills and accounting problem analysis (Jackling and de Large, 2009). For students in psychology, they are trained to be the "helper and the social engineer capable of moderating people's behaviour" (Dahlgren et al., 2006: 578). From the studies cited above, specific knowledge can be inferred as important attributes of the students' future professions related to the

programme of study. Although some students appreciate the knowledge and are keen to be a member of a particular academic and professional culture (Dahlgren et al., 2006), their counterparts might refuse to become a member (Chambers, Hobson and Tracey, 2010) when they feel that they are not willing to self-categorise themselves consistently with the professional attributes even though they are still commencing in the particular academic programme. For example, in teacher education, Chambers, Hobson and Tracey (2010) found that final year teacher education students decided to withdraw their intention to become teachers as they found that the teaching profession is not promising in terms of extrinsic rewards, such as pay, lack of confidence when dealing with school children, and receiving less support from schools during their internship programme.

This study implies that students potentially distance themselves from the group, as they have not adhered to one of the teaching attributes, represented by difficulties in handling school children. Consequently, this suggests that students do not possess favourable attitudes towards the teaching profession as they are not able to adhere to the teaching attribute. At this stage, students may think to pursue an individual mobility strategy if they assume that their career motivation is not in harmony with the existing group.

Besides knowledge, students in higher education, formally or informally, are trained in related skills. In a working environment, skills normally refer to technical procedures or practical tasks and people skills (Roselina, 2009). Technical procedures or practical skills are related to discipline-based knowledge, while people skills relate to generic skills that are transferable between different contexts, such as communication, leadership and decision making (Badcock, Pattison and Harris, 2010; Baik and Greig, 2009; Dahlgren et al., 2006; Reid et al., 2008; Sleap and Reed, 2006). Higher education is the platform for students to learn the skills needed to function in specific professional roles (Borgen and Young, 1982). However, for certain professions, such as teachers and doctors, students may be able to envisage the role of the profession as a result of their socialisation as members of a society.

In conjunction with knowledge and skills, professional and generic values are also embedded in the students' teaching and learning process in higher education (Berings, Fruyt and Bouwen, 2004; Pike, 2006; Taber, Hartung and Borges, 2011). In the learning process, students may absorb values through role models (Sleap and Reid, 2006); to some extent, students also acquire values through their interaction within formal and informal learning (Hafferty, 2008; Lui et al., 2008). Hafferty (2008), for example, suggests that medical students learn about their professional values through observing the attitude of their physician teachers.

The reviews in this subsection reveal that students will learn different types of knowledge, skills and values pertaining to the profession. It is assumed that this acquisition of knowledge, skills and values may help in the process of identification with the norms and values of the given profession.

2.5.2 Professional socialisation approach

Researchers categorise students in higher education as an individual in the adulthood phase (Arnett, 2000; Brim, 1966; Mortimer and Simmons, 1978). As an adult, students are expected to become more realistic in their decisions (Clausen, 1968) because they are able to synthesise what they have learned, become more independent and proactive (Brim, 1966; Cottrell, 1969). Additionally, Knowles (1984) also suggests that adulthood is a phase whereby individuals will arrive at a self-concept of being responsible for one's own life, including their decision in a professional career. Adults, therefore, are perceived as independent and able to develop themselves according to their beliefs and feelings in their lives.

In the professional socialisation literature, there are two main socialisation approaches through which students in higher education learn about the attributes of their profession; the functionalist and interpretive approaches of socialisation. Each approach is characterised by a distinct theoretical orientation towards describing the students' learning process in higher education. The following review aims to explore both approaches and suggests how the socialisation process affects students' identification with the important attributes of their assumed professional destinations.

2.5.3 Functionalist approach of professional socialisation

From a functionalist point of view, the socialisation process is seen as a platform where students have to adopt predetermined structures of the profession (Merton, 1957). The structure, however, varies accordingly in different contexts, as functionalists tend to assume that the observer (i.e. society) attempts to "relate what they observe to what they regard as important elements in the wider social context" (Burrell and Morgan, 1979. p 107). The underlying assumption behind this approach is stated by Burrell and Morgan (1979), who wrote that the social world "regards society as ontological preceding man and seek to place man and his activities within that wider social context" (p. 106). This assumption explicitly gives priority to the social structure, as its existence precedes man, and man has to fit in society. Thus, it implies that in professional socialisation, students who intend to become a member of one chosen profession must simply accept a set of professional values that are prescribed to its members (Baumeister, 1991; Hansen, 2010). The functionalist approach, therefore, results in passive learners, as students are assumed to adapt themselves to the attributes of the assumed professional destination, in order to fit in.

This approach also considers that there will be a role model for the socialisation process, as Burrell and Morgan (1979) literally state, "society comes before man" (p. 106). Van Scotter, Haas, Kraft and Schott (1985) support this view and argue that the socialisation process is similar to the process of teaching and learning where the teachers consist of the combination of society and higher education, and students are learners. The learner, therefore, will follow the teachers in order to fit into the given group in a given context. In higher education, the learning process also involves the teacher and as Merton (1957), and Radcliffe and Lester (2003) suggest, teachers become vital socialising agents for students to learn the attributes of the profession.

If students adopt this approach of socialisation, it would mean that student teachers have to follow all the predetermined knowledge, skills and values during teacher preparation programmes in order to learn about specific values and norms of teaching. Students, however, are restricted to synchronise what they learn from different people, as functionalists consider that learners must depend on the "social structure". Different social structures (i.e. teacher educator) might have different perspectives on teaching

attributes. It may be good for the students to learn in this way but they will be filled with teaching attributes in an unconscious, passive uniform process of socialisation (Artkinson and Delamont, 1985), and the role model to follow (i.e. the teacher educator in higher education) may not produce strong identifiers for the teaching profession. This assumption is supported by the view that in order to strongly identify with the teaching values and norms, the students must conform to the attributes of the profession, not with other people (i.e. teacher educator and society) who influence them. In addition, as this approach equates students with passive objects, with no ability to react to the socialisation process, it may be questionable, as not all students in higher education will approach their assumed professional destination after graduating from higher education institutions (Sinclair, 2008).

However, it is predicted that for some students, particularly those who enrol in specific academic programmes in higher education without knowing what their direction or future profession might be, they may find that a functionalist approach to socialisation is beneficial, as they have a role model to follow and the adoption of the knowledge, skills and values are in order to prepare themselves towards certain professions. The students are assumed to develop a positive social identity towards the profession with the help of scientific learning that allows the students to seek approval and passively conform to the performance standard (Baratti, 2004).

2.5.4 Interpretive approach of professional socialisation

As opposed to the functionalist view, the interpretive approach of socialisation assumes that the students in higher education develop themselves according to nonlinear processes, which offer an opportunity to continually engage in the active learning process on which they will become (Giddens, 1991; Niemi, 1997). With this assumption, the socialisation process focuses more on human agency, rather than a predetermined structure. Human agency in this context refers to the capacity given to the students to create their own meaning in their socialisation process. Therefore, students are seen as active participants in their learning process and have the potential to actively seek the information they need. Burrell and Morgan (1979) interpret this approach as a way to seek explanation "within the realm of individual consciousness and subjectivity, within the frame of reference of the participants as opposed of the

observer of action" (p. 28). This suggests that students develop their own meanings and "do what they do because of whom they believe they are" (Forte, 2007: 167).

So, what makes students active participants? Generally, studies show that students will become active participants if they are able to use their knowledge and skills to make decisions in certain aspects (Reilly and Spratt, 2007; Wang, Shen, Novak and Pan, 2009). In an attempt to clarify why individuals become active participants in the interpretive socialisation process, Eraut (1994) tries to make a distinction between the types of knowledge that individuals have; firstly, knowledge that is socially constructed, and secondly, knowledge that "individual persons bring to situations that enable them to think, interact and perform" (Eraut, 1994; 202). The author suggests that both types of knowledge are important and facilitate the socialisation process.

Put differently, prior work suggests that both personal characteristics and social interaction are important in facilitating individual learning and becoming active participants in the socialisation process. Similarly, other scholars also emphasise traits such as commitment and motivation that help to create a meaningful and successful socialisation process (Becker, 1964). Additionally, flexible, higher self-efficacy, independent and creative students are also predicted to contribute to the definition of active participants in the socialisation process (Bandura, 1969). This contrasted with the functionalist approach that claims that students' learning originates from outside and is derived externally. This contrasting approach implies that the motivation of students to search for their own stance in interpretive approaches are higher than for students that are involved in predetermined socialisation processes.

The interplay between the students' characteristics and social interaction in facilitating the socialisation process, as discussed in the preceding paragraph, emphasises that even though personal factors dominate the definition of oneself, individuals will also act differently according to the context where they construct their identification towards certain objects. As mentioned in the preceding paragraphs, individuals still hold their own perceptions about themselves in a different context, but as Heggen (2008) suggests, the reflective skills that individuals have will enable them to socialise in a different context.

Tomlinson (2007) also supports this assumption in the professional stance and suggests that profession is not only a technical matter, "it is also a personal matter which involves the location of self in an on-going process of engagement within which they operate" (p.287). Drawing on Tomlinson (2007), many studies emphasise how student teachers' engagement with significant others affect their self-definition on the teaching profession (McPherson, 2007; Rots, Aelterman, Vlerick and Vermeulen, 2007; Whitehead, 2007). The studies implicitly claim that teachers become active participants in the socialisation process; however, there is no evidence in the literature detailing how the socialisation process occurs.

The studies cited above suggest that becoming active participants implies that the students in higher education are not simply accepting replication or enacting teachers as dominant role models, but they also learn about their profession through interaction with the environment (Niemi, 1997). Interaction with the environment takes place from self-learning and students' agency, thus minimising the condition to simply accept any predetermined values and norms about the profession. As reviewed in the preceding paragraph, social environment might consists of significant others, such as teachers, faculty members, school children and peer students; however, the students do not conform to others but rather treat them as information providers (Baglin and Rugg, 2010).

The interpretive approach of socialisation fits in the social identity framework in determining the students' engagement with their assumed professional destination in a way that the students, as active participants in the socialisation process, can consciously determine which group they will be in. The students that envision themselves as members of a specific social group are motivated to differentiate between in-group and out-group members (Glaser-Segura, Mudge, Bratianu and Dumitru, 2010; Owens, Robinson and Smith-Lovin, 2010); therefore, undergoing the process of self-categorisation. The interpretive approach also gives greater flexibility to develop positive attitudes towards the profession, as by becoming active participants, the students have the opportunity to develop themselves, not to the extent that they only have a favourable or unfavourable attitude towards the group (i.e., profession), but to a certain extent explain how far they internalised the aspiration of the profession.

Professional socialisation literature continuously highlights the process of identification for certain professions among students in higher education from Western countries (Hong, 2010; Levine-Rasky, 1998). However, only limited studies from Eastern countries find that an interpretive approach is used extensively in the professional socialisation process (Ahmad, Anantharaman and Ismail, 2012; Pauline, 2013). However, the indicated studies focus more on how students become independent in their learning process to acquire general knowledge and skills in higher education; although they do not explicitly relate this with the learning of attributes of a specific profession.

In summary, the students' professional socialisation, regardless of the approach, helps the students to learn about the attributes of their assumed professional destination. Nonetheless, the students that prefer the functionalist approach of socialisation might have less opportunities to further explore the attributes of their assumed professional destinations, as compared to the students that have the opportunity to become an active participant in their socialisation process. In addition, it is predicted that the students undergoing the interpretive socialisation process internalise more professional attributes than the students undergoing the functionalist socialisation process.

This prediction is based on the underlying assumption that students are influenced by significant others, rather than trying to conform to professional attributes if they are engaged in the functionalist socialisation approach. Thus, low identifiers with professional members might result from this socialisation approach. Another important thing that emerges in the review is that interpretive socialisation does not always result in positive perception towards the profession. This is because to some extent, students with an interpretive approach of socialisation may leave the profession related to their programme of study, as their self-categorisation is consistent with the social categorisation of other social groups.

To date, research on interpretive approaches of socialisation dominates the professional socialisation of students from Western countries. Subsection 2.6 explores the factors that contribute to the unpopularity of the interpretive approach to professional socialisation in Eastern countries, including Malaysia.

2.6 Interpretive approach of socialisation: Malaysian perspective

Prior discussion in Subsection 2.5 implicitly highlighted that students undergo the socialisation process based on their learning preferences and the opportunities provided by the higher education institution. It can also be inferred that students' learning reflects the culture where they belong (Bragg, 1976). Thus, this means that culture has a significant impact on how students in higher education undergo their learning process. There is some debate about the differences between how students originating from Western cultures differ in their learning approach compared to students in the East (Liu, 1998; Richardson, 2004). A common argument is that Westerners are more inclined to learn independently, whereas Easterners prefer to depend on the teacher as a medium of knowledge transfer (Liu, 1998). Western learning models, such as the Constructivist Theory (Piaget, 1972) and Kolb Theory of Learning (Kolb, 1984) have been developed and designed for a Western culture. These theories, among others, posit that students synthesise their new experiences from interaction with the environment to make sense of the world (Brooks and Brooks, 1999).

This approach is contrary with the traditional beliefs of students from Eastern countries. Liu (1998) proposes that students from China refer to their culture as a "long tradition of unconditional obedience to authority" (p.5); therefore, teachers are credited as people that students have to obey. Socialisation literature in Middle Eastern countries also suggests the same results, as students do not welcome reflexive approaches as part of the learning process (Clarke and Otaky, 2006; Richardson, 2004). This implies that students from Eastern countries are passive learners who are reticent and taught to obey (Morris and Leung, 2010).

Additionally, the acceptance of a higher power creates distance between teachers and students (Hofstede, 1984), which also contributes to a passive learning environment. According to Hofstede, the higher power distance means that there is a formal relationship between teachers and students, and teachers are perceived as individuals who outline the path to be followed by students (Hofstede and Hofstede, 2005). As a result, students prefer teachers to make decisions on their behalf. One criticism of this learning style is that it hinders student creativity (Morris and Leung, 2010), and makes students passive followers; thus, they accept the professional socialisation process as

determined by the profession (Merton, 1957). So, how does this condition relate to the students' identification process? It is assumed that there is a high probability that students in Eastern countries identify themselves as members of the group because of influence from others. In contrast, students who have the opportunity to become active participants will become more realistic in determining the common attributes shared by the members of the profession.

Malaysian students, as Eastern learners, are also predicted to have the same attributes as students in China and the Middle East. Although there are no studies that clearly suggest that Malaysian students are traditionally inclined towards the functionalist approach, a considerable amount of studies in higher education are geared towards this approach. For example, studies on uses of Information Technology show that students in one Malaysian higher education institution are more satisfied and confident when they have instructor-led learning during their technology courses (Hong, 2002). Besides that, teacher education students are reported to depend on their teacher educator or their cooperating teacher at school when they want to perform their teaching tasks during their teaching practise at school (Hamid, Zolkifli, Rajuddin, Zakaria and Zahiri, 2007). Co-operating teachers are perceived as knowledge providers at school and they are expected to tell the students what to do. Thus, the socialisation process between cooperating teachers and the students are more inclined towards a one-way communication, rather than a collaborative approach that will give students the opportunity to conform with the professional attributes, not with the people who deliver the attributes.

The differences in professional socialisation processes between Western and Eastern countries are less influenced by the structure of curriculum. There are not many differences in the basic structure of the curriculum between Western and Eastern learners, which provide them with accessibility to the profession. Professional socialisation in higher education primarily involves two main components: academic (or clinical) and fieldwork components (practical). Fieldwork, mainly an internship, is part of the socialisation structure in professional education that offers students a chance to apply knowledge and skills and develop their attitude (Baretti, 2004).

Recently, the professional socialisation process of medical students in both West and Eastern contexts has been viewed as being similar. However, for teacher education programmes, there were slightly different cases where students in the West were found to be more flexible in terms of values, norms and their role definition of teachers (Kyriacou and Kunc, 2007; Levine-Rasky, 1998; McPherson, 2007; Rots et al., 2007). Students in the West are able to reflect on different teaching attributes, however, there is a barrier with students from the East. Students from the East are more tied to the cultural differences from the West and do not welcome reflection (Clark and Otaky, 2006).

Research suggests that Asian students commencing their higher education in Western countries are inclined to change their learning approach in line with their Western counterparts in the same university (Richardson, 2004). According to Richardson, the students show a positive attitude towards the changes, yet they lack the skills to do so as they are not trained to be reflective in their local education system. Furthermore, a considerable amount of recent studies try to challenge the traditional approach that differentiates Western and Eastern learners. Studies, particularly in Hong Kong, found that students who study in local higher education institutions also realise that the traditional approach of learning limits their development process as students (Gao and Trent, 2009). This implies that the interpretive approach helps to enhance students' learning.

In summary, the reviews highlight that different learning approaches will impact how students internalise the knowledge, skills and values pertaining to their chosen profession. The interpretive learning approach has many things to offer, particularly the opportunity to internalise the professional roles and dispositions, as the students have to understand the attributes of the profession if they want to become a member of the group. In contrast, the functionalist approach enables students to conform more to the people, rather than the professional attributes. Many studies in Western countries implicitly claim that student teachers undergo the interpretive learning process; nonetheless, the detailed processes have not been explicitly presented in the literature. Thus, it is difficult to identify whether the significant others become the "teacher", as imposed by the functionalist approach, or whether they are the socialising agents that facilitate the process of attitude development among student teachers.

2.7 Malaysian students and the interpretive approach to learning.

The previous subsection highlighted that the interpretive approach of learning gives more promising evidence on the way students can internalise the attributes of the profession, and at the same time proposed new attributes based on shared understanding among the members of the same group. This might help to enhance students' commitment towards future profession and construct a more positive social identity. Studies, although very limited, signal that higher education students in Malaysia are prepared to move from the restrictions of a functionalist approach in professional socialisation towards building a more flexible learning approach, thus making students have greater flexibility to learn and form the new attributes of the profession. Moving towards a new approach of learning does not mean that students will lose the essence of the profession, instead it offers more flexibility towards students' attitude development. It becomes a challenge to implement the interpretive approach, as Malaysian society is categorised as high in uncertainty avoidance (Hofstede, 1984). Societies with uncertainty avoidance show more emotion and intolerance of dissent because they feel threatened by change.

The assumptions of the transition from the traditional learning approach towards a more interpretive one are drawn from several national circumstances. The first one is associated with the educational reform implemented in Malaysia. Malaysia aims to be a developed country by 2020. In year 1999, the Vision of 2020 (a plan to build Malaysia into a developed country) was published; listing nine challenges to Malaysia becoming a developed country. The challenges are:

- 1. Establishing a united Malaysian nation made up of one Bangsa Malaysia (Malaysian nation);
- 2. Creating a psychologically liberated, secure and developed Malaysian society;
- 3. Fostering and developing a mature democratic society;
- 4. Establishing a fully moral and ethical society;
- 5. Establishing a mature, liberal and tolerant society;
- 6. Establishing a scientific and progressive society;
- 7. Establishing a fully caring society;
- 8. Ensuring an economically just society, in which there is fair and equitable distribution of wealth of the nation; and

9. Establishing a prosperous society with an economy that is fully competitive, dynamic, robust and resilient.

("Malaysia Prime Minister Department", n.d.)

Underpinning the expectation to achieve the challenges of Vision 2020 is the development of education at all levels in Malaysia (Lee, 1999). Students are encouraged to be remarkably independent in learning, rather that hoping that the knowledge, skills and values will be derived externally from teachers. The traditional paradigms, whereby teachers focus on transmitting knowledge and students simply adapt, have begun to be criticised (Pauline, 2013; Kasim, Aini and Furbish, 2010). The approach for life-long-learning that requires the participation of both learners and teachers has been voiced as an important shift in the education system.

Pauline (2013), among others, proposes that the education system in Malaysia has to create an opportunity for students to do self-reflection, rather than adapt the information and knowledge from teachers. If this is the case, there will be major changes in belief and practice, as traditionally teachers in Malaysia are seen not as facilitators, but a "fount of knowledge to be delivered" (Liu, 1998, p.5). Thus, this new approach will assist students with their learning of the attributes of the profession, and to conform to the attributes as a way to become a member of the profession.

Furthermore, studies extensively find that some of the students' beliefs have changed, and they responded positively to the new active role of students, which involves self-reflection, being active participants in the learning process, and implementing a constructivist approach in the learning process (Kasim and Aini, 2012). Kasim and colleague researched a group of teacher education students in one university in Malaysia. The trend of being active participants is promising, as in 1988, Zubir's study (Zubir, 1988) found that half of her participants, who were also teacher education students, were capable of learning on their own, but the dependency on teachers was high. Although there have been efforts to identify the shift towards an interpretive approach in students' learning process, most of the studies focus on the interpretive approach to learn the subject matter and skills and focus less on building the identity towards the profession. Only a few studies try to relate the interpretive approach in the

development of the students' professional socialisation. Among others are Ahmad, et al. (2012), Elias (2008) and Liau, Liau, Teoh and Liau. (2003).

Ahmad et al. (2012), for example, found that the Malaysian accounting students' motivation, perceived environment and involvement, reflect that the students are ready to make a move towards a more interpretive learning, as they showed a tendency to form a social group according to their interest. It is assumed that highly socialised students are more committed towards the profession that they wish to become, as they participate actively and increase their involvement in the classroom and out of class experiences; therefore, they enhance their knowledge-seeking and boost the internalisation of professional values (Elias, 2008). Teacher education students also agree that by having self-reflection they are able to solve problems that they encounter when they undergo their teaching practise in schools (Liau et al., 2003). However, previous studies lack discussion as to how the socialisation process takes place and the details are not clearly presented.

In summary, this subsection justified that the interpretive approach to learning has the potential to be implemented in Malaysia for the given reasons. Thus, this study will analyse to what extent teacher education students become active participants in the socialisation process in teacher preparation programmes.

2.8 Conclusion

This chapter justified the significance of social identity approaches in determining the students' identity development toward the teaching profession. Students' identity towards the teaching profession are viewed on how students possibly respond to the aspiration of the profession. Social identity (i.e. member of teaching profession) becomes salient when students are emotionally attached to the teaching aspiration, and this attachment brings the distinct meaning of their behaviour compared to their initial enrolment motives.

This subsection argues that in order to identify how the student teachers try to build commitment towards the profession and develop a more positive social identity towards the group that they belong to, then the way they socialise in higher education should be made explicit. There are evidences that the interpretive approach of socialisation helps to develop strong identifiers among the students. Strong identifiers are assumed to be active participants to adhere to the aspirations of the profession.

There are only a few studies of professional socialisation factors in the development of professional identity towards the profession. Of the few done, many of the studies are of medical, dental or nursing students. Investigation on the development of attitude among student teachers has been limited. Furthermore, the existing literature has restricted the scope of the studies, whether towards the traditional approach or interpretive approach. There are limited studies that aim to hear the professional socialisation approach from students' perspectives, whether they are undergoing functionalist or interpretive approaches of socialisation. This study challenges the dominance of a functionalist approach over the interpretive approach and identifies the influence of these changes.

The socialisation approach has helped to identify the extent to which students respond to social influence in determining their identity in a Malaysian context. From the literature, it is obvious that the current students are influenced by significant others in determining their identity. However, the reviews are restricted by a lack of information as to whether the student teachers perceive the significant others as people that they have to conform with, or as the socialising agents that will facilitate their identity development process.

CHAPTER THREE METHODOLOGY

3.1 Introduction

This chapter outlines the methodology and method used to conduct this study, and how they have guided the research strategy and selection of techniques to gather and analyse data related to the research questions. The chapter starts with the methodological rationale that underpins the research paradigm of this study. This includes the overview of the research methodology; an explanation of social constructionism epistemology that underpins the study and the interpretive framework that has shaped my social inquiry. The next subsection of the chapter discusses the research method and techniques and modes of analysing and interpreting data. The final subsection presents the issues of ethics and triangulation of the study.

3.2 Methodology

Researchers define methodology as a way or approach to conduct research in order to find answers for social phenomena (Barbour, 2014; Corbin and Strauss, 2008; Cresswell, 2013; Crotty, 1998). Some researchers, such as Cresswell (2013) and Silverman (2009), refer to methodology as a research paradigm. There are a variety of ways as to how researchers should conduct their studies, and there are different methodologies that help them to find the best way of acquiring knowledge about the world we live in (Denzin and Lincoln, 2011). Crotty (1998) suggests "different ways of viewing the world shape different ways of researching the world" (p.66).

Barbour (2014) further suggests that to enable researchers to answer the inquiry or research question, they will search for an approach or employ a paradigm that allows greater opportunities to understand the social phenomena being studied and its complexities.

As outlined in Chapter One, this study aims to explore the development of identity among student teachers, and the context is in Malaysia. In a bigger framework, the identity development process towards the profession suggests that it is important to identify how students express their perception of the teaching profession, and how the perception, particularly related to cognitive components, establish the teaching values and norms. As teaching values and norms are important factors in determining the social identity salience, exploring the factors will give the answer about social identity of teaching as perceived by the future members of the profession. The teaching values and norms will guide the students' behaviour thereafter. The students' cognitive definition on behaviour will explain their professional identity, which is the main construct in this study.

It is vital in this study to determine how teacher education students categorise themselves as future teachers according to the perceived values and norms and the role of social context in the development of identity towards the teaching profession. In summary, the students' efforts to develop their identity can be viewed as a continuous and complex process of determining the common characteristics (i.e. values and norms) of the teaching profession.

Correspondingly, this study also aims to explore the effect of social influence in the process of identity development. Chapter One has highlighted that the perception of teacher education students towards the teaching profession in Malaysia has not actively been researched, despite its status as a noble profession perceived by society, and the concern about the motive of current prospective teachers that choose to teach. Thus, the aims for this study suit the qualitative paradigm that focuses on the exploration and investigation of the social phenomena (Sarantakos, 2005). The next subsection explains further the qualitative paradigm.

3.2.1 Qualitative paradigm perspective of the methodology

The decision to employ the qualitative paradigm emerged as I intend to discover the evaluation that students have about values and norms of the teaching profession and their experiences in teacher education programmes as a platform to become a teacher.

As stated in Chapter One, the opportunity to get a job in teaching compares highly to other professions, regardless of the new policy of teacher placements as there is an increasing number of children entering primary and secondary education every year. Thus, I assume that employing a qualitative research paradigm, particularly through the stories, will give me an explanation of the students' perception of social identity related to their future profession.

Denzin and Lincoln (2011) explain that qualitative research "is a situated activity that locates the observer in the world [...] consists of a set of interpretive, material practices that make the world visible" (p. 3). They further explain, "qualitative researchers study things in their natural settings, attempting to make sense of or interpret phenomena in terms of the meanings people bring to them" (p.3). Thus, it is assumed that exploring and investigating teacher education students' socialisation experiences and social realities can reveal their lived experiences.

The term qualitative, as stressed by scholars, refers to processes that are not involved in the experiment and statistically significant relationships between variables (Barbour, 2014; Denzin and Lincoln, 2011; Miles and Huberman, 1994); instead looking at the explanation provided by the subjects involved in the research (Barbour, 2014). Therefore, the process of qualitative research suggests that there is an intimate relationship between the researcher and the object of study (Denzin and Lincoln, 2011). Consistent with the essence of the qualitative paradigm, current studies explore and investigate teacher education students' identity development towards the teaching profession and the process of identifying the social identity salience among students.

This includes gathering the views of the participants' world in order to understand their lived experiences, and the resulting identity, based on their perception of the teaching profession. ten Dam and Bloom (2006) argue that the dynamic nature of student teachers' identity formation is better explained by the students' reflection on the process of becoming a teacher. Furthermore, Antonek, McCormick and Donato (2007) suggest that the qualitative research paradigm allows researchers to get wider information regarding teacher education students' identity development. Antonek et al. (2007) use a student's online reflective journal as a research tool and find that their participants continuously develop their identity via feedback from the online user. Similarly,

Walling and Lewis (2000) appreciate the qualitative paradigm on identity development because this paradigm offers in-depth information about the process from the students' point of view.

The present study employs a qualitative research paradigm; specifically, the study is informed by the social constructionism epistemology that calls for an interpretive framework to social inquiry. The subsequent subsection outlines how a social constructionist epistemological perspective informs the research investigation. The explanation of the interpretive framework follows the explanation of social constructionism.

3.2.2 Social constructionism

This subsection outlines my assumption about the nature and characteristics of human knowledge. I believe that knowledge associated with social behaviour is intentionally built through lived experiences and social constructions of the world. This assumption guides my thinking as to how to get the answer for the social phenomena studied. Social constructionist perspectives outline that processes of knowledge formation are based on social interactions, where meanings are produced and reproduced, rather than constructed within the individual (Burr, 2003).

Social constructionist perspectives underline that our understanding of the world is based on the premise that there is no such thing as an objective fact (Barbour, 2014; Burr, 2003), and that knowledge is the product of continuous sense-making and interpretation that individuals use to interpret their social environment (Crotty, 1998). Within these perspectives, the participants are free to make their own assumptions about the object of study. Social constructionism, therefore, suggests that knowledge is best acquired by the induction or a bottom-up process where there is no predetermined hypotheses that need to be tested (Ormston, Spencer, Bernard and Snape, 2013).

In this study, I attempt to explore the student teachers' identity towards the teaching profession. Although the teaching profession in Malaysia is perceived as a noble profession, however, sociological works on profession have never regarded teaching as an established profession. Additionally, consistent with the economic and technology

development, some teacher education applicants have a different interpretation of the teaching profession (Ahmad, 2003; Harun, 2006). Some of them perceive that teaching is the least preferred choice of profession. As stated in Chapter One, the development of other industries in Malaysia, which offers opportunities for employment in private and industrial sectors, may contribute to this perception. The negative issues such as teachers' commitment have challenged the social identity perceived of teachers in the previous decade, hence this might impact teachers' behaviour considerably, as their intended behaviour is determined by the group they belong to. Thus, the need to identify the students' perception on teaching where the norms and values will depend solely on students and the understanding about their own world.

As a researcher, I am extremely aware that the students' identity development is highly subjective and can be assessed from different perspectives. As highlighted in Chapter Two, the students' identity, which are the main social inquiry in this study, are assumed to develop through student interaction with an object around them, including their environment. As the students' identity is assumed to be derived from the participation in the (teaching) group, therefore, identifying the groups values and norms are of paramount importance as the students' behaviour are predicted to follow the group aspiration (in terms of values and norms). In this sense, the students may refer to the accepted teaching traits, differentiate teaching with other professions, or issues surrounding them in order to identify the values and norms of teaching. The students are expected to construct and reconstruct their identity mostly during their teacher education programme, which provides them with the opportunity to access various information. The social constructionist perspective also allows the students to freely decide their information gathering approach as there are multiple social realities guiding their experiences as students. This premise outlines that students use the process of gathering information and knowledge to allow them to construct their own meaning, rather than accept the predetermined measure of their social world.

Correspondingly, social constructionist perspectives also emphasise the significance of the interaction between the researcher and the participants in the construction of meanings and interpretations (Crotty, 1998; Denzin and Lincoln, 2011; Lincoln and Guba, 1985). The interactions enable the researcher to analyse the participants' construction of meaning in their social world, through their experiences and the way

they gather information. Scholars argue that knowledge (from experiences and accessible information) is not value-free, and there are numerous interpretations of realities (Denzin and Lincoln, 2011).

Furthermore, Charmaz (2000) illustrates that "social reality does not exist independently of human action" (p.521), hence providing the ontological stance for this study. It is assumed that participants are able to interpret different meanings of their social world because there are multiple social realities that are attached to their everyday life. Boeije (2009) posits "human beings attach meanings to their social reality and that as a result; human action should be considered meaningful" (p.6). Thus, in this study, the categorisation of students into different groups provides the researcher with the assumptions that the students have multiple social realities and construct different meanings from their experiences, thus shaping their identity accordingly.

3.2.3 The interpretive framework

Based on the social constructionist epistemological perspective, this study adopts an interpretive framework of the study of teacher education students. Max Weber's views have shaped the interpretive approach in social research (Crotty, 1998). According to Weber (citied in Crotty, 1998), the understanding of why people do something will help to explain their actions in more detail. This is consistent with the main research questions addressed in this study: an interpretive framework allows the researcher to identify the 'why' aspect, which is the evaluation of the profession to be used among other researchers, to use attitude as a construct that will affect something else, such as behaviour.

In addition, Crotty (1998) posits that the important aspect in an interpretive framework is the concept of 'understanding' rather than 'explaining' the social phenomena. Denzin (1989) explains that the process of understanding involves "identifying different definitions of the problem being evaluated through the use of personal experience stories and thick description of lived experiences" (p.11). Interpretive framework works on the basis whereby the researcher has to understand the individuals' subjective experience (Bryman, 2012) and how the participants attach meanings to certain phenomena. To find meaning in a particular situation requires understanding of the

action, and as stated by Denzin (1989), by "identifying different definitions" or simply the effort to interpret a specific action (Schwandt, 2003).

It is necessary to understand that the interpretations that emerge from interpretive research "are constructed [...] there is no single interpretive truth" (Lincoln and Denzin, 2003, p. 37). Therefore, the analysis of this study reflects my attempts to make sense of the interview data in order to understand the phenomena being researched. This is to explore the participants' subjective experiences and perspectives, rather than generalise the participants' perspectives according to dominant literatures. Thus, there may be unexpected realities discovered in the process of interpretation.

In the following subsection, I acknowledge the position as a researcher and how it affects my epistemological and ontological beliefs. Following this, I lay out in detail the steps taken in the data collection and analysis process in Subsection 3.3.

3.3 Participants and methods of data collection

This subsection discusses the primary method of data collection in this study. Generally, this study uses qualitative data from interviews to understand and explain the development of teacher education students' attitudes towards the teaching profession.

3.3.1 Participants

The sources of information for this study were teacher education students in the biggest secondary teacher training institution in Malaysia. The institution is designated as a 'University of Education' and prior to 2011, it exclusively offered degrees in education, with a variety of areas of concentration that enabled graduates to apply for positions as a secondary school teacher after graduation. However, from 2011, this institution started to offer other non-education degrees as a way to optimise the expertise of the academic staff, particularly those who have academic certificates in non-education areas. In the interim, the academic staff with non-education backgrounds in this institution focus on the teaching of content knowledge while their counterparts with education backgrounds use their expertise to teach teaching methods.

In this institution, the students are known as student teachers as the majority of them choose to apply for the teaching profession after graduation. In Malaysia, teachers are referred to as 'guru', and are given a distinctive title, which is 'Cikgu'. The teacher education students in this institution are often called 'Cikgu' (teacher) by the society even though they are not yet teachers. An important point to note is that the majority of participants are Malays, as the Malays dominate the teaching profession in Malaysia. Additionally, the number of Malay students in public higher education outnumbers other ethnicities due to the Malaysian Federal Constitution 1953, which implements the affirmative action policies that benefit the indigenous groups, including the Malays. Other ethnicities involved in this study were indigenous ethnic⁶ from East Malaysia (10), Chinese (2) and Indian (1).

The selection of the sample is based on the purposive sampling method. Purposive sampling involves the selection of participants based on important characteristics under study (Bryman, 2012). In the first place, the samples were selected from teacher education students in a university that offers a wide range of degrees in education. The inclusion characteristics, as suggested by Daniel (2012) and Sarantakos (2005), were taken into account when the decision on the sample had been made.

The characteristics, which refer to the essential attributes pertaining to the study, were considered in order to assist a detailed exploration and understanding of the central themes and questions regarding the phenomena being studied. Thus, the students' route to teacher education programmes (undergraduate level and post undergraduate level) and year of study (first year or fourth year) were included. These characteristics enabled the provision of an explanation as to whether there will be differences in identity development according to the students' characteristics and experiences, as the basis of their perception of values and norms in the teaching profession.

_

⁶ In East Malaysia (State of Sabah and Sarawak), there are indigenous ethnic groups practicing their own unique cultures and heritages. Among the largest groups are Dayak, Iban, Bidayuh and Kadazan.

The composition of each group was homogenous in terms of their route to the teacher education programme and their year of study. The number of participants are summarised in the following table :

Members of groups	Group 1	Group 2	Group 3
	(Students in	(Mature students,	(First year
	teaching practise)	career change,	students)
		Semester 1)	
Number	54	14	12

Table 3.1: Number of research participants

In detail, there were three different groups of participants that contributed to this study. Group 1 was the primary focus for this study as they are nearing their graduation. Group 2 was chosen with the assumption that they have pursued individual mobility strategy and changed occupational groups for some reason. Group 3 was chosen as they are the first batch of students who entered the teacher education programme with no guarantee to be offered a place as a teacher after graduation (as highlighted in Subsection 1.9). This group was selected in order to explore the differences in their perception of values and norms (if any) as a result of the new teaching recruitment policy.

The first group consisted of 54 teacher education students undergoing their teaching practice (practicum) at secondary schools as part of their training process. Teaching practice serves as the culminating requirement in completing the teacher preparation programmes and teacher education students undergo their practice during their seventh semester (out of eight) (Universiti Pendidikan Sultan Idris, 2011). During their teaching practice, teacher education students begin to test their position as classroom teachers based on what had they learn in the teacher education programmes, as well as their informal learning from the environment (Schepens, Alterman and Van Keer, 2007).

All participants had their teaching practice in the state of Perak. Perak is situated in the west of Malaysia. The selection of this state was made because at the time of conducting the study, Perak had the biggest number of teacher education students

undergoing their teaching practice; hence, it provided a fair distribution of different programmes of study. I chose eight secondary schools where many students were undergoing their teaching practise. The schools are located in the rural area. Data on the participants were obtained from the university's practicum unit, and letters were sent to prospective participants to invite them for a focus group session (detail of the focus group session is discussed in the subsequent section). A letter was also sent to the school administrator to ask for permission to conduct the study on their premises. Participants who agreed to participate informed the researcher via text messages. The distribution of participants was as follows.

Area/School	Number of participants
1	7
2	3
1	7
2	11
3	5
3	3
4	9
4	9

Table 3.2. Distribution of participants undergoing their teaching practice

At the end of each session, the researcher asked the participants, who volunteered to be contacted after the session, for any clarification and to leave their contact number. Ten of the participants agreed to supply additional information if needed; however, only four of them volunteered to participate in an individual interview.

Moreover, participants in the second group consisted of 14 mature and career changed teacher education students. They came from diverse occupational backgrounds before enrolling in the teacher education programme. Some of them were administrators or trainers, and some had been in school as untrained or substitute teachers. Untrained or substitute teachers teach a school class when the regular teacher is unavailable because of illness, maternity leave or other reasons. They had been in school for a short period

of time. This group consists of first (out of three) semester students. This group of students will undergo their teaching practice during their third semester.

Finally, the third group of participants was included in the study. This group consists of twelve teacher education students in their first (out of eight) semester of the teacher education programme. Their route of entry into the teacher education programme is similar with the first group. The difference between the participants in the third group compared to the first group is that this group consists of students enrolled in the 2011 session, and they are the first batch of student teachers who are not guaranteed placement after graduating from the teacher preparation programme.

3.3.2 Methods of data collection

Data was collected using a narrative method. Narrative methods concern personal narratives or stories (Czarniawska-Joerges, 1998) and allow "participants to recall what has happened, put the experience into sequence and find possible explanations" (Jovchelovitch and Bauer, 2000: 58). The experience that the individuals convey through the story is able to determine their meaning towards certain phenomena (Crotty, 1998). This method is appropriate for exploring the students' perceptions of teaching values and norms, as it is predicted that the students' perception is built from various dimensions and have different structures and functions, as described in Chapter Two. The students had the opportunity to tell their own story directly as they are consciously engaged in the socialisation process, particularly during the teacher education programme as well as from the life experiences.

There is a significant amount of literature on student identity formation that has employed narrative as a data collection method (Alsup, 2003; Estola, 2003; Luehmann, 2007). Alsup (2003) finds that the student teachers' storytelling makes them aware that the preconception of the teaching profession is not suited to their current beliefs about the profession; therefore, this gives them an opportunity to develop themselves consistently with their beliefs and characteristics, particularly if it involves the values and norms of the profession. Furthermore, Beauchamp and Thomas (2009), in their review to strengthen teacher education programmes, also give credit to student teachers' stories that "to provide opportunities for exploring and revealing aspects of the

teachers' self' (p. 181). Thus, narrative as a data collection method is able to provide a rich and thick description of the students' perception of teaching values and norms, and the identity, as it provides a story of their experiences in teacher education programmes.

In addition to the participants' stories, data was also collected through observation. The main observations included the participants' non-verbal signals such as body language, facial expressions and intonations of voices, which convey a story. The observations captured the emotions and important points of the participants' stories. The observations were undertaken through the interpretation of a video captured in several interviews.

3.3.3 Research tools

The participants relate their own story directly to the researcher through two types of interview; (focus) group and individual interviews.

Focus group discussions

The first tool for data collection was focus group discussions among the three groups of teacher education students. The interviews provided rich and detailed descriptions. The first question asked was about the way in which the students are expected to see himself or herself as a member of the teaching profession. For the first semester students (Group 2 and 3), they are assumed to visualise their teaching image. For the students who were undergoing their teaching practice (Group 1), their narrative about themselves as a member of the teaching profession may come from their learning at the higher education institution and how they transfer learning into concrete situations, which is the teaching practice. The question was "who I am as a teacher". Through this, the essential prerequisites for understanding the theme are by referring to the dominant traditions of sociology of the profession.

The next question asked was to capture the dynamics that the students developed and act as teachers. The question was "share the journey from the decision to enrol in the teacher education programme until today". The participants were requested to share their feelings and experiences before and during their enrolment as teacher education students. A few questions were asked to establish a trust and comfort level within the

group, and the remaining questions served to guide the discussion towards the attitude towards the teaching profession. The participants were very comfortable in relating their stories and sharing experiences, and treated me as one of their group members as sometimes they used the phrase "as you are aware" or "you might experience the same" in their narrative.

The advantages of conducting focus group discussions are that participants have the opportunity to learn from each other and develop ideas together as a group (Jackson, 2003). This is consistent with the argument that the social identity approach underpins this study. Sarantakos (2005) further explains that a group environment encourages discussion and increases the motivation to address critical issues. Self-representation of the participants are minimised by creating a permissive environment in the group that encourages participants to share perceptions and points of view without pressuring them to vote or reach a consensus. Focus group discussions also encourage group interactions to produce data and insights that would be less accessible without the interaction found in a group because every participant has a different experience and background (Morgan, 1997). The focus group sessions were non-directive, allowing the participants to give voice to their experiences, which in turn fosters the emergence of themes from the participants.

The participants were asked to take part in in-group interaction sessions of about an hour in duration. Scholars suggest that the ideal amount of people in one focus group is six to ten people (Morgan, 1997; Vaughn, Schumm and Sinagub, 1996). Additionally, the New York State Teacher Centres⁷ suggest conducting three to five focus groups to gather information about the research project. All sessions were audiotaped, and most of the sessions were videotaped. Some were not videotaped due to a technical problem that could not be avoided. During the discussion, reflective notes were taken to record the activity and action of participants, which could not be captured by the video recorder.

For the first group, all the teacher education students attached to the selected schools were invited to the focus group discussion although not all were present due to teaching

_

⁷ http://www.programevaluation.org/focusgroups.htm

commitments. The group interview was held after the end of a morning school session; however, some schools have two teaching sessions due to a large number of school pupils. Most interviews were held in the schools' meeting rooms, but some used empty classrooms as a meeting room that were also used by the school's management team at the same time. Other than that, one focus group was held in the students' flat, as all of them rented the same property while undertaking their teaching practice.

For the second group, the participants were selected among their batch, and I took into consideration their previous occupations, their cumulative grades in their previous undergraduate credential programmes and their ages. The variables were selected as it helped to link the experiences and the anticipated identity, as reviewed in the literature. The focus group sessions were conducted in an institution's meeting room. For the third group, which was the first year students, I selected them based on their entry-level grades and their qualifications entering the teacher education programme, whether they were fresh from school or held a diploma certificate from another higher learning institution. The focus group sessions were carried out in a meeting room and an empty seminar room.

The focus group sessions were carried out in the Malay language. The participants were asked to comfortably sit around the table and the researcher started the group discussion by introducing herself to the group. Then the researcher started the discussion by asking the questions. The participants freely took turns to speak, and most of the participants had a chance to speak more than once. However, there was a participant who chose to be silent during the session after she mentioned that "I do not want to be a teacher". Further discussions on this issue are presented in Chapter Four.

The schedule and details of the focus group sessions are provided in Table 3.3

Area/School/Class	Number of	Interview venue	Duration
	participants		(hour/minute/sec)
1 (BJ)	7	Classroom	1.10.44
2 (HD)	3	Conference room	1.02.26
1 (SP)	7	Meeting room	1.05.20
2 (AB)	11	Library	1.18.53
3 (TIWA)	5	Meeting room	1.42.55
3 (BK)	3	Meeting room	0.57.09
4 (SK)	9	Students' flat	1.49.06
4 (TP)	9	Meeting room	1.09.00
DPLI 1	7	Meeting room	1.34.50
DPLI 2	7	Meeting room	1.03.05
FY 1	6	Meeting room	0.43.33
FY 2	6	Seminar room	0.54.51

Table 3.3. Schedule of focus group sessions⁸

Individual interview

Individual interviews were conducted over the phone. It was a semi-structured interview. The interview was conducted six months after completing the group interview. The researcher used the six-month interval between the group and individual interview to do an initial analysis and determine the interview questions. The purpose of the interview was to gather more in-depth data on the participants' views on the teaching profession. Participants were asked about their self- perception, a teacher's life and experience to become a teacher. When prompted by relevant questions, the participants gave meaning to their experiences, thus providing an interpretation of the phenomena under study. There were four participants involved in the individual interviews and all participants were from the first group. Similar to the group interview, each individual interview was recorded, transcribed and analysed. The transcriptions were sent back to the participants via email in order to make sure that their stories were transcribed accurately.

_

⁸ DPLI refers to participants in the second group and FY refers to participants in the third group.

The advantage of the individual and in-depth interview is that it empowers the researcher with the freedom of questioning (Denzin and Lincoln, 2011). Thus, the individual interview is considered an appropriate procedure to obtain richness in the data through a detailed and frank conversation between the researcher and the research participants.

Thus, data collected for this study consists of the participants' stories from the interview, and the video footage recorded during the group interview.

3.4 Data analysis

This study embraces a constructionist approach that affirms meaning as being socially constructed rather than given. This study adopted a thematic analysis (Attride-Stirling, 2001) to analyse data from the interview. The thematic analysis used in this study is inductively derived from the meaning extracted from the participants' stories. The strategy used in the analysis was informed (but not constrained) by the literature, building an emergent framework from the data. Attention is given to the story of the student teachers in constructing their identity during the teacher education programmes. Braun and Clarke (2006) define thematic analysis as a "method for identifying, analysing and reporting patterns or themes within data" (p. 79). The themes describe and organize possible observations and interpret aspects of the phenomenon found in the information gathered from the participants' stories (Boyatzis, 1998). With the narrative method of data collection, the thematic analysis yields a rich and detailed amount of data.

The qualitative data analysis software, NVivo 9, was used to help organise and manage the data and its analysis. The interviews were all recorded and transcribed verbatim, using Malay language by the researcher, in order to facilitate immersion in the data. The interviews in Malay language that will be used as a quote were translated as best as possible so that the original intentions were not lost. The translation process is guided by a native speaker (Malay) who teaches the English subject at school. There were cases where words or phrases in the Malay language did not have direct equivalents in

the English language. For these cases, the researcher had to paraphrase to convey the precise meaning of the original Malay content.

After the transcribing process, all transcriptions were uploading to the NVivo 9 package and sixteen data sets (twelve data sets from the focus group sessions and four data sets from the individual interviews) were created.

The coding process started with the transcription from the first group, which was the participants undergoing teaching practice. Codes are identified through "semantic content or latent" (Braun and Clarke, 2006, p. 88) that may appear interesting to the researcher and help in answering the research questions. Researchers search for a meaning (Miles and Humberman, 1994) rather than focusing solely on words or phrases that appear in the text to develop initial codes. The data was coded according to the participants' stories where each participant was given a pseudonym to ensure confidentiality and to protect their anonymity.

Early coding was descriptive and involved open and line by line coding. I employed an ongoing iterative process, identified the category of students according to their initial perception towards the teaching profession, sought patterns in the process of becoming and unbecoming a teacher, and tested whether there were differences among the sample group. The development of codes was completely based on salient issues that arise from the transcript itself, and most of the code labels were taken from the participants' own words. Codes were used to identify the initial phenomena and produce a list of themes from the data set. Constant comparisons from previous codes helped to further develop new codes from the transcription. Constant comparison means that every time a passage of text was selected for coding, a comparison was made between the new code and all those passages that were already being coded that way (Corbin and Strauss, 2008). This procedure helped to create significant patterns in the data. From the three groups of participants, 118 codes altogether emerged from all data sets, and the list of codes included behaviours, specific acts, events, activities, strategies, meanings, relationships, constraints, settings, and metaphors.

Then, the exploration of the themes began by setting up groups or clusters for the codes. Similar codes were grouped together to make it meaningful for further exploration. Identification issues that were discussed by the participants facilitated this process. There were 48 issues identified, and data sets relating to these issues were grouped together to facilitate the process of identifying themes. The data sets were re-read within the context of the codes under which they had been grouped together. This step allowed the researcher to identify the underlying patterns and structures. The data sets were arranged by assembling them into similar and coherent groupings and were named. This process refers to the development of the basic theme (Attride-Stirling, 2001). The data set in a basic theme then were re-read again in order to reduce data and create more specific themes, termed as a global theme (Attride-Stirling, 2001). The process produced a web-like diagram that summarised the main theme evident in the data. The table below shows the example of the process of grouping together the issues, basic themes, specific themes and the research questions to be answered:

				Possibility to
Issue	Basic themes	Specific themes	No. of quote	answer selected
				research questions
Enrol	Motive	Parents and	43	Motive
		siblings		• Group
				norms and
				values
Learning	Play games	Knowledge	14	• Group
				norms and
				values
				• Identity

Table 3.4 : An extract of an analysis process

Finally, five main themes were identified from the data set reading. Three themes represented the social identity salience and the others represented strategies the students pursued during the socialisation process that may explain their social identity. Therefore, the process of creating both the basic themes and more specific themes helped in data reduction (Miles and Humberman, 1994) and resulted in important themes from the overall data; it involved working inwards starting from the basic themes to the global themes.

3.5 Ethics, reliability and validity

3.5.1 Ethics

Participation in this study was voluntary. Letters were sent to the prospective participants to invite them for this study, indicating the aims and study objectives. The participants who agreed to participate sent their consent to participate via text message to the researcher. The participants were given an explanation of the study that was to be carried out, and they were also given an opportunity to ask the researcher about the interview procedure. In addition, the participants also were informed that at any time, they could withdraw from the interview session without any prejudice and their choice to contribute in the sessions would not affect their academic result in any way.

During the interview, all the communication was audiotaped and some of the interview sessions were videotaped for research purposes. The data management process including the anonymity of participants was handled with care. All the transcripts were labelled with abbreviated names that could only be understood by the researcher and were kept safely and locked in the researcher's filing cabinet. Due to the low proportion of male participants in the sample, all participants were referred as female to minimise the potential of their identification.

3.5.2 Reliability and validity in qualitative research

Trustworthiness and credibility of the explanation given in the study description are important for reliability and validity in qualitative research (Janesick, 1998). For the purpose of this study, the researcher followed the validity procedures suggested by Cresswell and Miller (2000). The first procedure is the disconfirming evidence. This process is similar to the constant comparison described in Section 3.4. It is the process whereby the researcher establishes preliminary themes or categories and checks for consistency in the data set. The NVivo 9 package facilitates this procedure by offering a query menu that allows the researcher to check for word consistency and word count (Bazeley, 2013). The words that mostly appear in the query menu are compared to the preliminary themes and categories established by the researcher.

Secondly, Cresswell and Miller (2000) also suggest that researchers describe the setting, the participants and the themes of the qualitative study in rich detail so that readers can 'feel' the situation experienced by the researcher. According to the authors, thick and rich descriptions allow the readers to "understand that the account is credible" (p. 129) and may help the reader to apply the findings in another similar context.

Additionally, Cresswell and Miller (2000) also suggested that using different techniques of data collection and comparing the results of different groups towards specific issues highlighted in the study will maintain the validity of this study. As highlighted in Subsection 3.4.2, a different method of data collection has been applied to all focus group participants, which is video capture. These videos capture the body language and the intonation used to support the students' narratives.

3.6 Conclusion

In this chapter, I have presented a theoretical perspective of this study, and outlined the methodology that guided data collection and analysis. The qualitative method allowed me to obtain a rich description of the subjective experiences of the teacher education students on their perceptions towards the teaching profession and the identity development. This chapter also outlined the research process, from sample selection criteria, data collection and analysis. Chapters Four and Five, which follow, presents the findings from the thematic analysis.

CHAPTER FOUR DISCUSSION OF FINDINGS

4.0 Introduction

Chapter Four and Five focus on reporting and discussing the findings of the thematic analysis as outlined in Chapter Three. To recap, the aim of this study is to identify students' self-definition on teaching by identifying their motives to enter a teacher education programme and their image of self. Based on the premise that a teacher preparation programme which offers early socialisation towards the profession has a significant impact on student's identity formation, this study explores the social factors that may contribute to the process of identity acquisition during the student's commencement in higher education. Students developing or declining the engagement with the teaching profession are identified by contrasting the initial motive that students have towards the profession at the time they decide to choose the teacher preparation programme with the identity that they currently hold at the time this study was conducted.

This chapter's focus is to answer the first research question; what influenced the students decision to choose the teaching profession. This chapter is also structured in a way to answer the second research question; what is the identity that the students bring to the teacher education programme. Subsequently, Chapter Five focuses on the student's socialisation process in the teacher preparation programme. The student's experiences during the socialisation process provide input to answer the third and fourth research questions; how social factors affect the development, or declining of student's identity towards teaching during commencement at the teacher education programme? It is also highlighted how the students try to achieve a positive social identity, and how they pursue different strategies to achieve that purpose. And finally, what is the professional identity that the students imagine of themselves. It is important to note that although the analysis and discussions in Chapter Four are related to the student's

socialisation process, this chapter focuses to answer only the three research questions as indicated above.

4.1 Important themes that describe student teacher's identity towards teaching

Important themes that capture profound issues in relation to answer two questions in this study, research question 1 and 2 are derived from the student's expressions on the reason why they chose a teacher preparation programme that primarily trains students to become teachers. The reason to enrol in a teacher preparation programme gives input of a student's knowledge and beliefs of teaching, the emotional reaction regarding teaching, and behavioural intentions towards teaching. To answer this question, the response from all three groups were taken into account. There are two themes that describe the student's evaluation towards teaching which emerged from the data, labelled as favourable and unfavourable motives.

The favourable and unfavourable motives have been interpreted from the student's expression of "Yes, I want to be a teacher", "I do not know", or "Maybe not". The favourable and unfavourable motives towards teaching are consistent with the earliest study indicated in Subsection 1.1 that suggests a favourable motive can be defined as student's intention to approach the given profession. The students who are in the first category, "Yes, I want to be a teacher", possess favourable motive towards teaching, and narrate their image of teaching together with their narrative. Meanwhile, their peer colleagues who are in the second and third categories possess the unfavourable motive, hence no identity or intended behaviour can be explained.

Furthermore, the student's early motive towards the profession is derived from their expressions on "who I am as a (future) teacher" in order to explore why they still commenced in the teacher preparation programme at the time this study was conducted. It is apparent from the data that during their commencement in the teacher preparation programme, the students were able to self-categorise themselves as future teachers. The process of self-categorisation will be detailed in Chapter Five; the thematic analysis also reveals that the students narrated both favourable and unfavourable motives and positive social identity at this stage. However, the favourable and unfavourable motives

are described in three different categories; general evaluation towards the teaching profession, professional relationship, and understanding of learning.

The discussions in this chapter are divided into two main subsections. The first subsection describes the identity that the student's hold at the time they decide to enrol in the teacher preparation programme. This identity emerged from how the students perceived their image as teachers while narrating their initial motive to enrol in the teacher education programme. Subsequently, the second subsection of this chapter interprets the student's expressions on their experiences during their commencement in the teacher preparation programme. The experiences explicitly describe a student's self-definition on teaching; consequently, explaining how they try to fit in to the attributes of the teaching profession. The conclusion of this chapter sums up the relationship between the student's initial self-definition towards the teaching profession, and the identity they currently hold about teaching.

4.2 Students' reason for entering teacher education programme

This subsection presents the findings and possible explanations regarding the students' reasons for entering a teacher education programme and their initial professional identity. It should be acknowledged that teacher education students in Malaysia have gone through comprehensive objective assessments for admission including screening on academic qualification, personality, career interest, value integrity, and emotional intelligence as outlined in Subsection 1.9. The students also undergo an admission interview that is purposely conducted to meet the applicants behind their application. In general, the students involved in this study had passed the minimum qualifications to embark on a teacher education programme that aims to train them to become teachers.

This implies that from the objective assessments, the students have teacher-related personalities and values, interest to pursue a career as a teacher, the ability to manage their own and other people's emotions, along with adequate academic qualifications. In social identity terms, the students possess more personal identity salience as they define themselves in terms of their personality, not in terms of their group membership. However, as argued in Subsection 1.9, the test is not sufficient to determine the

student's level of engagement with the profession. The qualitative paradigm employed in this study has disassembled more details about the teacher education student's level of engagement with teaching and its relation to their identity development.

As indicated earlier in this chapter, the results from the thematic analysis revealed three initial motives towards the teaching profession; "Yes, I want to be a teacher", "I do not know", and "Maybe not". Regardless of the groups of students (i.e. students in teaching practise, career change students, and first year students), the main finding from this study discovered that 87.5% of the students chose the teacher education programme because they aim for a teaching certificate which enables them to apply for a post as a secondary teacher after graduation. The rest of the students (12.5%) or 10 students had no clear direction after completing their teacher education programme. Out of 10 students, 9 of them are undergoing their teaching practise at different schools and one student is a career change student. As indicated in Subsection 1.9, the minimum requirement to apply for a secondary teacher position in Malaysia is to have a degree or graduate certificate in teaching. Students who enrolled before the academic year of 2011 will have the opportunity after graduation (Mokshein et al., 2009).

Thus, the enrolment in the teacher education programme means that the students are 'in the bag' of the teaching profession. However, this privilege was abolished in year 2011 when the Ministry of Education started to introduce a new policy in teacher recruitment whereby teacher education students have to apply for the post and are carefully selected, and only competent graduates will be chosen (Malaysia Prime Minister Department, 2010). This means that teacher education students who enrolled in 2011 are not automatically guaranteed a placement at schools after graduation. This policy applies only for the first year students involved in this study.

Table 4.1 presents the summary of themes from the thematic analysis with regard to the students' reason to embark on a teacher education programme and early identification with regard to become a group in the teaching profession.

Fa	actors	No. of response			
	Details		I do not know	Maybe not	
		be a teacher			
Individual	Person-job fit	/	/	/	
attribute		(10)	(2)	(1)	
Influences	Parent/siblings	/	/	/	
from		(15)	(3)	(2)	
significant	Former teachers	/		/	
others		(5)		(1)	
	Friends	/	/	/	
		(2)	(1)	(2)	
Former life	Experiences in	/	/	/	
experiences	society	(4)	(1)	(2)	
	Informal teaching	/		/	
	experiences	(5)		(1)	
	Positive/negative	/	/		
	experiences as	(6)	(2)		
	students				
Vocational	Always wanted to	/			
reasons	teach	(4)			
	Reward	/	/		
		(8)	(3)		
Others	Mistakenly chose		/		
	academic		(3)		
	programme				

Table 4.1 : Students' motives to enter teacher education programme

_

⁹ There are students who narrated more than one motive

In general, the result from the thematic analysis suggests that regardless of the reason to enrol in a teacher preparation programme, all students shared a common aim for enrolment, which is to get the credential or teaching certificate. Nonetheless, as presented in Table 4.1, the students' initial motive varies. There are students who intend to use the certificate to apply for a teaching post, students who are not sure, and students who intend not to use the academic credential for teaching purposes. Students who want to teach believe that their motive is influenced by personal characteristics, influences from others, former life experiences and vocational reasons.

The students who are undecided in teaching believe that their decision is influenced by the social environment. Ten students who clearly withdrew from the early group membership described that they are in a teacher preparation programme because of a technical mistake when choosing an academic major in higher education as well as influences from significant others particularly parents and former teachers. As students who successfully enrolled in the teacher preparation programme after undergoing the objective assessment, they are predicted as having the criteria to become a teacher as assessed in the MEdSI inventory (refer to Subsection 1.9); nonetheless, for reasons that will be discussed in the following paragraph, the students favour other professions rather than teaching.

The thematic analysis also provides evidence that identity formation is associated with both personal characteristics and social factors. Thirteen of the students claimed that they are attracted to teaching as they believe that teaching is congruent with their personality; this explains that identity is grounded in personal individuality such as interest, traits and personality. However, as stated in Subsection 2.5, student's individuality also stems from a long term socialisation process (Petty, Tormala, Brinol and Jarvis, 2006; Rudman, 2004) which involves long term socialisation. It is apparent from the students' expressions that their interest, traits and personality are influenced by the social environment as these are developed from observations and experiences. Furthermore, model influences, former life experiences and vocational reasons suggest that significant others have a significant impact on the students' motive. It is apparent from the data that family members have a significant influence on the student's decision to choose teaching. Besides that, vocational reasons in particular involves the student's interpretation that their identity is grounded in the group membership as they mentioned

that they are attracted to become a member of the teaching profession because of the professional attributes of teaching. This categorisation will be described further in Chapter Five.

The following subsections describe how identity is affected by both factors; student's individuality and social factors.

4.2.1 Compatibility between students' characteristics and teaching attributes

Holland (1985) suggests that professional decision making that is in line with an individual's personality, skills and interest are able to ensure that people stay longer in the profession they undertake. Holland suggests different dimensions of professional activities where each dimension has a different requirement and will suit certain individual attributes. The closer the compatibility between the attribute of a person and the requirement needed for a profession, the more likely that the person is successful in the field.

In detail, the trait theory such as Parson (1909), cited in O'Brein (2001), asserts that in order to make a rational career decision making, individuals have to know their own self and match that with the personality and skills needed in a certain profession. According to Parson, a person begins with exploring his or her interests, then recognises his or her values and abilities, and eventually determines a realistic career to pursue. In this study, 13 of the students expressed that they chose teaching as the nature of the profession matched their personal characteristics. There are no significant differences among the three groups of students regarding this response. The students relate their personal characteristics with the values and norms of teaching that are derived from experiences as school pupils in order to define the explicit nature of teaching activities and requirement as a teacher. A number of students expressed their beliefs about the compatibility between their attributes and characteristics of the teaching profession such as:

"I like teaching and sharing knowledge." [S9 TP¹⁰]

¹⁰ Student codes: S9 = student number 9 in the focus group conducted at TP school. S is for the

student's number and the other characters refer to the school's name, DPLI stands for group of career change students, and FY indicates group of first year students.

"I love children and look forward to working with them." [S5 SP]

S2 TP and S5 SP were undergoing their teaching practise at the time the interview was conducted. They exhibited a sense of pride and joy when sharing their interests during a group interview. In the same fashion, their junior in her first year also mentioned the personal quality that she has as a push factor to apply for the teacher education programme. With strong eye contact with the researcher she mentioned,

"When I enrolled in a pre-university programme, I was always engaged in community services particularly programmes that involved school pupils. I taught them accounting and mathematics. I was immensely satisfied when I was able to help out young children that have difficulties in these two subjects. I am able to motivate people to get the job done." [S1 FY2]

It is obvious that these students eagerly expressed that they approached teaching as the nature of the profession suit their interest (i.e. serve the young people). However, one of their colleagues in the DPLI programme mentioned that:

"I am eager to learn new things and do things differently. The teaching profession offers an opportunity for continuous learning." [S5 DPLI1]

This fellow student has an interest in teaching but she relates the interest with the intrinsic rewards offered by the profession which seems more individualistic in nature, compared to the other two students who related their interests with the group of people (i.e. young people) that they will serve throughout the profession. However, as argued in Subsection 2.4, intrinsic rewards are internally derived and can promote engagement with the teaching profession. Further expression from S5 DPLI may explain the level of her engagement towards teaching, "we learn and we teach what we have learnt".

It is apparent from the three quotes above that the students try to conceptualise their individual differences and their ability, and relate them with the perceived norms and values of the teaching profession, importance of knowledge, and function in the society.

The quotes also explain some of the traits that sociologists claim as essential for teaching to become a profession.

Their interest starts with identification of their strengths and attempts to relate with the profession that they are familiar with. Although Holland (1985) and Parson (1909) suggest that it is important to identify the people – profession suitability, only 13 students emphasized the reason to become a teacher with personal characteristics. Furthermore, the students' narratives about their attributes highlighted two significant factors that became a focus of interest. Firstly, the students mentioned about liking and working with children. Liking children has a significant relation with a student's decision to become a teacher both in the Western and Eastern countries (e.g. Brookhart and Freeman, 1992; Harun, 2006). The teaching profession provides plenty of space to interact with children, thus be able to understand them better. Secondly, the student's personality, interest and ability are associated with their personal needs of achievements. The students expression of their desire for continuous learning (S5 DPLI1) and sharing knowledge (S9 TP) is explained by Giddens (1991) as a choice of lifestyle that influences the students' decision on their professional choice. In this sense, the students saw the teaching profession as not benefitting only the students but also for their self-development. The student's ability to represent their personal attributes in conjunction with the profession also suggests that they showed their connectedness, sense of care and pride in their professional choice.

S9 TP, S5 SP, S1 FY2, and S5 DPLI1 expressed that they were interested in teaching and intend to apply for the teaching profession after graduating; however, one of their colleagues undergoing teaching practise stated that she was not sure of her future undertakings. This student, S10 AB intends to become a member of the teaching profession but does not fully adhere to what is imposed on teaching, which is inspired to teach. This situation is explained by Bruinsma and Jansen (2010) that people have different levels of engagement with the profession. S10 AB indicated that:

"Initially, I was not interested in teaching but the nature of profession suits my personality. I mean, teaching offers opportunities to meet other people and become the centre of attention. I love to socialise, work with other people and get people's attention." [S10 AB]

S10 AB has stereotyped the teaching profession as a profession that is social in nature and offers the opportunity to meet other people, and she tried to relate her personality with the 'requirement' of the job thereafter. It is proven that the school years have a strong influence on a student's experience. Through experiences as school children, a student learns the main activities in a teaching profession and begins to stereotype the profession as they experienced it before. This situation is also explained by researchers in sociology who suggested that people tend to make an occupational decision based on their knowledge about the job and to some extent their first job was the only one that they knew about (e.g. Bendix, Lipset and Malm, 1953, cited in Chinoy, 1955).

The relationship between personal attributes and assumed professional destination gives one important input for this study; the student's decision to approach the teaching profession is not always related to the identification with teaching if the level of engagement is considered. This subsection suggests that while the students expressed that they will approach teaching, the level of engagement with teaching varies.

Only 13 of the students expressed their reason to embark on a teacher education programme as related to their personal attribute that influences them to choose the teaching profession; however, 30 of their peer colleagues expressed the influence from significant others as having an enormous impact on how they view the teaching profession. The next subsection explores the role of significant others, or termed as a model in influencing the student's decision.

4.2.2 Model influences

Besides the compatibility between the students' characteristics and the nature of the teaching profession, the thematic analysis revealed that almost 38% of the students in this study pointed out that they favour the teaching profession because of influences from other people, such as their parents and former teachers. This process is explained by Ginzberg, Ginsburg, Axelrad and Herma (1951) through the theory of occupational choice; they posit that there are many phases involved in career decision making, which begins in childhood whereby children try out various occupations through role play or pretend play and the use of their imagination. Ericson's (1968) view cited in Smith and Hogg (2008) on the socialisation process posits that children start to build their own

image and continue to develop during adolescence and early adulthood. He also stressed that powerful and attractive role models provide the children with a prototypical of the profession. The explanation by Mead (1934) of the concept of "playing at" occupational role also supports the influence of role models into the development of an individual's attribute. This concept, according to Mead. is the idea where "the child plays at being a mother, at being a teacher, at being a policeman..." (p. 150). Thus, parental or family members' occupation as well as other occupations occur in a social setting such as teachers, police officer, nurse and doctor are often imagined when children are asked about their occupational aspirations (Lee, 2012; Slocum, 1959). One student implicitly mentioned about the significance of a model in their decision:

"I wanted to be a teacher since I was small. When playing with my friend, I chose to be a teacher. My neighbour called me *cikgu* (teacher) too! (laughing)" [S4 FY2]

In this study, 26 students from Group 1 (undergoing teaching practise), two students from Group 2 (career change), and two students from Group 3 (on their first year) consistently expressed that parental influence was the single most important influence for them when choosing this programme. Some students whom parents are or were teachers, are inspired by the recognition from the society towards teachers. They mentioned that the respect that the members of society gives to teachers are spectacular. For example, one student whose mother was a retired teacher stated that,

"... no matter where, people that recognised my mother will greet her with *Assalamualaikum* (peace upon you) *cikgu* (teacher) or good day *cikgu*. I have no offence when my mum asked me to become a teacher. I feel blessed!" [S5 BJ]

Furthermore, S5 BJ admired the teaching profession and perceived that the profession is favourable due to the positive scenario which she experienced with her parent. She observed from social learning how society appreciates teachers' work and that experience served as a motivation for her to choose teaching. Her colleague, S9 AB also expressed that:

".. teaching is a noble profession. My dad is always remembered by his former students."

Similarly, S2AB expressed that,

"My father and three of my sisters are teachers. They are very respected by the society."

These three quotes are among the students' expressions on how family members who are, or previously were teachers, influenced their decision to enrol in a teacher preparation programme. This situation is explained by Albert Bandura through the Social Learning Theory. Bandura (1969) posits that people naturally learn from imitation or observation of certain people whom they treat as a model. In this case, the students learn to discriminate the status of the teaching profession when they observed the recognition received by people they modelled. They applied for a teacher education programme to imitate the model (parents) that is respected by the society. However, while society recognition is a motivation for some students to embark on teaching, a small number of their colleagues tried to understand their model in a highly technical manner.

Some students, through observation and experiences expressed that they do not aspire to follow their parental footsteps in becoming a teacher. These students, S1 SP and S3 DPLI1 mentioned the following; "busy profession" and "always spend more time to teach the other person rather than their own child", as reasons to refuse teaching, hence they try to escape from becoming a member of the profession. Their refusal, however was overcome with strong influence and pressure from their parents which led to the willingness to choose a teacher education programme. In this case, parental influences became the reason for students to comply with the profession but was not able to persuade the students to voluntarily become a member of the teaching profession.

Furthermore, the parents that never actually taught or expressed a wish to become a teacher provided more pressure for students to choose teaching. Many students expressed that either their mother or father, or both, insisted that they apply for a teacher education programme. Two students stated,

"... discuss with family, my dad asked me to apply for the teacher education programme. Becoming a teacher is just to satisfy my family's aspiration." [S6 AB]

"I have loved architecture since I can remember. I love arts. My parents, however urged me to apply for teaching training." [S1 TP]

The students' narrative clearly distinguished their interest and motivation towards teaching before they embarked on this programme. It is apparent from the stories that the students who admired the recognition from the society showed more identification with teaching compared to the other students who had parental pressures. The latter indicated that they had no interest in teaching at the time they decided to apply for the teacher education programme. As a consequence, students such as S6 AB and S1 TP did not have any idea on what type of teacher they will be in the future and activities that they will encounter during their teacher education programme. Their motivation to become a teacher is lower compared to the students in the previous group who believed there was compatibility between their personal attributes and the nature of the teaching profession, and the students who believed that they were positively influenced by their parent who is or was a teacher. Ginzberg et al. (1951) posits that the process of professional choice often involves compromise as there are so many aspects to consider regarding the future profession. In this case, the students are seen as compromising their disinterest with the employment opportunity after graduating. In social identity terms, the students may pursue social mobility as they try to identify with another group of occupation that suits their motivation.

Further investigation and understanding of the situation of parental influence suggest that the encouragement stems from the parents' desire to maintain or move from the current social class to a higher social class. It is apparent from the data that the majority of students in this study are from the working class income. Again, the employment opportunities after graduation became a main contributor of parental influence. A story from S6 SP confirmed that the socioeconomic status influences the parental aspiration:

"I planned to pursue a degree in business management. My mum discouraged my choice because she said that there are no job guarantees, no future." [S6 SP]

Concomitantly, these findings suggest that students followed their parent's aspirations without themselves becoming an active student and expressed their values in their own decision making. The parents appeared as neither reinforcing nor expanding the

students' desire to teach, rather just making a decision on behalf of their teenage child. Except for the group of students who perceived teaching as a respectable profession, the other students who conformed to parental desire to become a teacher showed that parental aspirations became an external influence for the students' decision making, rather than embedded with the students' values in deciding their future undertakings.

The findings also suggest that work on structuralism dominated the reason for students to embark on a teacher education programme. Structuralism echoed that students do not make rational decisions towards their career choices. They were inclined to follow the tradition of the society or socioeconomic status to make a choice to embark on a teacher education programme. The concept of habitués also explains the role of parents in determining their teenage child's future undertakings. Young people in Malaysia, particularly females, still are dependent on their parents until they get married (Tan and Rey, 2005). Therefore, a secure attachment relationship between parents and a teenage child creates a habitués for most Malaysians. Nonetheless, Ginzberg et al. (1951) claimed that young people explicitly make rational decision making because when they agree with the structural influence such as from parents, they actually have compromised their interest, capacities, values, and employment opportunities towards other professions. These findings are contrary to the Western studies of students becoming active students in the learning process.

Besides parental influence, former teachers considerably affected the students' decision to become a teacher. The influence comes from both sides, whether they honour or dishonour their former teachers. As a sign of honour, the students indicated that they admired their former teachers and aspired to capture some of the qualities that they valued in the teachers who positively impacted their interest. The students mentioned teachers that had a profound and meaningful impact on their life. They also appreciate their experiences with 'good' and 'effective' teachers during their school years. As a result, the students feel that they would like to offer the same opportunities to the new generation that will have their education in schools. One of the students happily stated:

"My former teacher during secondary school just loved her job, and she truly cared. Seeing that was the biggest thing... I want to be a teacher, be like her (former teacher)." [S7 TP]

Although some students described the values of the referent group that made them choose a teacher education programme, a number of their colleagues expressed the contrast. Quite a number of students expressed their frustration of unfortunately becoming a victim of social inequality during their school years:

"She (the teacher) rarely paid attention to me because I was not excellent in mathematics. My friend with good mathematics was very lucky." [S5 DPLI2]

They felt upset when some of their former teachers neglected their right to learn. This phenomenon challenged the social identity brought by the teaching group. For decades the Malaysian education system applied an exam-oriented approach whereby students are assessed solely based on their academic achievements (Hwang and Embi, 2007). Thus, some of the teaching approach is advantageous to school pupils with high academic abilities. Bourdieu, cited in Herr and Anderson (2003) explained that this situation, termed as "symbolic violence", is one way to create social inequality through the teaching and learning process. Bourdieu uses this concept to explain the mechanism used by dominating a social class to impose an ideology, habit or lifestyle to another lower social group. As a result, the lower social group will have difficulty to be content with their lives. In an educational platform, pupils with a lower academic ability are struggling in their academic lives. The situation becomes worse when there are cases where teachers focus on high achievers and are less focused towards the low achievers. One of the students related her memory as:

"I wanted to be a teacher since I was in secondary school, after experiencing an unfavourable moment with my former teachers. Teachers refused to teach our class. Yes, we were not smart as our peers in the advanced class. But we want to learn as well." [S7 TP]

To add, most Malaysian primary and secondary classroom in typical¹¹ schools implement the streaming system. Streaming refers to separating the school pupils according to their academic abilities. Pupils attend classes only with their peers who have the same academic achievement as them ("Ministry of Education Malaysia", n.d.-c). Although this system has a positive affect because it allows teachers to implement

-

¹¹Typical schools refer to regular schools that offer comprehensive national curriculum.

the teaching and learning process within the pupils' ability in the classroom (Zevenbergen, Whitehurst and Zevenbergen, 2003), the effort to neglect pupils with lower achievement is unacceptable.

There are two factors that cause teachers to discriminate their pupils according to streamed classes. Firstly, it is a stressful task for the teachers to raise the school's academic achievement; hence the focus on the high achievers will help to ensure that schools have a good achievement. Secondly, the majority of pupils in the lower achievement classes are from low income families compared to their counterparts in the high achiever classes (Hyland, 2006). Researches consistently suggest that pupils from a lower socio-economic status are problematic pupils in school (Sirin, 2005). As a result, some teachers may be less interested to teach in these classes. Students in this study voiced their dissatisfaction with these discriminating teachers during their school years.

Furthermore, the students also dishonoured their former teachers who did not show professional characteristics as teachers. This student expressed anger on what she experienced during her school years:

"She (former teacher) was so emotional. She brings her personal problems to school. She is not professional!" [S3 AB]

The unfavourable moments with former teachers had motivated the students to choose teaching with the aspiration to ensure that their unpleasant experiences with former teachers are not repeated. In this case, the students willingly approached the teaching profession with the enthusiasm to fix the negative social identity caused by their former teachers.

To conclude, this subsection highlights the influence of models towards the students' decision to embark on the teacher education programme. As discussed in this subsection, parental influences have a significant effect on the students' professional choice particularly among the first group compared to the third group of students (students in their teaching practise and first year students). It is assumed that first year students have a lesser parental influence as a result of the new teacher placement policy imposed on them (refer to Subsection 1.9).

Implicitly, students that rely heavily on their structure pay less attention to their self advantage, and rely more on the structure. Furthermore, some of the students appreciate their former teachers and aspire to take in the positive qualities of the teachers who impacted their interest to become a teacher. Some other students voiced their depreciation towards their former teachers, and stated that their decision to embark on a teacher education programme was to show their former teachers that they could also be successful even though they were once low achiever students in school. The students with an unfavourable experience also vowed not to allow social inequalities to dominate the educational system. Thus, it can be summarised that with the model influences, the students try to achieve a positive social identity by explaining their intention to approach the profession, but in some cases, there is more room for improvement as the level of engagement is still low and the enthusiasm has to be improved particularly in relation to the acquisition of knowledge.

4.2.3 Former life experiences

The most influential factor for students to choose the teaching profession is related to their beliefs about the recognition of teaching, as opposed to other professions. As mentioned in Subsection 2.4, the Malaysian society perceives the teaching profession as a traditional and non-prestigious profession, yet the virtue of teaching cannot compete against other professions. The status of teaching is in harmony with most sociological perspectives on teaching as a profession; however, the virtue of teaching has turned teaching into having a positive social identity, even though it is only to the eyes of the local community. Society appreciates teachers because they believe teachers are a professional group that gives priority to the service rather than monetary incentives, and responsible in all actions and consequences made (Sufean Hussin, 2004). Quite a few students sustained these beliefs while acknowledging their reason to become a teacher. The students relate the virtue of the profession with their experiences in society. One career change student expressed the experiences with the youth in the society and believes that the teaching profession is able to help to develop positive values among young people:

"Starting in the 90s, the situation of young people was so alarming. Social problems are more glaring especially amongst the Malays. Drug abuse, sex before marriage¹², smoking and bullying. Being a teacher and having a connection with the young people at school is a way to further educate them with Muslims as well as Malaysian values." [S4 DPLI1]

S4 DPLI1 believes that basic education starts at home and parents are responsible for their children's well-being. However, she conveyed the message that teachers are important people who are responsible to take over the parents' responsibility while the children are at school. She believes that the legitimate and expert power that a teacher holds is able to help the school pupils development. In the teaching profession, teachers as social agents have legitimate power inherent in the role of teachers. Additionally, teachers should have expert power because school pupils view teachers as someone who is knowledgeable and can help them in their learning process. Cheng (1994) posits that the power that teachers have may be used as tools to influence school pupils' values and behaviour. The students' commitment to develop school pupils' values is consistent with Malaysia's Rukun Negara (Pillars of Country). Rukun Negara provides guidelines for Malaysians especially in creating unity among the different ethnics in Malaysia. The students mentioned schools as a social platform to develop school pupils' values consistent with the Rukun Negara.

Besides developing values, most of the students believe that knowledge acquisition is the main agenda in an educational process. One student with working experience shared the consequence of not having adequate knowledge in this new era. The student [S5 DPLI1] worked in the financial sector before deciding to apply for the teacher education programme. She expressed her uncomfortable feelings regarding her previous job. She mentioned that there was discrimination whereby Malay customers were not given appropriate advice compared to Chinese¹³ customers in financial services. The

¹²Islam prohibits Muslims from having sex before marriage. It is stated in the Al-Quran (sacred book for the Muslims) '. .. and do not approach unlawful sexual intercourse. Indeed, it is ever immorality and is evil as a way'. [17, Verse 32]

¹³ Chinese is second largest ethnic group in Malaysia. The present multi-ethnic climate of Malaysia is marked by the prominence of the Chinese in business and trading. The Chinese owns most of the private financial services institutions.

lack of information by Malay customers affected their chances to fully utilise the financial institution's facilities:

"I want to be a teacher so that I can strengthen the young people's knowledge, especially the Malays so that they will be competent in business and trading like the Chinese." [S5 DPLI1]

A colleague of S5 DPLI1, of Indian ethnicity seconded that there exists an inequality in the educational process. She expressed:

"I am involved in an Indian non-government organisation (NGO), providing help and motivation for unfortunate Indian youth. It is very sad that some of them are left behind. Being a teacher gives me more chances to contribute to the development of young people especially the Indians." [S6 DPLI2]

Besides experiences within the society, a number of students indicated that their experiences in teaching had given them some intangible and positive feelings which were very motivational. This is inconsistent with their colleagues who shared their experiences in helping the society; these students conveyed that the feeling they got when working with school pupils in the classroom was frustrating. For example, S4 DPLI2 sadly stated:

"My experiences as a substitute teacher at a secondary school were particularly frustrating. Many young people are not interested in learning, not interested in their own education. Some struggle to succeed."

The students saw themselves as potential educators who could make a difference in the society, particularly with young people. Regardless of the students' experiences, they were among many other students in this study that experienced the social reproduction and believed the educational process influenced the construction of social inequality in a society. From the narrative, it is apparent that the students assumed that young people's knowledge is in the form of cultural capital as stated by Bourdieu. The students strongly believe that the teaching profession offers an opportunity for them to contribute to the process of reproducing the cultural capital through the educational process, with teachers as the main contributor. The students have intentions to minimise

the social inequality that has been transferred from one generation to another. The inequality, according to the students, provides more harm than good. The students' intention also suggests that they are inclined to help the society through their profession. Harun (2006) also suggests that an individual with an inclination in the teaching profession has the quality of altruistic motive, that is seeing the profession as socially worthwhile and beneficial.

Besides the experiences with young people and teaching familiarities, their experiences as students also influenced the students' reason to embark on a teacher education programme. As presented in Subsection 4.1.2, the experiences of having inequality in education enhanced the enthusiasm to become a teacher. To add, student S1 FY2 emotionally narrated:

"Teachers must be fair to their pupils regardless of their academic achievement. During my secondary school years, pupils with satisfactory academic performance will get more attention from teachers as their academic excellence will increase the school reputation¹⁴. What about the unfortunate pupils that were academically left behind? The teachers should focus or give more attention to this group of pupils."

S1 FY2 is not alone. A few students mentioned that they were students who struggled in school. Their unpleasant experiences with former teachers increased their desire to teach. As mentioned above, an unfavourable experience with former teachers served as a motivational factor for the students to choose teaching. It may be concluded that the students tried to pursue a social creativity strategy in order to enhance their group self-esteem by defining more positive characteristics of teaching.

This subsection describes how the students' experiences during their school years influenced their attitude perception about teaching. Most experiences became the

accessible to all. The new assessment system, namely the School Based System (PBS) has been implemented by phases in both primary and secondary schools throughout Malaysia.

¹⁴ The Malaysian Education system is dominated by an exam-oriented system whereby the assessment system is more towards the *assessment of learning* which gives an advantage to high academic performers, instead of *assessment for learning* (Ministry of Education, n.d.-b). However, there are several changes to improve the assessment system and make education more

motivation for the students to favour teaching, and showed a more positive emotional reaction towards approaching the profession, thus maintaining the salience of social identity.

The next subsection discusses the most popular reason for the students to embark on a teacher education programme. The vocational reasons comprise the students' passion to teach, content driven and the extrinsic reward offered by the profession.

4.2.4 Vocational reasons

Analysis on the students' stories in all three groups consistently suggest that the desire to become a teacher began in childhood and was described by students as a 'passion':

"My passion to become a teacher since my childhood." [S2 AB, S3 BK]

"I have wanted to be a teacher since as far back as I can remember."

[S2 SK]

The passion to teach was also felt by their juniors in the first year and career change students. One of the career change students mentioned that she applied for the teacher education programme since she completed her secondary school. Since the admission to the teacher education programme in Malaysia is highly in demand, and admission to Malaysia's public higher education institutions are based on a meritocracy¹⁵ system, she was never successful with her application due to non-promising results during secondary school:

"I was interested in teaching since primary school, but my (secondary leaving exam) result did not qualify me to apply for a teacher education programme. I enrolled in a business management programme and graduated. After graduating, I always found an

¹⁵ Meritocracy is a selection system to the public higher education institutions based on the academic performance during the leaving exam in secondary school.

opportunity to become a substitute¹⁶ teacher at school. Every year I applied for the graduate teaching certificate programme. My last employment before I enrolled in this programme was a preschool key worker. My passion is in teaching." [S1 DPLI 2]

Further investigation of the students' passion to teach revealed that activities in the teaching profession had attracted the students towards the teacher education programme with the intention to apply for teaching thereafter. The students through their stories expressed different situations on how they appreciate the nature of the teaching profession and decide to become a member in this group. There is evidence that the students related the motives with religious belief. One of the students added:

"In Islam, man is created to be a caliph, the leader.... Teacher is one of the leaders in a society." [S4 DPLI2]

Islam sees religion as an encompassing way of life (Hassan Farooqi, 2006; Rice, 1999). This suggests that a Muslim's everyday life includes the social, economic, political aspect and human relations which must be based on the teaching of Islam. One of the Islamic teachings related to the social aspect is performing a task or job that will benefit themselves and others. This behaviour is treated as *ibadah* (performing a religious duty, a good deed). In this case, the students related their reason to become a teacher as a way to fulfil their obligation as a Muslim. The practice is different with Western cultures which tend to regard religion as a private matter (Rice, 1999). Consistent with the image of a teacher as a role model for the young generation's development, society has not given much choice on who should become ¹⁷ a teacher. Teachers, according to the Malaysian society should be knowledgeable and obedient to religion in order to become a role model for students. It is consistent with the first pillar of the Rukun Negara,

1

¹⁶ Substitute teacher is a person who teaches a school class when the regular teacher is unavailable for a short period of time because of maternity leave, illness and other reasons. Paid maternity leave in Malaysia is three months.

¹⁷The basic beliefs for a Muslim falls into six main pillars of faith, the first of which is called the article of faith, namely belief in Allah, His Angels, His Books, His Prophets, and the Day of Judgment or the Day of Resurrection (life after death), and the belief in Destiny or Fate (Qadar) (Kayed, 2007). The main duty as a caliph, among others stated in the Al Quran [2, Verse 30]: "And [mention, o Muhammad], when your Lord said to the angels, "Indeed, I will make upon the earth a successive authority." They said, "Will you place upon it one that causes corruption therein and sheds blood, while we declared Your praise and sanctify You." Allah said, "Indeed, I know that which you do not know.""

which is a belief in God. The religious obedient means that the teacher should have faith in their life.

Besides religious beliefs, the students indicated that they personally valued education and positioned themselves as someone who has the ability to facilitate the learning process among the younger generation. Representative statements included "opportunity to educate", "take care of the younger generation", and "opportunity to share knowledge". Many also indicate that teaching is a noble profession. Research on motives to become a teacher particularly in Islamic countries has consistently found that the teaching status dominated the factors that influence the prospective teachers to choose teaching (e.g. Mogra, 2010).

Besides their peer students who have a passion in teaching, a small number of students expressed that they are attracted to the teacher education programme because of the extrinsic rewards offered by the profession. This includes guaranteed employment opportunity, long holiday, short working hours, and job security. As suggested in Subsection 2.4, research in higher education consistently found that extrinsic rewards are a popular reason for students to choose teaching as a profession. Although only a small number of students are inclined to relate their reason to embark on a teacher education programme with rewards, this situation was very worrying because the students were perceived to become a member of a group with the 'wrong' motives (Sinclair, Dowson, and McInerney, 2006). Sinclair and colleagues posit that material reasons for teaching will negatively affect the students' desire to teach and remain in this profession. This is consistent with other literature that relates a negative relationship between rewards and engagement in teaching. A few students perceived the teaching profession as vocational in nature, without any emotional attachment. One student undergoing her teaching practise stated that:

"Schools are everywhere...... so I will still have a job." [S2 HD]

Similarly, her colleagues mentioned about one specific benefit of being a teacher:

"Long holiday." [S3 SP]

Others mentioned about the shortest working hours compared to other professions and the job guaranteed after graduation as pull factors of choosing teaching as a profession. Influence from external factors towards the professional decision making can be related with the individual requirement to meet their needs. The Maslow Needs Theory for example, posits that an individual will meet the requirement of a lower level of their need, which is physiology before moving to other higher hierarchical needs (Robbins, Judge and Vohra, 2013). The students' decision to become a teacher because of the rewards seem to fulfil their lower level needs before they go to a higher level that offers more opportunities for them to be emotionally attached to the teaching profession. They also seem to possess more individual salience rather than social identity salience.

4.2.5 Other reasons to embark on a teacher education programme

Miller and Form, in Slocum (1959) claim that a majority of people do not make a rational decision on their professional choice. They argued that people accidently make their decision because they compromise their interest, capacity, values and opportunities offered by a specific profession, but are not motivated to become part of the profession. The students may have a genuine interest in another field and employment opportunities, however the guaranteed place offered for a graduate from the teacher education programme attracts them to apply and embark on this programme. The student seems to not compromise as they have excellent secondary academic qualification:

"I have no idea. I just follow my friend. We filled the online application form (for university admission) together." [S6 TP]

Some students however, do not see the teaching profession intrinsically. They only want a job offer, and by enrolling in a teacher education programme, the students have made themselves as an individual "who is soon to get a job". A student in her first year expressed:

"My mission is to get a place in the undergraduate programme and graduate, and enjoy the privilege of a graduate job." [S2 FY2]

Furthermore, there were a small number of students who embarked on the teacher education programme due to a technical mistake. As the admission in higher education

is coordinated by the Ministry of Education, the chances for students to change to other academic majors or higher education institutions is a hassle. Quite a lot of students accepted the offer as it is a chance for them to get a place at a higher education institution. There are two students in this study who expressed that they applied for the teacher preparation programme due to a technical mistake¹⁸, and they chose to accept the offer and have been part of the teacher education programme community. The students in this group had made a rational decision by accessing their own abilities and interests, evaluate the range of opportunities, and then made a choice which matched ability to opportunity. One of students in the teaching practise expressed:

"I am aiming for a legal qualification. During my school years, I was involved in debates. I applied for a law degree and courses...." [S1 SK]

However, this student by mistake chose the teacher education programme as her first and second choice of study programme¹⁹:

".... it was a misunderstanding, I put teacher education programme as the first and second choice of concentration area. That is why I am here."

The student S1 SK embarked on teacher education by chance. However, her initial decision to enrol and get a law degree is influenced more on her belief of the ability to pursue law related courses. This student has the tendency to pursue individual mobility as they perceive the outgroup as having a more positive social identity. However, she has to sacrifice her interest towards law at the moment because the process of changing higher institutions is a hassle in Malaysia. The current institution does not offer law related courses.

-

¹⁸ Misunderstanding while filling in the application form.

¹⁹ Application for admission to undergraduate programmes in Malaysia's public higher education institution is coordinated by the Ministry of Education. To apply, candidates must fill in the online application opened between January and March every year. There are eight choices of degree courses for the candidates to choose. Candidates must choose the courses based on priority, and those who are interested to apply for the teacher education programme must choose the degree courses in education in their first and second place.

4.2.6 Subsection summary

Results from the thematic analysis suggest that the teacher education students' reasons to embark on the teacher education programme can justify their initial teaching identity. Not just the thematic analysis, at certain circumstances the student's body language give meaningful inputs regarding the identity that they possess. In general, the majority of students expressed that they have a positive feeling towards the teaching profession at the time they decided to apply for admission to the teacher education programme. The thematic analysis helps to classify students that have an intention to approach teaching profession after graduating by "Yes, I want to be in a teacher's group". Video recording also captured that the students in this category are happier to express their positive views on teaching. The body language presented by the students are added to the quotes by expressions such as 'eagerly', 'happily' and 'emotionally'.

The students in this group believe that their enrolment in the teacher education programme is influenced by many factors such as their personal attribute, influences from the social environment and the nature of the profession itself. It is apparent from the data that the students create their own meaning of a teaching job by presenting the stereotype of the teaching profession. However, the students' positive social identity is influenced differently by those factors, and some of the factors may become a challenge to maintain a positive social identity. From the quotes presented throughout this subsection, there are students who are attracted to the employment opportunity offered by this profession, hence this becomes a major source of motivation to enrol in this programme; thus, their intended behaviour cannot be explained using the Social Identity Approach.

Additionally, there are quite a lot of students who were influenced by significant others, particularly parents, but did not have any interest to become a teacher. This group of students indicated that they approached teaching because of pressure from others, particularly because of the employment opportunities contained therein. This situation is consistent with the concern that society imposed towards a new generation of prospective teachers, where the genuine reason to embark on a teacher education programme has been overshadowed by the employment opportunities after graduation,

not because of the greatest inspiration to become a group member of the teaching profession.

The findings of this study also revealed that there are a small number of students who are not sure whether they will apply for the teaching profession after graduation. This second category of students, who implicitly expressed, "I do not know", consists of students who embarked on this programme mostly by chance. This includes students that have no clear reason of why they embarked on the teacher education programme. Students with peer influence are most likely to categorise themselves in this group. Finally, the third category, may indicate that the students possess an unfavourable attitude towards teaching as they may not continue with teaching after graduating. This group consists of students who are in the teacher education programme due to a technical mistake. These students aim for the academic credential offered, which is a Bachelor of Education, and may decide on other employment opportunities after graduating.

Furthermore, the findings on the students' reason to enter a teacher education programme suggest two major issues. The first important issue revealed that the students' decision to embark on a teacher education programme was influenced by different factors. There are personal factors, structural factors as well as external influences. These influences accumulate and develop the students' identity towards the teaching profession. As discussed in Subsection 2.4, the values and norms of teaching are hard to identify as the profession itself is not regarded as a profession. Apparently, the student's evaluation towards the teaching profession as presented in this subsection is linked with the assumption that the stereotype of teaching will be used as the norms and values of teaching. They aligned themselves with many admirable traits that they believed teachers possess according to their beliefs of teachers; ambition, optimism, ability to inspire, to lead, and to change lives. Nonetheless, their counterparts who embarked on the teacher education programme with parental influences had not viewed themselves as a teacher when accepted into this programme, thus bringing an unfavourable acceptance of teaching as a profession. Furthermore, the students with experiences with young people, whether within teaching activities or not, have a more favourable perception towards teaching.

Moreover, as mentioned in Subsection 1.9, the students have undergone a variety of assessments and selection criteria to enter the teacher education programme; however, their effort to achieve a positive social identity in teaching as presented in this subsection is not shown as always promising. It is apparent from the findings as discussed in this subsection that the students can be classified into three different categories. Among these three categories, the students included in the second group seemed to be passive entrants, and there is a possibility that they become a virtual tabula rasa or blank slates, which means prepared to absorb whatever is given to them during the professional socialisation process (Artkinson, 1985).

Apparently, the differences in the reason to embark on a teacher education programme among the three groups in this study are related to their experiences and interaction with the environment, not on their understanding of the national policy imposed on teachers, or any rules and regulations. The reason for students in Group 1 (undergoing teaching practise) and Group 3 (first year students) to embark on a teacher education programme is less realistic compared to the reason stated by the majority of the career change students. Career change students viewed the teaching profession more in terms of what the profession offered as opposed to other professions. Their working experience in other industries before deciding to apply for a teacher education programme is influenced by the positive reinforcement on what teaching offered as opposed to "my previous work or job" (i.e. S5 DPLI1). In contrast, their counterparts in Group 1 and 3 perceived that teaching is a platform for them to get a job.

The next subsection explores students' self-definition on teaching at the time this study was conducted.

4.3 Student's identity during commencement in teacher education programme

This subsection continues to search for answers related to student's identity towards teaching; focusing on the emotional attachment and behaviour students possess after undergoing the socialisation process at the teacher preparation programme. It is important to note that the students from different groups have undergone different types of socialisation process depending on the semester they are in and the route to the

teacher education programme. As mentioned in the introduction of this chapter, results from the thematic analysis revealed two major themes of how the students viewed teaching; favourable (positive social identity) and unfavourable.

Besides the reasons to embark on the teacher education programme, the students' narrative discovered that their commencement in the teacher preparation programme and holding the status as a 'prospective teacher' have a significant effect on their perception towards the teaching profession and their subsequent attachment and intended behaviour as a member of the profession. A majority of the students communicated the persuasive messages that induced the categorisation process as members of the teaching profession which will be detailed in Chapter Five, as this chapter will focus on the identity students hold and not the process of identity acquisition. The discussions of these factors, where appropriate, are presented according to the students' initial motive to enrol in the teacher education programme. A possible explanation for this arrangement is to identify the students' effort to achieve a positive social identity or to decline the group membership in the midst of the teacher education programme as compared to their initial motive at the entrance of the programme. The students' narrative on "who am I as a (future) teacher" gives the richest data on their identity and level of attachment to the teaching profession. The level of attachment is added to determine the identity as when the student progresses in their study, they are not just expected to possess a positive social identity by their decision to approach teaching, but must be able to determine the intended behaviour as a consequence of being in the teaching profession. The thematic analysis suggests three different categories to describe students' attitude; general evaluation of teaching profession, professional relationship and understanding of learning.

4.3.1 Evaluation of teaching profession.

There is consistency in the students' narratives from across the groups about the status of the teaching profession. It is generally accepted by the students that teaching is a noble profession. Noble profession, according to the students, refers to a profession that bring an enormous impact to the younger generation's future and well-being. Representative statements include "transform the knowledge", "work hard for the sake of students", and "treat students as we treat our own children". The findings from

Muslim students showed their beliefs of an obligation to spread the dakwah (calling and making an invitation to Islamic faith) to Muslim students. They invoked the hadith²⁰ statement "Convey from me, even one verse" (Bukhari). Muslims believe that conveying a dakwah is part of their obligation as a Muslim. The current situation in Malaysia shows that there are many social problems particularly among Muslim teenagers. Muslim students in this study believe that they can play a role in educating the young people, regardless the religion and ethnic, towards a better generation. Their non-Muslim counterparts also mentioned that "it is important to develop the young generation spiritually" [S6 DPLI2], which refers to educating the younger generation not only in academics but also in the way of life.

The students' perception is influenced by their experiences as students where they were involved with teachers. The students with teaching experience, both career change and students undergoing their teaching training, have in depth involvements in exercising the role of teachers in a real setting. One of the students, S3 AB who is undergoing teaching practise mentioned eagerly about her valuable experiences and how proud she was when attending her first class:

"I cannot tell you how satisfied I am when I stand up in front of the class, looking in their (students) eyes. You cannot have that feeling unless you experience it yourself!"

Additionally, the perception is also derived from interaction with the society. There is evidence that the Malaysian society perceives that all teacher education students soon will become teachers. The perception comes from the previous practise of teachers' recruitment in Malaysia where almost all teacher education students were offered a position as a teacher after graduation. One of the students, S6 AB in her teaching practise shared the moment that her neighbours called her 'cikgu' (teacher) since they know that she is in the teacher education programme. S6 AB expressed that she felt overwhelmed with the appreciation from her neighbours even though she is not yet a

-

²⁰Hadith are collective sayings of Prophet Muhammad (peace be upon him), a messenger of Allah. Islam does not separate public and private life, and the Islamic values are based on the Our'an and Hadith.

teacher. This appreciation, according to S6 AB becomes a source of development of favourable attitude towards the teaching profession.

Besides the status of the teaching profession, the students also perceived that teaching is an immensely challenging profession. The students from Group 1 and 3 could foresee that challenges will come from the students and expectations from educational stakeholders such as parents and society. In contrast, a career change student tries to justify the challenge in terms of the characteristics of teaching. Among all, they anticipated that teaching no longer offers short working hours. Data from this study suggests that the students perceived these challenges as part and parcel of the profession, and interestingly the challenges have increased their engagement with teaching. The responses from the students implicitly indicated that they tried to understand the reality of the teaching profession. As described in the preceding paragraphs, career change students with teaching experience shared their understanding of teachers' work based on their previous experience. Likewise, the students in teaching practise narrated their experience during the teaching training. Other students who did not have any teaching experience imagined teaching as they had experienced during their school years. One significant challenge highlighted by the students regardless of the group, is to teach students who attend below average classrooms, as described in Subsection 4.1.2. One student in her first year expressed that:

".. with the low achievers, we have to be patient, need to know how to handle (the students). Be sincere to teach, do not underestimate their (students) ability, keep it up, do not be too depressed." [S2 FY1]

Challenges are also associated with the prospective teachers' self. One of the students in her teaching practise stated that future teachers must accept that some of the students nowadays are very technological literate. Thus, future teachers must equip themselves with knowledge beyond their content knowledge that is taught in school:

"Students nowadays are exceptionally advanced. They are information seekers. Teachers these days do not have a lot of information compared to their students. Sometimes teachers have to admit their weakness." [S1 SK]

This statement challenges the traditional perception that the teacher is a know-how person and always recognised as a role model of students. However, findings in this study revealed that the majority of students conjointly perceived themselves as an example of a successful adult for their students. The majority of the students expressed their inspiration to become a role model for the students. They believed that a teacher's manners is a critical factor in school life and serves as implicit values education for students. The majority of the students in this study expressed this aspiration in their narrative. For example, S1 FY2 in her first year stated that:

"Teachers are an important icon in the society. They are always observed by the society, the way they talk, dress, their manners. Teachers are important to develop the society."

Likewise, her colleague in teaching training shared the following experience:

"Students show no respect if you are not good. They will not inspire you too." [S9 TP]

Students' perception on teachers' role as explained by Giddens (1991), posits that the modernity of the society depends on the agent who can lead to such changes. The students believe that they can become an agent through their important role in the development of the young generation. One of the important roles in being an agent is being a person whose behaviour, example or success can be emulated by others, particularly their students. Indirectly, the students believe that they have to develop themselves continuously to be a better person. Some students perceive that the teaching profession offers opportunities to develop potential and life essential skills for the prospective teachers. One student undergoing her teaching practice stated:

"I am happy and proud when people call me 'cikgu'. I see myself as a leader, as a mentor. I have to be a better person from day to day, because before I change other people (student), I have to start with myself." [P5 TP]

What are the perceptions of students with the intention to teach say about the teaching profession

The students who believe that they have interest in teaching at the time they enrolled in the teacher education programme expressed that they see themselves as 'progressing well' in the teacher preparation programme. Progressing well refers to their ability in categorising themselves as future teachers by imagining and demonstrating the image of the teacher that they will be, and develop a positive attitude towards teaching. However, there is one student in the teaching practise who feels that she will not apply for a teaching position after graduating. She feels that teaching is not challenging enough, hence it does not suit her personality. She expressed that:

"I was working part time before I enrolled in this programme. I feel that teaching is less challenging because it involves a routine job." [S7 SK].

S7 SK decided to embark on the teacher education programme driven by her interest in teaching. However, her experience in teaching practise reduced her motivation to teach. Further investigation on her stories revealed that her decision is related to the frustration of teaching. She mentioned that she loves to teach, but feels frustrated as the students rely more on the teaching of tuition²¹ teachers rather than their teachers at school. She mentioned in her story that: "I feel sad. Some of them (students) have no initiative to learn at school. They said that they will learn during their tuition classes."

What are the perceptions of students who do not intend to be teachers say about the teaching profession

The students who do not intend to become a teacher have similar beliefs with their colleagues regarding the status of the teaching profession, the challenges of the profession and being a role model for students. However, they refuse to teach as it is not the choice since they enrolled in the teacher education programme. This group consists

_

²¹ It is common in Malaysia for school students to attend tuition classes on top of their ordinary classes in school. It is assumed that the pressure to get good grades in the examination influences the students to attend tuition classes provided by private individuals or organisation. Tuition classes are charged.

of only a small number of students and all of them were undergoing teaching practise at the time the interview was conducted. These students made extraordinary efforts in teaching even though they are not preparing themselves to be a member of the teaching profession. The details of the efforts are described in Subsection 5.2.3.

What are the perceptions of students who are unsure to be teachers say about the teaching profession

Some of the students who initially had no interest in teaching have strived to understand teaching and the teaching components during the teacher preparation programme. For students in teaching practise, the experiences at school have a significant effect on their attachment to teaching. They are aware about the virtue of the profession and their role to meet the challenges in teaching for the sake of the students. Surprisingly, they perceive the challenges as something that increase their motivation to become a teacher. For example, one student, S2 TP who initially embarked on the teacher education programme because of parental influence, had difficulties when she went for teaching practise in a school. She teaches fine arts and found that the most challenging task in teaching is to control students in the classroom. However, she found that the challenge made her more creative, in terms of finding a way to control her students in the classroom.

Her colleague, S2 HD who had no interest in teaching when she started the teacher education programme expressed that the biggest challenge in teaching is how to influence the students to learn. She underwent teaching practise in a small town, with the majority of students living in the village, where their fathers work as either farmers or lorry and truck drivers. The area is surrounded by paddy fields, a flooded parcel of arable land used to grow semi aquatic rice. S2 HD expressed the challenge of being a teacher trainee at this school:

"Quite a few students come from a poor family; farmers and truck drivers. Truck drivers are seldom at home. Who wants to help with their children's learning?"

These two stories are among the cases of how students perceived teaching as a noble profession, but immensely challenging. Their motivation to teach, however was

amplified by the concern for students who are the product of social inequality at school. It is apparent from the findings that the students who had no interest in teaching tried to accept the challenges positively and change their initial attitude towards teaching.

As a conclusion, the students with an interest in teaching have developed their image as a teacher starting from the initial programme of teacher education until teaching practise. However, their colleagues with the initial intention not to teach, have learned from their interaction during teaching practise, and decided to pursue a teaching career as they can cope with the challenges during their teaching practise.

4.3.2 Professional relationship

The second variable which emerged from the thematic analysis refers to the students' beliefs on the professional relationship between teachers and educational stakeholders, particularly the pupils, colleagues and parents. The narrative of student-teacher relationship dominated the students' stories about their experiences and inspirations as prospective teachers. It is apparent from the data that the majority of the students believe that pupils come to school with different characters, inspirations and limitations. With this uniqueness, the students are aware that they must be sensitive to the individual pupils' needs. They are most aware with below average pupils in school. One of the students emotionally expressed;

"... try to understand their (student) emotion in the classroom. What creates a barrier for them to learn?" [S1 BK]

S1 BK is among the students that aspire to eliminate the social reproduction that is extensively developed in a school context. The students are aware that schools are working more towards academic excellence, however they feel that it is unfair to neglect the below average pupils. The students who initially have little interest in teaching assumed that this situation boosts their motivation to teach:

"... I feel attached to this (teacher) profession while I started my teaching practise. The pupils out there need a teacher not just to teach, but motivate them to learn." [S9 SK]

S9 SK imagines that the relationship between teacher and student has to be beyond the formal relationship at school. Although there is commonly a barrier between teachers and students because of the higher power distance as mentioned in Subsection 2.5, the students believe that they should lower the distance in order to maintain close relations with the pupils. The majority of students across the groups shared S9 SK's aspiration as a teacher. Surprisingly, many of them do not intend to focus on students' academic development, rather they insist on developing the young generations in terms of emotional and spiritual aspects.

"For me, if the students do not have the capability in academic achievement, or are experiencing learning difficulties, that is not the end of the world. There are a lot of opportunities for this group of students. I like the proverb – do not give a fish to our student. Teach them how to fish and they can use the skill forever. Motivate them, give them the appropriate skill." (S4 SK)

The students mentioned about the development of values among students twice then only they brought up academic issues. It can be reflected that most Muslim students relate the teaching profession to the role of a caliph as discussed in Subsection 4.1.4.

In addition to teacher-student relationship, the students in teaching practise and career change students also relate the role of current in-service teachers in the development of their beliefs towards teaching. The students that were undergoing their teaching practise had the opportunity to work with in-service teachers during the teaching practise. However, they perceived that the relationship 'is not going smoothly'. Findings from this study found that the majority of the students have no consensus with their inservice colleagues in terms of the way teaching takes place in school. They are consciously aware that there are a lot of cases whereby the in-service teachers promoted social inequality in school. For instance, one student in her teaching practise stated:

"I asked senior teachers what should I do to the students that are left behind academically. They said – use your creativity, or just teach them as you teach other students. Oh! I should teach them in a way that they will understand." [S1 TIWA]

One of the students who worked as a substitute teacher before embarking in the teacher education programme also expressed her frustration at the absence of support from her former in-service teacher colleagues. For these students, she experienced the gap between the novice teacher and the senior teachers. However, from the students' narrative, the absence of support did not demotivate them to search for the meaning of being a teacher.

S9 TP who evaluated her attribute as consistent with the needs of the teaching profession embarked on a Guidance and Counselling²² programme. Although she feels frustrated because a counsellor will not teach at school as she aspired to, she perceived that she can do the job better by sharing the knowledge with students during a counselling session. She mentioned her experience with young children in teaching training:

".. narrowing or closing the gap between teacher and students."

She expressed concern about the students' future:

".. this school is situated in a rural area. Majority of the parents work as security guards, operational line in a factory, and farmers. Most of them live below the poverty line. They do not see themselves beyond what their parents have."

She mentioned the following:

"I am willing to help them, but I only meet them at school."

Unlike the findings from the previous subsection which suggests that students from different groups have different perceptions about teaching, the findings on professional relationship found no distinct differences on the perception of the professional relationship among groups. All students perceived that there should be a close relationship between students and teachers, at and beyond a school context. One career change student mentioned that teachers carry a job title as a teacher for 24 hours a day,

²² Students with a Guidance and Counselling major are trained to become a counsellor in school. It is compulsory for Malaysian schools to have at least one trained counsellor. The counselling services in Malaysian schools gained momentum with the setting up of the

which showed her enormous commitment towards teaching. These findings suggest that there are positive perceptions on the nature of teaching from the students, particularly those who initially expressed no interest in teaching.

4.3.3 Understanding of learning

Understanding the learning refers to the epistemological and ontological beliefs about learning. The students' view on what constitutes knowledge and how to acquire knowledge affect their aspiration and belief on their teaching style. The literatures reviewed in this study suggest that Malaysian higher education students tend to memorise facts and their learning depends on the teachers. The findings from this study support the reviewed literature, where students suggest that the teacher education programme must provide comprehensive knowledge and skills for their trainees. One student mentioned that she lacks socialisation skills and has a problem to mingle during their teaching practise. She expressed:

"It should be better if there are courses on how to socialise with a colleague of different ages at school. It will help the trainee to communicate and socialise well during their teacher training." [S3 BK]

Similarly, her colleague put forward that other courses such as managing emotion is important for prospective teachers. This finding suggests that some students assume that the image of the teacher should be prescribed, rather than acquired through interaction with the environment. However, there is no evidence that students imitate their teacher educator as their role model, thus rejecting the assumption that teacher education students' professional socialisation process is determined by the functionalism approach.

However, the findings from this study found that the students' aspirations towards learning contrast from what has been reviewed in the literature. The students believe that there are multiple social realities, and knowledge is not static. These beliefs influence their teaching inspiration. One of the students shared her experience during teaching practise:

"Even though I am in my teaching training, I rarely use text or reference books in my class. I do not want my students to rely only on books that were written two or three years back. I am using current newspaper cuttings. Let's say, in a syllabus, students should know about landslides. We have a landslide phenomena over here (Malaysia). Give students the relevant paper cutting, ask them to try to understand what was written by the reporter on that phenomena, think and tell their friends what they think caused the landslide. Discuss! I believe students will remember what they are discussing rather than reading from a book, and trying to understand." [S1 TIWA]

As S1 TIWA shared her teaching experience, the students in their first year and some career change students who do not have teaching experience expressed their inspiration towards students' learning based on their experience as students:

"Say never to text books." [S5 FY1]

"I think teaching aids will help the learning process." [S6 DPLI 2]

These findings suggest that the students refuse to use textbooks as the only medium for teaching and learning process. The excessive use of textbooks in schools and Malaysian higher education contribute to the functionalism socialisation process where students are more dependent on textbooks and minimise the effort in discovering their own knowledge.

The students, regardless of their initial motive to embark on a teacher education programme, believe that students nowadays should be encouraged towards self-learning where they have to search for their own meaning. The concepts that the students highlight refers to the constructivist approach of learning. The students believe that the constructivist approach gives equal chance for students to learn. They believe that some students have the capability to remember the facts while their counterparts preferred the 'learning by doing' approach.

4.3.4 Subsection summary

This subsection presents the students' identity after they embark on the teacher education programme. In accordance with the normal practices in Malaysia whereby teacher education students will be offered a post as a teacher after graduation, these groups of students already perceived themselves as a member of the teaching profession. The first year students who do not have the privilege of a guaranteed job also reported as considering themselves as future teachers. As prospective teachers, they voiced their beliefs of their future profession and related matters. There are three factors identified from the thematic analysis associated with the student's identity that might guide their behaviour; perception about the teaching profession, professional relationship and understanding of learning. These views help them to evaluate the teaching profession accordingly. Most of the students are seen as trying to build a genuine interest towards teaching. However, there are quite a number of students still finding a way to build genuine interest, thus leaving them with the perception that teaching is just an occupation, without feeling attached to the profession, and behaving according to the teaching values and norms.

It is also apparent from the findings that the young generation, represented by 'school pupil' in this subsection, becomes an important agent for the students' identity development. The students' aim for social mobility among the young generation, and that becomes their main motivation that translates from their beliefs of teaching. It is apparent from the data that the students expressed the socialisation agents that help them to develop their attitude. The socialisation agents, however are not treated as significant others that pressure the students to conform to the profession, rather they exist as an agent of change.

4.4 Socialising agents who influence students beliefs about teaching

4.4.1 Environment

The findings discussed thus far suggest that the social environment and the students' interaction with the environment are central to the process of professional socialisation. In this process, the students become active students in their own learning and

professional socialisation. This framework explains that professional identity development is a process that is socially constructed rather than individually constructed as the knowledge acquisition is dependent upon the context in which it occurs. The environment that the students experienced are the students themselves that shaped their knowledge about the profession. This finding contrasts with a small scale study in Malaysia that found teacher education students are more inclined towards their co-operative teachers or teacher educators at the teacher training institution. The students benefit from the positive knowledge that they get from the environment.

4.4.2 Former teachers

It is a matter of fact that the students have a set of beliefs about teaching before they embarked on the teacher education programme. Their experience as students for more than ten years has a much bigger effect compared to their four years in the teacher education programme. This study found that the students who lost their direction in becoming a teacher try to adapt to the identity of the referent group. Most students indicate that they are more likely to follow their former teachers' practices during the teaching and learning process. This situation is explained by Giddens (1991) in his Theory of Structuration, that people's actions are based on their existing knowledge.

"The basic domain of study of the social sciences, according to the theory of structuration, is neither the experience of the individual actor, nor the existence of any form of societal totality, but social practices ordered across space and time." (1991; 2)

It is apparent from the findings as discussed in the preceding subsection that the students reflect on the practices of their former teachers in order to act for certain uncertainties. To some extent, the students modified the practices according to their current beliefs on teaching.

4.4.3 School pupil

The main finding in this study highlights the students' consciousness that the school is a place that promotes the development of social reproduction. The students described how their experiences during the school years had taught them about social inequality

among school pupils which became a source of motivation to proceed with teaching. The students who are in their teaching practise described in further detail about this situation compared with their juniors in the first year of study. The students from the second group who are career change and mature students described their experience during the school years and have the advantage to describe the consequences of social inequality in a very practical way. They described their experiences at work and how social inequality affects the young people considerably.

4.4.4 Subsection summary

This subsection explores the students' experiences after they enrolled as teacher education students, particularly their belief of teaching and the level of attachment to become a member of the teaching profession. The main variables that explain the students' experiences are perception about teaching, professional relationship and understanding of learning. It is a noteworthy fact to remember that the students come from three distinct backgrounds; first year students, career change students who have teaching related and non-teaching working experience, and final year students undergoing their teaching practise at schools as part of the requirement to graduate. Thus, their experiences vary according to their engagement in the teacher education programme and working experiences. The students in their first year generally conveyed their beliefs and aspirations mostly from the experiences with teachers back in their school years as students. Their counterparts in teaching training and career change group with teaching experiences have a more realistic involvement in teaching. Being teacher education students in Malaysia generally makes them a part of the teaching profession members, as the society perceives that all of the students will graduate as teachers. This subsection explores the answer as to what extent the students think of who they are and who they will be as a future teacher.

The main finding in this study revealed that students, regardless of their initial reason to embark on a teacher education programme, expressed their concerns on social inequality among children and young people, and that became a main motivation to discriminate against the teaching profession. Their perception of the status of the teaching profession, their belief on professional relationship and understanding of the nature of learning showed their concern on social inequality and the education system

that produces such differences. The students aspire to be an agent for social mobility among children and young people.

Past literature on professional socialisation predicted that Malaysian teacher education students are inclined to have minimal interaction with the environment in the process of professional identity construction. The students also have a greater possibility to depend on significant others and their role models in order to learn about the norms and values of the profession. This is inconsistent with the literature reviewed for this study; a majority of the students are seen as trying to reduce the structural influence in the construction of themselves towards the teaching profession. In contrast with the strong dependency on the structure that potentially prevents them to search for their own meaning of teaching, the students voluntarily move from their status quo and become active students in the system that they become a member of, which refers to the teacher education programme. The students are seen as trying to interact with the environment and fully utilise their experiences in order to develop a positive attitude towards the teaching profession.

This study also found that a majority of the students developed their positive attitude as a future member of the teaching profession during their professional socialisation process. Even though the findings revealed that the students involved in this study are active and constructive in the learning process, however, their active process is guided by the structural culture that dominates the profession. Thus, it can be concluded that the teaching profession is still a profession that is important in Malaysia regardless of the new professions' evolvement as a result of economic development in Malaysia. The students that initially did not have a positive perception in entering the teacher education programme have learnt from their professional socialisation and are aware that they have the related cultural capital. This situation is explained by Bourdieu; an individual who enters a specific sphere, they actually have the capital ownership on the arena, although they are not aware of its existence.

It can also be concluded that the overall students' motivation to teach is relatively higher, particularly for the students undergoing their teaching practise. The teaching practise seems to provide a context to enhance the students' motivation to teach, and build their positive attitude towards the teaching profession. The students that are

emotionally attached to the teaching profession from the start (at entrance level) possess more commitment towards teaching. There are also isolated cases where students sustain or decrease their motivation during the teaching practise at school. This is the limitation of this study that provides an opportunity for further exploration. The students in teaching practise who struggle with their student's behaviour expressed a higher motivation to teach. This finding is contrary with most of the literature which found that a student's unfavourable behaviour can easily discourage a prospective teacher's motivation as their primary goal to become a teacher is related to an intrinsic desire to help students (e.g. Sinclair, 2008).

The investigation on how the students acquire the attitude during their socialisation process also suggested that the students and young generation are the most influential socialising agents. This is consistent with the study from the West which suggests that students are the most influential socialisation agent to develop pre-service teachers' professional identity (e.g. Chambers et. al., 2010; Sinclair, 2008).

4.5 Conclusion of chapter

This chapter discusses the findings related to the first, second and fourth research questions. Two major themes of how the students acquired the identity emerged through the analysis process; firstly they possess the initial identity at the time they decided to apply for the teacher education programme, and secondly, the students developed, organised and changed their identity during the professional socialisation process at the higher education institution. The findings suggest that the majority of the students have a positive attitude towards the teaching profession at the time they decided to choose the teacher preparation programme. Most of them aim for the teaching certificate offered by the teacher education programme which enables them to apply for the teaching profession. However, the perception is not promising as the genuine reasons do not show that the students are genuinely attached to the teaching profession. The analysis found that some students perceived there was a compatibility between their attribute and nature of teaching, and others claimed that they were influenced by parents and former teachers. Additionally, there were students who

decided to apply for the teacher education programme as a result of their positive or negative experiences related to themselves and other young people. The results also suggested that many of the students were attracted to teaching for vocational reasons. As a consequence, the motivation to teach is higher among the students who perceive there is a compatibility between themselves and teaching, the students who have had enjoyable and unpleasant experiences as students with their former teachers, and the students with a passion to teach. Their other colleagues were attracted to teaching because of extrinsic factors and the others do not have an attachment to teaching. Thus, their motive is extremely external, which is to have a job.

Furthermore, the exploration on the ways where the students develop their teacher identity focused on the stages when they became a community of the teacher education programme. Correspondingly, the findings suggest that the students' beliefs of teaching after they became a community of the teacher education programme explained their development or withdrawal of interest in teaching. Thus, the student's attachment to the group (which is teaching profession) has a significant impact on how they view themselves as a member of the group and this reflects their intended behaviour. However, the students who were attached to the profession from their initial entry to the teacher education programme showed more commitment to teach. Their counterparts on the other hand, are in the process of developing the commitment to teach.

To conclude, the students acquired the teacher identity mainly when they become a member of the 'teaching profession' and tried to achieve or maintain a positive social identity; in this case become a member of the teacher education programme. The students that have been admitted to the teacher education programme try to adapt to the demands and characteristics of the group, and are influenced by the socialising agents; environment, former teacher, and students. Although the students' entrance to the teacher education programme involves a very rigid selection process, the students' narrative showed that their professional socialisation process is not determined by the profession, as suggested by Merton (1957).

The subsequent chapter presents the answer for the third research question, particularly in describing the process of self-categorisation among the students during their commencement at the teacher preparation programme.

CHAPTER FIVE DISCUSSION OF FINDINGS

5.0 Introduction

The previous chapter described the students' identity towards teaching in two different phases; firstly, when they decided to embark on a teacher preparation programme, and secondly, when they became a member of the teacher preparation programme community, where they underwent their socialisation process to become a teacher. It is apparent from the previous discussion that the formation of identity is an ongoing process of change embedded in the context (i.e. the teacher preparation programme). Some changes were evident in the data, whereby the students in the midst of their teacher preparation programme, particularly the students undergoing teaching practice, showed more positive identity changes towards teaching compared to their initial reasons for embarking on the programme. A positive social identity captured from the thematic analysis came from one category of the students, as described in Subsection 4.1, which are those who initially wanted to approach teaching after graduation.

The findings also found that the attitudes, in fact, can be conceptualised from both personal and social perspectives. Chapter Four generally described both personal and social factors that influenced the development of teachers' identity in both context; before and during their commencement in the teacher preparation programme. This chapter extends the discussion of the research findings related to the third research question: How do social factors affect the effort to maintain or achieve positive social identity, or encourage individual mobility among students? Following the argument presented in Subsection 2.4, this chapter explores the students' socialisation process framed by the social identity approach. It is assumed that when the students are in the teacher preparation programme, their social identity as future teachers becomes salient if they try to learn and adhere to the teaching norms and values. Thus, it is important to explore how the students undergo the process of self-categorisation. It is important to note that during the interview, the students also mentioned other social categories in

their narratives such as "as a Muslim", "as an Indian", "as an indigenous group from East Malaysia", "comes from poor background", and "low income". As these themes emerged from the data, special attention has been given to find the significance of these social categories towards the students' identity development.

Furthermore, Subsection 2.3 also presented the argument that through the self-categorisation process, the students will adopt the teaching values and norms in order to differentiate themselves from the out-group (i.e. students who do not want to be a teacher). It is argued here that the students will possess a positive social identity towards the profession if they conform to the values and norms of the teaching profession, rather than having been pressured by others to comply with it. The following subsection describes the extent to which the students try to conform to the teaching profession. The discussion includes the process of self-categorisation, how the students learn about the values and norms of teaching and how they assimilate the values and norms of teaching to themselves.

5.1 Students' self definition on teaching

The discussion on how the students define teaching starts with an exploration as to how the students acknowledge the values and norms of the profession. This is consistent with Subsection 4.2, which highlighted the general values and norms perceived by the students. The students perceived that teaching is a noble and respected profession for various reasons, one of the most common being that teachers help to educate younger generations. Besides this, the students implicitly expressed that their behaviour is guided by the National Philosophy of Education, as they mentioned that they have a responsibility to develop the potential of young generations in a holistic and integrated manner, as stipulated in the philosophy. These features serve as a standard of teaching, or the values and norms perceived by the students. Besides these features, it is apparent from the data that the students assigned their former teachers as role models of the profession. As expressed by S7 TP (Subsection 4.1.2), she adored her former teacher as being a very generous teacher and hoped that she would become like her. However, the data also showed that the students did not always view their teachers as role models. Some of them indicated a negative image of teachers, such as an "unfavourable moment

with an irresponsible teacher" [S7 TP], and "a tactless teacher, did not teach properly" [S3 AB], which the students tried to avoid. Other students also mentioned that their parents, who are teachers, also became role models. They acknowledged that their mother or father, who is or was a teacher, is very patient, hard working and helpful with the school children. S9 AB described her father, who was once a teacher saying "He loved what he was doing. He had great enthusiasm for teaching".

The thematic analysis also helped to determine how the students self-categorised themselves towards the norms and values of teaching. Subsection 4.2 described how the students evaluated the teaching profession while progressing in the teacher preparation programme. It is apparent that when the students evaluated the teaching profession, they put themselves as "who am I as a teacher?". The evaluation, however, was different across the groups. The students who were undergoing teaching practice evaluated teaching more in terms of their emotional reaction towards the profession. For example, one student mentioned how satisfied she was when she had the opportunity to teach. S3 AB eagerly expressed that "you cannot have that feeling unless you experience it". Her colleague, S1 BK, stated how she tried to understand the school children in class and "what creates barriers for them (school children) to learn".

Two students who were initially unsure whether to approach teaching or not at the time they enrolled in the teacher preparation programme, developed a positive social identity towards teaching as they felt that they were very important to the school children. One of the students, S10 AB, with open palms and face upward stated:

"Yes, initially, I wasn't interested in teaching. However, during teaching practice, I felt guilty when the school children in my class hoped to learn something from me, but I was not interested to teach. Oh God. Slowly, I forced myself to love teaching. Now I can't wait to become a teacher!"

Similarly, her colleague, S6 AB, had an unfavourable attitude towards teaching at the time she chose the teacher preparation programme. As discussed in Subsection 5.1.2, S6 AB chose this programme of study due to her mother's influence. However, in contrast with S10 AB and S1 TP, the socialisation process had no impact on her engagement

with teaching. With arms crossed in front of her chest, exhibiting defensiveness, S6 AB stated,

"Until now I am not interested in teaching, but I will soon be a teacher, as I will get the credential in teaching. My mother will be very happy. I will do all the teaching work, that is the responsibility, but I am sure I will not enjoy my work."

The students in their first year tried to self-categorise themselves according to their knowledge and beliefs of teaching, which were gathered from the socialisation process and their experiences during the school day. For example, 11 students out of 12 expressed their behavioural intention of becoming a teacher. Among others, some students stated that they should be role models for school children: "responsible to teach the children" [S6 FY2], "exhibit good manners" [S1 FY1], and "determined in teaching" [S3 FY1]. Furthermore, their counterparts, the career change students expressed in more detail what differentiated them from the previous profession when the teaching profession became salient. One student, S6 DPLI1, expressed with tears in her eyes:

"I worked in a bank before, the workload [...]. Oooo, it was very stressful and I was getting very grumpy and over-sensitive. I quit and applied to be a substitute teacher at school. Teaching teaches me how to be more patient. How could you get mad with children who come to school with hope and just want to learn?"

Similarly, her colleague, S1 DPLI2, stated:

"People will change when they become a teacher. Becoming a teacher means that they become a role model for school children. It is different when you are involved in another profession."

The statement presented above gives insight into the factors that may motivate students to self-categorise themselves as future teachers and achieve a positive social identity. Most of the students' narration indicates several strategies to achieve a positive social identity. The students pursued social creativity strategies by evaluating the in-group in a more flattering dimension by highlighting the function of teaching in the society. This has been done by mentioning the advantages of becoming a teacher, ranging from

developing the young generation and becoming a role model. To pursue the strategies, another theme emerges, which is commitment. It is implicitly stated that the students' commitment differentiates how they categorise themselves into specific groups and learn the norms and values of the group. The following subsection highlights the commitment as an important variable in the categorisation process. This commitment implies that the students adopted the interpretive approach of socialization as this approach allows them to become active participants. The strong identifiers give greater commitment to learn about teaching norms and values.

5.2 Students' commitment

There is evidence from the literature review that a positive social identity can be achieved by enhancing an individual's work commitment (Chen and Chiu, 2009; Joireman, Kamdar, Danields and Duell 2006). Further analysis of the students' narratives found that commitment exists among the student teachers towards the teaching profession. The sense of commitment was observed from the students' narratives on their sense of responsibility and attachment towards the profession, and their body language while delivering the stories. The students were then asked relevant questions in order to validate their stories, particularly related to their sense of commitment towards the teaching profession. Again, the thematic analysis helped to classify the different levels of students' attachment towards the teaching profession. In general, the students with a positive social identity had all three levels of attachment towards teaching; higher, moderate and lower levels of attachment. Interestingly, the students who decided not to teach also envisaged a higher level of commitment to teaching. However, students who had not made a decision whether or not to teach, expressed moderate and lower levels of commitment to teaching. Given these points, almost all students in this study constructed their anticipated professional identity based on the same platform, which was a commitment towards the teaching profession.

The narratives showed that the students envisaged their sense of engagement through their inspiration to take responsibility towards three different parties, namely, the students, society and the profession itself. The prominent findings of this study revealed that the students' higher levels of attachment were related to the desire to serve others, particularly students and society. These findings are consistent with the students' beliefs, as discussed in Chapter Four, where students considered themselves as agents for social mobility who would try to minimise the social reproduction resulting from the educational process. The findings suggest that accessibility to the profession is very important in developing a positive social identity.

5.2.1 Commitment towards prospective students

The students voiced different degrees of responsibility and attachment towards their main audience, i.e. the prospective students. The main themes that emerged from the analysis on commitment towards the students related to the process of educating are the desire to act as secondary parents to their students, the desire to attract students to learn, and the perception that teachers must be fair in their relationships with the students.

5.2.1.1 I want to educate the younger generation

The overall response related to the students' commitment with regard to their perception is that they have a responsibility to educate prospective students. The students, regardless of the group, agreed that they were responsible for 'educating' the students rather than 'teaching'. For instance, one student in her first year stated that "For me, the teacher is not just teaching, but is educating the young generation." (S2 FY1)

There was consensus among all students on the definitions of 'educating' and 'teaching'. Students perceived that 'teaching' is related to disseminating knowledge and giving information about the subject matter. In contrast, students referred to 'educating' as a process of building the human capabilities of the students, thus focusing on the development of physical, emotional and spiritual aspects, rather only on cognition. During the focus group discussions, the students from all groups repeatedly expressed their desire to educate students. One of the intended outcomes of the educating process is to develop students with positive values and accepted behaviours. The students stated several values such as good *akhlaq* (Arabic term referring to virtue and morality), spirituality, generosity, courtesy and respect, which they expect to promote and disseminate among their prospective students. One of the students in teaching practice

expressed, "It is my intention to change the students, particularly their character." [S9 SK].

S9 SK stated that she was aware that some of her current students came to school with inappropriate manners, thus she aimed to shape her students in cultivating moral values. This student observed inappropriate behaviour including irresponsible learning, and a tendency to show negative behaviour such as smoking and absent from school. Moreover, the students aim to educate their students to be better people, even though they realise that some of these students are less academically able. One of the students expressed her passion to educate her current students in the classroom during the teaching process:

"There are students (during teaching practice) who do not perform well in their examination. Yes, they are not competent academically, but I tried to develop them to be useful human beings." [S1 BK]

The students' intention to educate rather than teach implies that they reject the current national practice that puts academic excellence as an important factor in determining students' success and the prerequisite for admission to the next phase of education level, thus promoting social reproduction in school. Even though the majority of the students had good final exam results (during their school years) or satisfactory grades (during their undergraduate years), being in the professional teaching community made them aware that outstanding results would not help to prevent students from being trapped in the social problems that are plaguing the Malaysian society.

Moreover, the findings of the current study do not support the previous research on the students' approaches to learning. The literature reviewed in this study suggests that the majority of Malaysian students, both in school and higher education institutions, still demonstrate rote learning, which refers to the memorisation of facts, despite continuous efforts to promote a constructivist approach to learning. Explicitly, the traditional approach focuses more on the cognitive aspects. It is apparent that the majority of the students in this study declined the tradition of rote learning, and suggested that students had to be induced by educating them with different values in order to foster their own learning.

Although the majority of the students eagerly expressed their sense of attachment towards the teaching profession, there was evidence that a small number of their colleagues were not actually enjoying their status as a prospective teacher. They had a positive feeling towards the teaching profession; however, they were unwilling to actively get involved in teaching activities, such as approaching students with new learning methods. There were two students who voiced a negative sense of dedication towards teaching. Both of them were undergoing teaching practice in school:

"I do not love my teaching job. Nonetheless, I consider myself as a responsible person. I do my work (during teaching practice), I prepare my teaching material, and I teach students the content of the subject, to which I was entrusted. I am just responsible for my title as a teacher." (S7 AB)

"Teaching is not my first choice, yet it is a secure profession. Schools are everywhere; young generations need to go to school, so I will still have a job. If I am posted in any school, I will do my job. I will do it because I have to do that. Children need to learn something from school. I will make sure they will learn something." (S2 HD)

It was noted that both S7 AB and S2 HD did not have an initial interest towards teaching, and their decision to embark on a teacher preparation programme was influenced by their parents. In the case of commitment, S7 AB related her desire to teach with the status of the profession itself. This situation demonstrated that the professional status of the teaching profession, as reviewed in this study, provided the main motivation for the teacher preparation students, despite their low levels of commitment, to become part of the profession. Her peer colleague, S2 HD, also expressed a low degree of attachment towards the profession; however, her disengagement was moderated by a desire to serve the students. This was shown when S2 HD mentioned "children need to learn something from school" in her narrative. These two situations suggest that even though the students had a low level of attachment towards the teaching profession, there were internal forces in the profession that bound them to it. However, these two students were not inclined towards the 'educate' aspect, unlike the majority of their colleagues involved in this study. Reviews

of other professions in this study have not found such internal binding, excepting the employment opportunities and rewards offered by the profession.

5.2.1.2 I am a second parent to the students

Besides the determination to educate their prospective students, the students also expressed the responsibility to become in loco parentis²³ (in the place of the parent) while at school. The students in their teaching practice understood that they have a responsibility for their students' safety and well-being at school, as underlined in the Malaysia Child Act 2001²⁴. S7 SK, who was undergoing the teaching practice, stated that, "Teachers are students' parents when they are at school. We (teachers) must take reasonable care of them." (S7 SK).

The expression of 'must' in the statement above suggests that the student was committed to her potential role as a teacher. S7 SK's perception of her role as a second parent, also seen from the perspective of 'to educate', thus suggests the strength of the 'education' rather than 'teaching' process. Similar with S7 SK, who was undergoing her teaching practice, the students who did not yet have teaching experience also understood that they have a responsibility towards their prospective students. For instance, one first year student, S5 FY1, expressed that, "Teachers are the mother or father of students at school". Similarly, one career change student, S1 DPLI1, added that, "We have to treat students the way we treat our own children".

The representative statements from the students showed that they understand the importance of maintaining a safe and family atmosphere, particularly in the classroom. There are a few implications related to the students' desire to educate and become in loco parentis to the students. Firstly, the students expressed their motivation to like and love the children. The findings showed that the students' understanding of this trait was consistent with the literature reviewed in this study, which suggests that teaching is a

²³ Teachers as students' guardians. The Malaysia Child Act 2001 (Act 601): 'Guardian' in relation to a child, includes any person who, in the opinion of the Court for children having cognisance of any case in relation to the child or in which is concerned, has for the time being the charge of and control over the child. 'Child', means a person under the age of eighteen years. ²⁴ The Child Act 2001 is a main law related to children.

profession that requires its members to love the children (Aldemir and Sezer, 2009). Secondly, the students indicated that they would develop and keep their positive values, behaviours, and accepted appearance as determined by society, in order to motivate their prospective students to follow their positive attributes. As discussed in Chapter Four, the students perceived that they are a role model for their students. The students, through their narratives, expressed their commitment to be a role model for their students, as described in the preceding chapter. One of the students cynically stated:

"There is no place for people with unacceptable behaviour²⁵ to become a teacher. Do you want to be like a mother crab teaching her children how to walk straight²⁶?" [S3 AB]

The findings of this current study are consistent with those of other research in Muslim countries (Mohd Dahan and Ghazali, 2007; Goh and Atputhasamy, 2001; Kahveci, 2009; Minnis, 1999; Mogra, 2010; Tok, 2011), which award the teaching profession and teachers as having considerable virtue in society.

5.2.1.3 I want to attract students' attention to learn

Another prominent theme that described the students' commitment was their inspiration to make learning meaningful for their prospective students. In conjunction with the students' beliefs on learning, as described in the preceding chapter, they expressed in detail their inspiration to reinforce the learning process, particularly for the group of students who have lower academic abilities. The students indicated their enthusiasm to help in learning the subject matter and also to motivate their prospective students to learn. The students identified five main interrelated approaches to attract their students' attention: encouragement, friendliness, sense of humour, creativity and forcefulness.

First of all, most of the students believed that they had to encourage their students to be more self-confident in learning and minimise their despair if they could not yet do something. The students used different strategies to encourage their students' learning. Some of them thought that words of encouragement would assist the students' learning process. Some other students thought that friendly eye contact, smiling, helping

-

²⁵ Unacceptable behaviour according to Malaysian culture and customs.

²⁶ The Malay idiom, meaning teaching by example.

behaviour and facial expressions would make their prospective students feel comfortable and take an active role in the process of learning. One of the students in her first year shared an unfavourable experience of learning. She expressed that she was not interested in a certain subject because her former teacher did not encourage her or her friends to learn. She mentioned that one of her former teachers just stood at the blackboard and taught: "She never tried to understand and encourage students to learn. Until now, I am not interested in the subject matter taught by that teacher" [S5 FY1]. S5 FY1's unpleasant experience in her school years had stifled her interest in exploring certain topics, and strengthened the argument that teacher based learning is still practised in Malaysia.

Besides direct encouragement, the students intended to bridge the gap between teacher and students. As reviewed in this study, there is a significant power distance between teachers and students in Malaysia (Hofstede and Bond, 1984). The students agreed that this situation still happens, based on their experience in their school years and their observations during teaching practice at school. For example, one student undergoing her teaching practice expressed that senior in-service teachers gave more concentration to teaching and learning the subject matter. S3 AB, expressed that:

"Some of the senior teachers do not care about the students in lower ability classes. Yes, it is hard to teach the students in this class. I believe that I should take an interest in those students, being warm and courteous."

Her counterpart, S10 AB, who initially did not have an interest in teaching, also felt overwhelmed when appreciated by the students during her teaching practise: "They (students) say that I am very kind! I realised that they are always happy when I enter the classroom".

Similarly, students in their first year, including the career change students, also imagined that they would be teachers who could "cheer a student's soul", support their learning and care for their needs. They also mentioned the need to "help as many students as you can" [S5 DPLI2], and that "teachers should know students' personal problems that hinder them from learning" [S5 FY2]. Equally important, students also agreed that teachers with a sense of humour had the ability to engage, activate and

connect with students in ways that allowed them to present memorable learning experiences. Consequently, the students also believed that a sense of humour would evoke laughter, smiles and excitement for the teachers. S5 SP expressed that:

"I believe that, as a teacher, I need to have a sense of humour to relieve tense classroom situations before they become disruptions, and build a strong relationship with my (prospective) students."

Additionally, some of the students were tremendously aware that in certain circumstances, teachers have a personal attachment with students, including giving them support when they have problems related to their private life, which may affect their learning process. The students' perceived that they needed to know their students and the lives their students live. Furthermore, the students also declared their intention to help their prospective students with learning beyond school hours. For example, one student stated that:

"Teachers' efforts are not only applicable in school, but outside the school hours. I will allocate time if students wish to come to my house and ask about homework, or stay longer at school. It is for their own sake." (S2 HD)

The overall response to the willingness to work beyond the prescribed task was tremendously positive. Another student, S1 DPLI1, a career change student, voiced that she literally foresaw teachers working for 24 hours a day. Explicitly, she believed that the title of teacher was not only applicable when they were at school, but it extended into the teacher's personal life. This implied that teachers are always expected to help students, even though it is not directly related to the teaching and learning process. Furthermore, teachers need to promote ethical behaviour even beyond the school premises. This is consistent with the students' perceptions that recognised teachers as a role model for the students. Thus, the second theme that resulted from the thematic analysis suggested that in order to help students, the students assumed that teachers have to prepare themselves for working beyond the designated working hours. Moreover, they also believed that teachers who are favourably disposed, and intend to help and support their students would be successful in attracting the students towards learning. In other words, the students had a desire to do their best for the sake of their

prospective students. This is consistent with the famous Malaysian quote related to teachers; "a good teacher is like a candle, it consumes itself to light the way for others."

Thirdly, the findings from the thematic analysis also suggested that the students communicated their willingness to transcend traditional ideas in teaching to create meaningful new ideas, which help their prospective students' learning. One of the efforts that appeared in the narrative was the act of being creative. For instance, one of the students shared her experience in her teaching practice:

"We have to be creative. I am not a creative person, but I will try my best. When I first came to this school for teaching training, I tried to be a fierce teacher because I wanted students to listen to me and learn. Yes, they were listening, but it seemed that they did not understand anything. Then I changed my approach [...]." [S2 SK]

S2 SK is among many of the students that responded to the need to be creative in the teaching and learning process. The majority of the students were aware that students at school have different learning abilities, thus teachers have to use their expertise and creativity to determine teaching and learning strategies. The strategies, according to the students, must stimulate their students to be active and interactive. The students also felt that they must act fast when they find that their students are not attracted to learn. The students' expressions of their efforts were consistent with their vision of becoming an agent of social mobility. Another student in teaching training shared her efforts to encourage students to learn. S2 BK, majoring in mathematics education, and who had her teaching practice in an urban school, expressed that her students did not enjoy mathematics because they perceived that it is an unexciting and hard subject. S2 BK stated that:

"I think there is no fun in teaching mathematics. Mathematic deals with facts and figures. But applying mathematical games helped the students to understand complex equations and problem solving. I started to surf for mathematical games on the Internet, bought relevant books and thought about new games related to my topic."

Similarly, one of the career change students who had been a substitute teacher before enrolling in the teacher preparation programme stated:

"I had the experience of teaching students in a lower academic ability class. It is a shame that some of them still cannot read, at the age of 13! It was extremely challenging to teach them a science subject. What I did was I put all the teaching material in diagrams. Yes, students very easily recognised the diagram and started considering the things charted in the diagram." [S1 DPLI2]

Although some students expressed that being creative was an immense challenge for them, their body language and facial expressions when narrating their passion about their role and image as a teacher teaching demonstrated positive signs that they were prepared for the profession. They presented themselves as 'blissful', 'radiant' and 'appealing' when sharing their inspiration during the interview. S2 TP, who initially had no interest in teaching at the start of the teacher preparation programme, suggested that the desire to sacrifice guided her to work hard for the betterment of the students. She expressed:

"I am never satisfied if they (students) do not get what I am trying to get across. I keep thinking of a way that would be easy for them to understand."

Furthermore, one of the students expressed that:

"It is very enjoyable being a teacher (trainee). It is not a routine job. You think about something new every day in order to help students develop themselves." [S1 BJ]

The group discussion among the first year students also found that they were aware of the requirement to be creative when they become a teacher. Their awareness came from their experiences in their school years. Some of the students whispered that their unsatisfactory experiences with their former teachers' teaching approach affected their motivation to learn. Hence, the majority of them were determined to find an approach that suited students' learning and will deliver meaningful outcomes:

"I remember one of my teachers, being possibly the most uninteresting person I had met. Every day our learning activities were the same: reading a chapter, copying down notes from the blackboard. [...] I just memorised information, took a test on it and then forget everything." [S4 FY1]

The students quoted above eagerly narrated their plan to implement various strategies that would attract their prospective students to learn. It is apparent from the narrative that the students were concerned about how to support students with academic difficulties. They signalled that this group of students would learn better if the teaching approach was more towards the 'teaching by doing', rather than 'teaching by telling'. The students' inspiration to attract students to learn suggested that there was a sense of commitment among them in encouraging students' learning. In particular, the students' views on encouragement, creativity, and their willingness to help and motivate prospective students helped the students to construct their own knowledge. It is apparent from the findings that some of these students were inspired to promote or consistently promoted the element of constructive learning by encouraging the students' self-learning; this is something that is formerly not well implemented in Malaysia. Besides that, encouragement in learning also helped the students to become motivated in learning, thus facilitating the social mobility process at school.

5.2.1.4 Education for all

The previous chapter discussed how the students realised that school institutions became a context that promoted social reproduction among the young generation. They believed that the current practice at school gave an advantage to students with high academic abilities. They also believed that some of the teachers became agents that promoted the process of social imbalance. This finding supported the statement that Malaysia continuously focuses on academic achievement to measure students' success. One career change student stated that:

"Our philosophy of education²⁷ states the aim to produce students that are balanced in terms of intellectual, physical, emotion, spiritual and

_

²⁷Malaysian Philosophy of Education: Education in Malaysia is an on-going effort towards further developing the potential of individuals in a holistic and integrated manner, so as to produce individuals who are intellectually, spiritually, emotionally and physically balanced and harmonic, based on a firm belief in and devotion to God. Such an effort is designed to produce Malaysian citizens who are knowledgeable and competent, who possess high moral standards and who are responsible and capable of achieving high levels of personal well

social aspects. The philosophy is perfect for students' development, but the implementation (for me) is more towards the development of intellect alone. Teachers need to think about it." (S1 DPLI1)

So far it seems from the findings discussed that most of the students were inspired to promote social mobility through their commitment to educate, being a second parent for students, and attracting them to learn. Above all, the findings also suggested that the students have a desire to be fair to their students, regardless of their academic achievement in school. They believed that by not discriminating against the student, the chances for them to succeed were promising:

"For me, if the students do not have the capability for academic achievement, or are experiencing learning difficulties, that is not the end of the world. There are a lot of opportunities for this group of students. I like the proverb – do not give a fish to our student. Teach them how to fish and they can use the skill forever. Motivate them and give them the appropriate skills." (S4 SK)

S4 SK's counterparts shared an experience in her classroom during the teaching practice that she termed as a 'fresh start' for students to have an equal chance in their self development:

"Unfortunately, I got a class full of 'undesired students' in the school. They are not capable academically or respectful to teachers. Speechless, I tried exceptionally hard to be their friend, not just a teacher. I found that this group of students is extraordinarily creative. They have been left out because they did not get any 'A's in their subjects. School nowadays really focuses on academic aspects. We try to do something that will make the school proud of us. We did it! We made a mural, and won a competition in a recycling campaign. Aha, I realised that the students only need attention and motivation." (S5 TIWA)

being as well as being able to contribute to the harmony and betterment of the family, the society and the nation at large.

128

One student, S1 SK, who decided not to teach after graduation, also voiced her concern about the ways teachers approach students with lower academic achievements. She stressed that students must be treated equally, regardless of their academic level. Furthermore, she was also concerned about equality in education among multi-ethnic students at school. She is from an indigenous ethnic group from East Malaysia. Even if she decides not to teach, she expressed uncommonly strong convictions that fair treatment and education for students is mandatory, and not a choice. Moreover, one student majoring in special education expressed her concern regarding special needs students:

"[...] students with special needs are not able to pursue their studies to a higher level, so what they need is skills, so that they can survive in the future." (S3 BJ)

S3 BJ, however, was extremely pessimistic about the opportunities for students with special needs to further their education at a higher level. The Person with Disability Act (2008) established that people with special needs cannot be excluded from formal and public education despite their disability. Although there are not many disabled students entering higher education, they increase every year²⁸ ("Ministry of Education Malaysia", n.d-b). Thus, the support from the teacher during their school years is crucial to help them qualify for entry in higher education, which in turn will provide more opportunities for them in terms of cultivating their knowledge, developing social skills, and obtain higher academic qualification. S3 BJ's perception of the ability of students with special needs has developed since she enrolled in the teacher preparation programme. Her experience in teaching practice made her more responsible to students with special needs. She mentioned that, "I love the students, and I hope that I will make a difference in their lives".

5.2.1.5 Conclusion for the subsection

This subsection described and discussed the students' commitment towards their prospective students. There were four main themes that emerged from the students' narratives on their perception and depiction of their attachment and responsibility as a prospective teacher. Firstly, the majority of the students expressed their commitment to

²⁸Total of disabled students enrolled in higher education institutions was 979 (2009), and 1115 (2010) (Ministry of Education Malaysia, n.d.-b)

comprehensively develop their prospective students. Secondly, they were willing to act as second parents while their students were at school. Thirdly, the students were also committed to applying a different approach in order to attract their prospective students to learn. Finally, the students expressed their commitment to be fair to students regardless of their academic ability and their ethnicity. The next subsection describes the students' sense of commitment towards society, with reference to parents, the local community and the nation.

5.2.2 Commitment towards society

In addition to the students' expressions regarding their commitment towards their prospective students, several themes on the commitment towards society and the country also emerged from the thematic analysis. Generally, students believed that a teacher is a member of the community, thus in addition to the formal role prescribed for a teacher, their intended behaviour is also influenced by the needs of society. The students believed that teachers' actions and gestures are always observed by the society, consistent with the status of a role model that society has conferred on the teachers. First year students, S1 FY2 and S5 FY2 believed that "teachers have to be involved in the community activities in order to build a close relationship with the local community in the area which they work". According to the students, the advantage of having a close relationship between the community and teachers helps teachers to understand their students more. Some of them expressed how they felt entrusted as a teacher, for instance:

"Parents and society are the school's clients. They should have confidence in the services offered by the school. They invest their 'belief' that a teacher will educate their children. Yes, we will. Tell them we will." (S3 AB)

Her counterparts, who were in their first year, also had a positive feeling about their commitment towards society, particularly the parents: "Effective parent and teacher communication is essential for a teacher to be successful" [S2 HD].

Moreover, the students perceived that the teaching profession is unique because it directly contributes to the community as well as the nation. The students believed that

the greatest contribution of teachers towards society is being agents in delivering national values that are essential for living in a collective society. As mentioned in Subsection 3.2.3, Malaysia is a multiracial country and its education policy indicates that all residents of Malaysia have equal access to formal education provided by the state. Despite some students' desire to help their own ethnic group, as previously stated by S5 DPLI1 in Subsection 4.13, they believed that students should be treated equally in the classroom. This was consistent with the students' belief that education is for all, as described in the Subsection 5.1.1.4.

The narrative provided strong evidence of the students' inspiration regarding their role as agents in delivering national values; for example, "Without education, the process of nation-building cannot be achieved. Teachers have a crucial role" [S2 BJ]. S2 BJ was aware that teachers are necessary to build or structure the national identity of future generations. As a multiracial country, it is extremely important that formal education in Malaysia assists the process of developing and sustaining a united and integrated society, consistent with the national inspirations. This responsibility shows the role shouldered by the teachers. One of the students, S3 BJ, expressed her enthusiasm regarding this matter, with the statement, "emphasising values across the curriculum". According to S3 BJ, apart from educating the students towards being a successful person, teachers must promote and internalise the values of a caring attitude, ethnic tolerance, mutual respect and cooperation among the students, regardless of ethnicity and religion, in order to foster unity.

Other responses on this issue were expressed by S3 SP, who was also undergoing teaching practice at school. Contrary to S3 BJ, who believed that instructional medium in the classroom would help to promote a sense of unity among multiracial students, S3 SP considered active involvement in co-curricular activities as more significant to constructive nation building, as well as creating unity. Co-curricular activities²⁹ refers to any non-academic activities that gives students the opportunity to increase, consolidate, and practise skills and values learned in the classroom. S3 SP was actively involved in a physical development activity, the 'Girl Guides', during her teaching

_

²⁹ It is compulsory for each student in Malaysian primary and secondary schools to be involved in at least one co-curricular activity offered by the school.

practice; she expressed that, "the activities emphasise teamwork and a sense of unity both within the school and local community".

5.2.3 Commitment towards the profession

The students expressed their loyalty and sense of belonging towards the profession itself. They expressed a responsibility to elevate the teaching profession. The majority of those interviewed suggested that they would sustain the status and reputation of the teaching profession as perceived by society. As reviewed in this study, society views teachers as role models and respected community leaders (Mogra, 2010). The overall response from the students showed that they understood the responsibility to sustain the privilege awarded to the teaching profession. The students expressed their awareness of the accountability towards the profession with a promise that they would carry out the responsibilities with full dedication. Representative statements included, "honest and trustworthy in carrying out the responsibilities" [S8 TP], "sacrifice for the sake of the students' [S2 TP], and "proactive" [S9 SK].

Other responses included the students' determination for self development, as they believed that they would be their students' source of reference. The students described themselves as a source of reference, instead of an information provider or content expert, as normally described in the literature regarding the function of teachers in school (Merton, 1957). This implied that the students realised that the teacher is not a primary source of knowledge, thus rejecting the functionalist approach to learning reviewed in this study. One excerpt from student S1 SK clearly stated her beliefs on a teacher's role:

"We must realise that the new generation of teachers are no longer perceived as 'expert people'. In the new era of information technology, sometimes students know more than their teacher, regardless of the accuracy of the information. Teachers nowadays have to prepare themselves to be challenged by students. This is the beauty of learning, when there are arguments and both teachers and students have to use their critical thinking to build new knowledge and solve problems."

This prominent point expressed by the students showed their understanding that commitment towards the profession is important to the teachers' subject knowledge. The majority of the students felt that teachers have well developed subject knowledge, which is used to inspire most students and contribute to their satisfactory progress. However, some of the students expressed their difficulty in managing and delivering the subject knowledge. One student majoring in mathematics expressed the challenges with regards to subject related knowledge:

"We are trained (during teacher training) in a higher level of mathematics calculation. It is difficult for me to teach a lower level of mathematics, though." [S2 BK]

Although S1 SK's comment suggested that the teacher preparation students were preparing themselves towards self-learning, S2 BK was more inclined to suggest that teacher preparation programmes should prepare their trainees with sufficient subject knowledge, but that this should be restricted to the syllabus offered in school. In this case, it was suggested that some teacher preparation students were inclined to absorb only the relevant information during the teacher preparation programme, thus the functionalist approach of socialisation was more desirable. Treating this case in isolation, the majority of the students, however, communicated their commitment for long life learning and assumed that teachers' creativity, as described in the preceding subsection, could overcome the discrepancy related to subject knowledge.

The thematic analysis also revealed that a majority of the students understood that they were responsible for their own career development, which in turn would give them a competitive advantage in performing their role as a future teacher. The students mentioned the following among others: "sharpened the thinking skills" [S6 AB], "set up a learning community with experienced teachers" [S3 HD], and "proceed to second degree" [S6 AB]. The overall response to career development suggested that the students agreed that by setting up a learning community, their commitment to become a member of the teaching profession increased. One of the students mentioned that, "While in the teacher preparation institution, the environment set our minds that this is the place to develop a teacher community. The local community started to address us as

'cikgu', the institution dress code, all together have motivated me to think of teaching as a profession" [S1 HD].

Furthermore, some students rewarded the profession as something that gave them challenges, excitement, and personal rewards, and these positive values enhanced their commitment to become an outstanding teacher, thus contributing to the growth of the teaching profession:

"I am looking forward to coming to school every day [...] I am very anxious to wait for the next day, I appreciate students' feedback and I am really happy when we can achieve the learning outcomes together." [S6 TP]

S6 TP embarked on the teacher preparation programme without any interest in the profession, and her decision to apply for this programme was solely influenced by her friend. Towards the end of the teacher preparation programme, she felt an attachment to the profession and the main socialisation agents, which increased her motivation to teach, were the students during her teaching practice at school. Furthermore, her counterparts, who initially did not have an interest in teaching also gave highly compelling statements that motivated her to do better as a teacher; for example, "Do not concentrate too much on what we will get from this profession. It is better if we think of what we can contribute to it" [S3 SP].

Likewise, students who were inclined to teach for vocational reasons, also stated their sense of accountability; however, the element of teaching with 'heart and soul' was missing from this group of students. As stated in the previous subsection, S7 AB³⁰ has no interest in teaching, although she is in the final year of her teacher preparation programme. S7 AB stated that she tried to inculcate the interest to teach, however, she failed, as she does not have the ability to sacrifice her interest in other professions. As a note, these students typically enter the teacher preparation programme because of great parental influence. She, however, survived until her fourth year, as she had a lot of interest in her major subject matter, which was physical education. This finding

_

³⁰ As described in Chapter Five, this student was offered a place in a teacher preparation programme by chance, as she mistakenly put an educational course on her electronic application form for admission to higher education institutions.

supports Arja Virta's (2002) study, which finds that the subject area precedes the teacher preparation students' decision to become a teacher, rather than an interest in the profession itself.

5.2.4 Challenges for students' nurturing sense of commitment

The thematic analysis also revealed the pattern of the students' commitment, and suggested that the most destructive challenge for some students to develop their commitment was a lack of support from the in-service teachers, particularly during the students' teaching practice. Society and practicing teachers have doubts about the interest of the new generation of teacher preparation students in the profession. This stigma continues up until this group of trainees had their teaching practice in school. One of the students in teaching training sadly shared her experience:

"When I want to do something different in the teaching approach, such as presenting a new type of teaching aid, they (senior teacher) said that I did that purposely to get an excellent grade in my teaching practice. Did they really know why I did that? I did that to assist students' learning." [S5 BJ]

The other students also claimed the same situation whereby, "Senior teachers underestimate our ability to teach" [S4 SK].

From the interviews, it was found that the students felt distracted and demotivated by such destructive challenges. They described the challenges as 'turning down' their inspiration to develop themselves to become a teacher. Equally important, this stigma created a gap between trainee and practicing teachers at school. Thus, it can be concluded that the practicing teachers in school at some point may neglect the teacher preparation students' sense of commitment to become a teacher, thus rejecting the assumption that they are the main socialising agents for the students' professional socialisation. Indeed, one student mentioned that:

"Senior teachers add a layer of complexity to me to learn about teaching. Rather, their behaviours are more stressful than students' behaviour in the classroom." [S3 BK]

5.2.5 Conclusion of the subsection

Prior reviewed studies have noted the relationship between the motives to enter the teacher preparation programme with a commitment to teaching. Contrary with the findings from most of the past studies reviewed, this study found significant differences in the relationship of motive to embark on a teacher preparation programme and the students' commitment. Most of the studies found that the teacher preparation students' initial interest in teaching was positively related to a strong commitment towards teaching; however, this study found that the students' understanding of the needs of young people, their own experiences as a school student, and the image of their former teachers, all had a significant impact on their commitment to teach. These findings support the assumption that the development of a professional self-concept is constructed in the social sphere.

The findings also revealed that in terms of the level of attachment, the students in teaching practice developed more realistic views and opportunities to assume the role of teachers. However, some of them felt that they did not develop their subject knowledge well, even though they had nearly finished their teaching practice at school. In contrast, their counterparts in their first year appeared to have limited opportunities to assume the role of teachers. This was evident when only a few of the first year students, including their career change counterparts, responded to the discussion on commitment to the profession. Thus, it can be concluded that being a member of the teaching community, and the trajectory of holding a title of prospective teacher, had enhanced the students' commitment to the profession. This situation explained why the students undergoing teaching practice felt more attached to the profession, rather than their counterparts in their first year.

Furthermore, in response to the fourth research question, further analysis of the students' experiences found that they not only expressed their perception of teaching, but also the social identity that they had built and negotiated, based on their attributes, motives, beliefs, experiences and a favourable or unfavourable attitude towards the profession. In particular, students expressed their feelings about their abilities, and expectations as to what they would encounter as a future teacher. In the next subsection, I analyse the students anticipated social identity as teachers, or the professional identity

and how they tried to construct and negotiate their identity as a future teacher and a member of the teaching profession. This is consistent with the findings from the thematic analysis, which grouped together the students' attitudes to the teaching profession with themes on how they understood their relationship to teaching, how this relationship was constructed over time and context, and how they understood possibilities in the future. The students' abilities to understand future possibilities meant that they tried to negotiate and improvise aspects of their anticipated professional identity, as a future teacher (O'Connor, 2008).

The analysis of students' narratives suggested that the teacher preparation students' sense of professional identity was built on their commitment towards the teaching profession. The students explained that a strong sense of responsibility and attachment to the teaching profession was a prominent feature of expecting and demonstrating their commitment, and explained the construction of their professional identity. The next subsection describes how the students' sense of commitment, as indicated in this subsection, relate to five professional identities that the students tried to negotiate based on this commitment:1) professional orientation, 2) personality, 3) self-efficacy, 4) autonomy, and 5) emotion. All these variables emerged from the data. These identities either strengthened or weakened the commitment to teach, thus they were immensely influential in shaping and reshaping the students' sense of becoming future teachers.

5.3 Sense of professional identity that fostered the students' commitment

Further exploration in the thematic analysis revealed five key characteristics of professional identity that students tried to negotiate, based on their statements of commitment, as described in Subsection 5.1. They are professional orientation, personality, self-efficacy, autonomy and emotion.

5.3.1 Professional orientation

The first characteristic of professional identity that the students tried to negotiate was related to their professional orientation. Within this study, professional orientation refers to what teacher preparation expects of students' contributions and their role in the

teaching profession. Previous studies, such as those of Schepens, Aelterman and Van Keer (2007), and Lamote and Engels (2010), suggest that professional orientation is one of the products of professional socialisation process that informs the prospective teachers' sense of identity. At the entrance level of a teacher preparation programme, professional orientation describes individual commitment as to why the students chose the programme. The framework of work orientation drawn by Rose (2005), perfectly fits with the findings of this study. Rose classifies work orientation among university students into three categories: whether students aim to seek work for self-actualisation, to support everyday life, or for its economic return. It is apparent from the data of this study that the most prevailing commitment was related to the students' perception that the teacher preparation programme provided access to become a teacher and the privilege that offered teacher preparation graduates a guaranteed post as a teacher. Thus, the majority of the students inclined towards the second and third point of Rose's work orientation framework. As the students engaged in the teaching profession community by becoming teacher preparation students, they were more inclined to describe their commitment in terms of the role of teacher. The students' commitment, found in the preceding subsection, confirmed the classification of extended professional orientation, as Hoyle (1974) posits.

Subsection 5.2 discussed the findings that the majority of the students for this study perceived that their role was becoming more extended from what is traditionally posited by Hoyle (1969); a teacher's role is to place emphasis on moral training and instruction. There are prevailing characteristics that showed the students movement into the extended professional orientation. Some of the students communicated their willingness to work with their students after the school day. It suggested that the students viewed themselves as not only working to meet the teaching and learning needs in the classroom, but continuously working on the development of students' emotional and spiritual aspects. The students also perceived that their role had shifted from emphasising what and how to teach in the classroom, to a more extended role with greater autonomy. For instance, the students' inspiration to become second parents for their students was beyond a traditional teacher's core professional activities, which focused on immediate responsibilities and classroom concerns. Additionally, the students' inspiration to unite the students in a multiracial society was also seen as resulting from their extended professional orientation. Their inspiration to be involved

with the society also indicated that they were willing to support their extended professional orientation.

However, there were a few weaknesses that restricted the students from fully shifting to the extended professional orientation. The students were found to have limited abilities to acquire a wide range of knowledge and skills to cope with the new demand. This was evident when the students were in the process of developing themselves to become an active learner. Being in an education system that promotes rote learning for all ages, the definition of extended profession may not be the same as it is in Western research. Furthermore, the students also imagined that performing other roles that were not associated with students created a burden for themselves. The majority of the students from all three groups of respondents stated that they did not aspire to contribute to the administrative or decision making process in school. However, they were aware that teachers would have to do clerical and administrative work. The students expressed that "I do not want to spend less time with students because of being scheduled to do administrative work" [S3 TIWA].

5.3.2 Personality

Secondly, students tried to relate their professional identity with the unique qualities that every teacher should have. As described in Subsection 5.2, the students continuously expressed characteristics such as liking children, having a sense of humour, caring, and being friendly. One interesting point observed from these findings was that the students seemed less inclined to separate their personal and professional self-concept as a teacher. It was apparent from the narratives that the majority of them were determined to bring the professional self-concept into their private lives. This was evident when the students consistently stated that teachers are role models for their students, and their perception that teachers work beyond the designated hours and school premise.

Furthermore, the students were also determined to change their personalities to suit the description of teachers, as perceived by society. This was clear when some of the students perceived that the teaching profession allowed them to change towards a more

positive image, as they would become a role model, as described in the preceding paragraph.

5.3.3 Self-efficacy

The third characteristic that the students tried to negotiate was a sense of efficacy. The students' sense of commitment revealed the sigh of effort and persistence that students shared in their narratives. The students' efforts mostly came from their desire to help others, particularly the students and society. Tschannen-Moran and Hoy (2001) define teachers' self-efficacy as a "judgement of his or her capabilities to bring about desired outcomes of student engagement and learning, even among those students who may be difficult and unmotivated" (p. 783).

There were varieties of effort that the students expressed in their narratives. The main motivation for such efforts came from the students' desires to minimise social reproduction at school.

5.3.4 Autonomy

The students also believed that, as a teacher, they would have a certain autonomy to work as a teacher. Autonomy refers to an individual feeling that they have control over an aspect of their life (Bogler and Somech, 2004). In the literature, autonomous people are defined as being intrinsically motivated, believing that they can control their decision making and take responsibility for their outcomes (Grolnick and Ryan, 1989; Niemiec and Ryan, 2009). The students expected that they would be given the autonomy to conduct their teaching and learning process. They communicated this desire by voicing the importance of creativity to assist students' learning. Autonomy is also associated with learning outcomes, in that support will result in better learning outcomes, and frustration when there are poorer learning outcomes (Niemiec and Ryan, 2009). Thus, the destructive challenge expressed by the students could be a barrier to becoming an autonomous teacher.

The students in the teaching practice believed that their teaching practice or practicum gave them great experience of teachers' autonomy and the school climate, which may

shape their identity. Hakanen, Bakker and Schaufeli (2006) suggest that a lack of significant job resources for prospective teachers, such as job control, access to information, supervisory support, innovative school climate and social climate to meet the job demand (e.g. disruptive pupil behaviour, work overload and poor physical work environment), may be associated with burnout, which may further decrease the students' sense of commitment.

5.3.5 Emotion

The students consistently stated that the heart and soul of teaching were very important to every teacher candidate. This was evident when they consistently mentioned the nature of teachers' work (i.e. dealing with individual human beings), which may negotiate with emotions. Hargreaves (1998) suggests that teachers should smile even when one is not happy. Following this, Flores and Day (2006) suggest that emotion is crucial to identity formation and emphasise the need for emotional labour – a term introduced by Hochschild (1983), as citied in Abstract (1989). Hochschild argues that common expectations exist concerning the appropriate emotional reactions of individuals in a certain profession. The regulation of emotion in response to these expectations is carried out in two ways: surface acting (changing one's outward emotional expressions without attempting to field the emotion displayed), and deep acting (changing one's outward emotional expressions while attempting to feel the emotion displayed).

One way that the students intended to show their emotions was through empathy. Empathy is the ability to mutually experience the thoughts and emotions of another person. Joireman et al. (2006) suggest that individuals who show empathy may exhibit altruism, civic virtue, conscientiousness and courtesy. A sense of empathy can be investigated through cognitive empathy, which relates to an individual's capacity to understand another person's perspective, rather than being exclusively self-oriented (Hogan, 1969). Thus, the students' abilities to be pleasant, charming, friendly, cheerful, sociable, sentimental, imaginative, discreet and tactful may describe their sense of empathy. In this sense, the students were trying to develop awareness towards their prospective students, their parents and local community.

5.3.6 Conclusion of the subsection

This subsection discusses the courses of teacher preparation students in negotiating their professional identity. Generally the findings suggest that students responded to their attributes, motives, beliefs, and attitudes towards the profession by creating imagined professional identities. It is also apparent from the data that the students were building their identities based on negotiations between the social context and their agency. Social context, such as a society and prospective students, informed the student teachers as to what they expected from their teachers. Subsequently, student teachers' own attributes, motives, beliefs, and attitudes towards the profession responded to the needs of the society and students through their agency. It was apparent that the students aspired to have enough agency and opportunities to practice their own orientation, thus have more commitment and positive attitudes towards the profession. In particular, this subsection suggests that the students differentiated different professional identities, based on individual professional orientation, personality, self-efficacy, autonomy and emotion. These five factors became a reference when people were asked about "who are you as a teacher".

5.4 Social identity intersectionality

As stated in Subsection 5.0, student teachers mentioned various social categories during the interview and relate their membership on those categories such as ethnic group and socio economic group with their desire to become a teacher (refer to Subsections 5.1.3, 5.1.4, 5.2.1). This implies that the student teachers have multiple social identity dimensions, and it intersects when the students try to self-categorise themselves as members of the teaching profession. Much research on social identity intersectionality involves the gender and ethnic (e.g. Crenshaw, 1991; Yuval-Davis, 2006); however, the thematic analysis of this study suggests the unexpected results where student teachers expressed a more positive attitude towards the teaching profession when both categories (profession and ethnic or socio economic group) become salient. It is apparent from the data that student teachers (i.e. S7 TP, S4 DPLI1, S5 DPLI1, S6 DPLI2) have more

desire and enthusiasm to educate the young generation when they perceive that the young generation is a member of the social category that they belong to.

5.5 Conclusion of chapter

As a conclusion, this chapter provides the answer to the fourth research question: How social factors affect the development, or decline of students' attitudes towards teaching during the commencement of the teacher education programme? This chapter describes the self-categorisation process among student teachers and it is apparent from the data that students respond to the values and norms of the profession in order to develop their identity as future teachers. Significant others exist as influencers, however, they become a member and not as somebody that students must conform to. Furthermore, as discussed in the two major subsections in this chapter, the students' narrations on their evaluation of the teaching profession explicitly explained that they are negotiating their professional identity as a teacher.

Based on their attributes, motives, beliefs, experiences, and a favourable or unfavourable attitude towards the profession, the students began to explore the meaning of their self and their aspirations of the profession. It was suggested from the thematic analysis that students placed their commitment as the basis for negotiating their sense of professional identity. There were five categories of anticipated professional identity found from the students' narratives. The characteristics were related to students' professional orientation, personality, self-efficacy, autonomy, and emotion.

Additionally, this chapter implicitly revealed that the attitude towards the teaching profession varied among the teacher preparation students at their entrance level to the teacher preparation programme. However, when they became a member of the teaching community by engaging in the teacher preparation programme, the majority of the students moved into a more positive attitude towards the profession. Thus, it can be suggested that being a member of the teacher preparation community provided more accessibility towards the nature of the teaching profession, thus bringing a positive impact on the students' evaluation of teaching.

Finally, the discussion in this chapter, as an extension from the previous chapter, also provided an answer for the last research question. The students, through their narratives, consistently suggested that they tried to engage in the socialisation process based on their professional expectations. The students' awareness of the education system, as promoting social reproduction, increased their determination to help the students by becoming agents of social mobility. Hence, the main socialisation agent for the majority of the students in this study was their own students, or their prospective students.

This chapter also concludes that student teachers not only acquire their professional identity through the socialization process, but the multiple social categories also have an influence on their identity development.

CHAPTER SIX CONCLUSION

6.0 Summarising the findings

This study has analysed students' professional identity development at higher education institutions, and the significance of group membership in determining the students' attitude and behaviour towards their intended profession. In particular, this study highlights the socialisation process in higher education which helps teacher education students define themselves as future teachers, and how the definition influences the decision to remain in the profession after graduation. The first half of this concluding chapter reflects the key findings of the study and outlines how it contributes to the knowledge about the identity development process among teacher education students during the commencement of the teacher preparation programme. In the second half of the chapter, I clarify the theoretical and practical implication of this study, including the limitation and make suggestions for future research that would be built on this study. This chapter concludes with a personal reflection on the value added obtained during the process of completing the study.

Retaining teacher education students in the teacher education programme until graduation, motivate them to become teachers, and uphold their commitment towards teaching have become crucial to teacher educators particularly when the entrance to the teacher education programme is restricted and controlled by the government. As higher education becomes a platform for students to cultivate the qualities of themselves, this study aims to identify the student teacher's self-definition on teaching. This aim resulted from consistent evidence from research on higher education; students' self-definition about specific social categories and social group has been shown to strongly influence attitude and behaviour relevant to their self-definition (Bennet, Rowly, Dunbart-Hall, Hitchcock & Blom, 2016; Hermann, Bager-Elsborg and McCune, 2016; Platow, Masica & Jones, 2016; Maxwell-Stuart & Huisman, 2018; Smyth, Mavor & Platow, 2018). Thus, it is crucial to identify how students define themselves as teachers,

particularly nearing to professional practice as this would explain their level of engagement with the teaching profession.

This study has given an account of the use of the social identity approach to acquire student's self-definition of teaching. This approach provides a useful lens to explain how teacher education students give their self-definition as a member of a group that they belong, together with their sense of belonging to the group. Students become attached to the group if they feel that their group is a source of pride and self-esteem. This includes the norms and values of teaching that might differentiate teaching from other professions. This approach yields good results in data as through the narratives, the students try to negotiate a positive social identity of teaching. Their commitments towards teaching are presented well in the data. This is an important finding in the understanding of an underrepresented group in the social structure. As written in Subsection 1.4, teaching is never regarded as a full-fledged profession, and students with a positive social identity try to negotiate as emerged from the data, and are seen as the motivation and commitment of the group members to upgrade their group status. This study also furthers the understanding of professional socialisation among higher education students in Malaysia, which assume that they have undergone the functionalist approach of socialisation, thus limiting the opportunity for students to interact with the environment in order to construct their own identity.

The findings of this study suggest that teacher education students generally perceive themselves as part of the teaching professional once they enrol in the teacher education programme. This perception arises based on the teacher recruitment practice in which all teacher education students are guaranteed a post as a teacher after graduation, except for the second group consisting of first year students who are excluded from this privilege. Furthermore, the social identity approach helps to explore the development of teacher education students' identity towards the teaching profession. This approach provides answers on how teacher education students discriminate themselves in a teaching group, and how they define themselves as a future member of the profession. This study has found that the majority of teacher education students are influenced by the desire to teach when they decided to apply for a teacher education programme although their genuine interest for teaching varies. The early categorisation process to become a member of the teaching profession comes from the students' perception that

their attributes fit with the characteristics needed in the teaching profession, influence from significant others such as parents, young people and former teachers, and solely because of vocational reasons. The findings also suggest that students' initial self-definition on teaching are influenced by their experiences as a school pupil, and not based on teachers' characteristics as outlined in formal documents by Ministry of Education such as Malaysia Teachers' Standard document.

It was shown from the data that at the beginning of the journey to become a teacher, more than half of the teacher education students interviewed expressed that their genuine reason to apply for a teacher education programme was not related to the interest of teaching, rather they were attracted to the privilege offered to graduate from this academic programme. These findings corroborate with previous results (Mohd Dahan and Ghazali, 2007; Harun, 2006; Mohd Ilias, 2008) on prospective teachers' disinterest and less commitment on teaching. The rest of the teacher education students either do not have the decision whether to teach or not, or do not plan to teach as their enrolment in this programme was not by their own choice.

Furthermore, the students' narrative confirmed that the self-categorisation process becomes more apparent when the teacher education students socialise in a teacher preparation programme. The students, through their beliefs and perception of the teaching profession, are aware that they are now part of the teaching group and try to negotiate their social identity based on their subjective experiences, before and during the teacher preparation programme. The perception that they will become a teacher after graduating and expectations from the society of the role and behaviour of teachers become significant factors that influence the students' identification in teaching.

What is surprising from the data is the relevance of school pupils as a socialisation agent. Teacher education students with teaching experience prior to their enrolment in the teacher education programme, and those who underwent the teaching practise describe a realistic influence of school pupils on their self-definition as a teacher. Their counterparts, which in the first year of the teacher education programme are not left to imagine the influence of their prospective school pupils in the construction of their self-definition towards teaching. A majority of the teacher education students expressed their concern on social reproduction emerging from the national educational system.

Their initiative to cease the practice that promotes social reproduction conveyed their commitment to become an agent of social mobility in school.

The result of the analysis showed that teacher education students interact with their environment to construct self-definition as a teacher, but at the same time have preserved the structural expectation imposed to the teaching profession. Thus, the socialisation process rejects the assumption that their socialisation process is solely determined by the functionalist approach. On the other hand, teacher education students bring their agency into their socialisation process to construct their identity. This is consistent with S. Bolster (1983) who claimed that individuals "must be considered both as a product and creator of which they live" (p. 303). Therefore, it seems that the findings also answer the question from the society whether the teacher is self-made or is born. The former explains the identity towards the teaching profession is subject to change with active participants from the teacher education students.

One of the more significant findings that emerged from this study is the five main characteristics of professional identity which teacher education students try to negotiate. Professional orientation, personality, self-efficacy, autonomy and emotion emerged from the students' sense of commitment derived from their narratives. Teacher education students give a different impression regarding these characteristics, as a result of their subjective experience before and during the teacher education programme. These characteristics suggest that the teacher education students' negotiation take into account their private lives, professional lives, as well as their relationship with the society. Thus, it has been recognised that the teacher education students' negotiation on the professional identity as a future teacher also intersects with other social identities, such as religion and ethnic group.

Returning to the research questions posed in Chapter One, it is now possible to state that 1) the teacher education students' decision to teach is influenced mostly by social factors such as significant others, former life experiences and vocational reasons; 2) these motives bring the students to imagine their intended behaviour of becoming a member of the teaching profession; 3) a majority of the teacher education students possess a more positive social identity towards teaching. The teacher education students' commitment towards teaching is driven mainly by a sense of responsibility

towards the school pupils. Additionally, other parties such as the society and a sense of responsibility towards the country also influence the students' self-definition on teaching; and 4) the teacher education students develop their professional identity mostly from their understanding of the function of the teaching profession, not because of the pressure to comply with other people. Furthermore, a positive social identity becomes the foundation for the teacher education students to express their sense of commitment towards the profession, and stimulate negotiation among the students on whom they want to be as a teacher. It is apparent from the findings that the teacher education students respond to the structural characteristics determined by the society, and nurtured by their self-agency.

The findings from this study suggest that generally there are differences between the socialisation process for teacher education students and their counterparts majoring in a different field at higher education institutions as reviewed in this study. The most significant aspect is the role and influence of prospective clients, which are the school pupils as an important source of motivation to remain in the teacher education programme. Teacher education students perceive that the school pupils influence their commitment towards the profession, whereas other students described that more attention has to be given towards the nature of the job as a way to determine a positive social identity in their chosen profession. This finding is not expected; however, this provides the indicators that the students are motivated to upgrade the social status of teaching. So far, the students believe that social status is determined by the function of teaching; however, as stated in Chapter One, teachers' knowledge and autonomy have to be given special attention in order to enable teaching as an established profession.

6.1 Theoretical implication

It is the contention for the researcher that some aspects of the present study be regarded as contributory to new knowledge. The current findings add to the growing body of literature on the professional identity formation, social identity approach and interpretive approach.

6.1.1 Professional identity formation and development

As discussed in Subsection 1.4, the theoretical framework underpinning the research on professional identity formation and development among higher education students mostly focuses on learning and teaching about professional work. Besides learning and teaching approaches, a number of studies applied different perspectives such as Lave and Wenger's work on community of practice (Jackson, 2016), reflective practice (Simmons & Dicks, 2018) and self-determination theory (Mylrea, Sen Gupta & Glass, 2017). These approaches comprehensively explain how higher education students get to know their anticipated profession, particularly the work characteristics, roles and attributes.

However, as some professions are tightly controlled in terms of the numbers admitted, there is a need to identify students' professional identity development beyond the work preparations. Importantly, it is crucial to inspire students to remain in their intended profession and have a positive orientation with their future works and responsibilities. It is predicted that individuals will remain in the profession if they believe that the profession itself is an important source of pride and self-esteem. The social identity approach underlines the assumption that individuals are just not proud of themselves as members of a group, rather they try to act accordingly to maintain the group's positive social identity, and they are also inclined to seek positive values and norms that will differentiate their group from others. Thus, applying the social identity approach on a professional identity study gives input on how the profession itself become important to students. This approach explains what constitutes the profession, the motivation of the members of the profession, and their commitment to achieve, maintain, or enhance the positive social identity of the profession.

6.1.2 Social Identity Approach

In previous years, Social Identity Approach is used in different context to investigate group processes and intergroup relations such as in socio-political conflict (Colvin, Witt and Lacey, 2015), consumer brand identification (Rather, 2018), cultural identity (Chung, Jetten, Cruwys and Haslam, 2017), leadership development (Haslam, et. al., 2017) and firm's social performance (Nason, Bacq and Gras, 2018). It has been

suggested by scholars in group processes and intergroup relations to further investigate group-based emotions, particularly on how people represent emotions of their own groups (Abrams and Hogg, 2017; Mackie and Smith, 2018). Thus, current study measured group-based emotions on one occupational group. Firstly, Social Identity Approach provides a practical framework to figure out the characteristics of one occupational group that becomes a sense of pride and self-esteem of its members. This study signalled that the intended occupational group becomes the students' referent group, and may explain why they are attached to the group even though they have not yet become a member of the profession. The findings give a better understanding on the result of the minimal group studies by Tajfel (1974) that proposed perceived membership in one group regardless of the status, which will result in a distinct behaviour among the group members.

Secondly, it has been revealed the significance of the Social Identity Approach in determining the motivation factors to remain in what is called the member of the underrepresented groups in a social structure. As stated in Subsection 1.4, teaching is never regarded as a full-fledged profession, and this status might affect the students' pride and self-esteem of becoming a member of such a group. However, the findings of this study signalled that students are able to define their group's norms and values which clearly differentiate teaching from other professions. In other words, the findings clearly showed the strategies underlined in the Social Identity Approach have been fully utilised by students to show their self-definition of teaching. The students that appreciate the group norms and values used the social creativity strategy to upgrade the position of the group in the social structure by narrating the ultimate contribution that they can offer to the society by becoming a teacher. The students who were undecided seemed to search for new professions that suited their professional aspirations, thus indicating that the social mobility strategy works for individuals who view other professions which have more distinctiveness norms and values.

Additionally, this approach explains how the students identify their own group aspirations, mostly from their experiences as school pupils. The findings of this study also revealed how the assumption underlined in the social identity approach helps to identify factors that increase the group members' self-esteem and pride of an underrepresented group. The findings showed that the values and norms are not

necessarily gathered from formal rules, procedures and guidelines, but it may come from other sources such as the environment and significant others. Thus, future research aiming to identify the strength of underrepresented groups may apply the social identity approach as this approach helps to give the answer for what makes people attached to one particular group.

Thirdly, it has been revealed that there are different social identities interwoven in the formation of the teacher education students' identity towards the teaching profession. The findings of this study are able to present the multiple realities when the students narrated the underlying norms and values of the teaching profession which originated from cultural and religious beliefs. The study has gone some way towards enhancing the understanding that social identity related to the professional, cultural and religious groups intersect to strengthen the construction of professional identity among future teachers in a Malaysian context. This may add to the growing body of literature on social identity intersectionality as most of the studies thus far research the experiences of marginalised individuals of groups such as race, social class and gender (Carrim & Nkomo, 2016; Rogers, Scott and Way, 2015; Kim & Hong, 2019).

6.1.3 Interpretive approach

Additionally, this study devised a methodology of using the interpretive approach in researching the professional identity development in higher education. The interpretive approach is steadily gaining momentum as it may explain in depth the students' socialisation process, not limited to the predetermined factors offered by a positivist approach. This predominant quantitative approach has a tendency to limit the understanding on how teacher education students acquire their identity, and try to maintain or achieve a positive social identity. It has been revealed from the findings that the students narrative on teaching is based on their experiences. As the experiences may vary from one person to another, the interpretive approach helps to explain the undiscovered reality of the students' professional identity formation and development.

6.2 Practical implication

This study has gone somewhere towards enhancing the understanding of professional identity development among teacher education students in higher education. Generally, the findings suggest that it is possible for higher education administrator and academics to support and enhance students' professional identity development. The study also provides a glimpse on how perceived membership in one occupational group explains the student's intended behaviour. In particular, the findings of this study have a number of important implications for future practises; firstly, it has been shown that students value teaching from different perspectives. To recap, the findings clearly revealed that there were three distinctive groups of students commencing in the teacher education programme; students who want to teach, students who are undecided, and students who do not want to teach. Regardless of the group, students teachers need specific support and professional development that goes beyond learning and teaching the professional roles. It becomes a crucial task for higher education administrators and teacher educators to understand the variability of students and treat them accordingly.

Obviously, there will be a need to address different approaches to each group of students in order to achieve, enhance or maintain their positive social identity. As the teacher education programme aims to retain students in the profession, and produce teachers with a positive orientation towards teaching, the teaching and learning on professional work and attributes are not sufficient to develop their positive social identity. Teacher educators have to focus on the values and norms of teaching that will foster the students' commitment and provide a conducive environment for them to assimilate the teaching values and norms. In addition, teacher education programme needs to create cultures where students feel welcome to the group consists of future teachers. Positive cultures have strong effects on students' attraction and retention, and conducive to develop students' commitment (Kontoghiorghes, 2016).

Secondly, the findings suggested that the students, regardless of the group have a limited definition on what constitutes teaching. To clarify, there are more positive values and norms required for teachers and subsequently makes teaching different from other professions. This is consistent with the literature reviewed on the professionalism that highlighted the needs to enhance the teacher's knowledge related to their

profession. This includes the pedagogy of teaching, managing school pupil's emotions, classroom management and assessments (Ministry of Education Malaysia, 2012). Thus, it becomes a responsibility for teacher educators to enhance the student's knowledge on professional dispositions and roles as documented by the Ministry of Education. Additionally, the higher education administrators should help in cultivating students' professional identity by encouraging them to engage in multiple communities within higher education in a way to make students make sense of their future profession. This includes professional associations, students' societies and career services (Jackson, 2016).

Thirdly, this study has suggested that a trait-oriented approach used by prior studies as a way to predict a positive social identity among students in higher education institutions is no longer successful to mould students in a particular way; however, this must be accompanied by an approach that highlights the lived experiences of students. In other words, the professional socialisation not only depends on the attributes of the students, but their interaction with the environment and perception on group distinctiveness will make a difference on how they evaluate their future profession. This is consistent with the fact that students enter higher education institutions with different motives and abilities, thus this makes a difference in self-definition towards the profession and how they negotiated their future professional identity. Repeatedly, these findings provide crucial input for teacher preparation course design and delivery. Specifically, the input from the findings can be used to develop relevant non-academic programmes for teacher education students as this effort will help them to have more interactions with significant others and the environment (Jackson, 2016). The structure of the teacher education programme will not be revisited as recent research reveals that the current university-based course is aligned with real classroom practices (Goh, Canrinus & Wong, 2019).

Course delivery will focus more on deep and active learning, as research suggests that deep learning has a strong connection with the students' achievement and understanding about the subject. The literature in learning indicates a clear relationship between the characteristics of individual leaner social self-concept with the learning approach and positive learning outcomes (Biggs, Kember & Leung, 2001; Bunce et al., 2017). In particular, a deep learning approach associated with students' intentions to

understand and to engage appropriately in meaningful learning, 'focusing on the main theme and principles, and using strategies that are appropriate for creating such meaning' (Asikainen and Giibels, 2017, p.209). Additionally, students should be encouraged to attend lectures regularly, read related materials, boost an effort to study, spend a lot of time studying, enjoy learning and being at the university to learn about the profession (Bunce et al., 2017). These efforts are predicted to attract the undecided group to re-think of their existence in the teacher education programme and nurture their interest and sense of belonging towards teaching. Nonetheless, this will help to reattract the third group who decide to leave to make a comeback.

Moreover, the findings also support the beliefs in students' involvement in the interpretive approach of socialisation. The beliefs that teachers have a close relationship with the young generation imply that teachers are required to play a role to their utmost capabilities. There is evidence from this study which suggest that student's motivation to learn is impacted from their teacher's attitude in the classroom [for example, refer Subsection 5.2.1: S5 FY1 and S4 FY1]. Thus, it is crucial for teacher educators to become a mentor to teacher education students and regularly communicate the importance of teaching profession in society in order to motivate students to engage with the profession. Mentoring is widely used in higher education particularly to nurture students' identity and enhance academic performance (Hagler, 2018; Ligadu, 2012). Referring to this study, mentoring for teacher education students will help to minimise the negative consequences when the students' views and beliefs on teaching are confronted or even challenged during their commencement in the teacher education programme, resulting in students who either have to find their own way to assimilate the new construct with their existing beliefs and cultural models, or struggle to sustain their beliefs they brought along.

To summarise, the present study confirms that students' identity is subject to change, depending on the socialisation process and the agents perceived as important to the teacher education students. Thus, a positive yet challenging environment is needed in order to promote a positive social identity, which in this context refers to the teachers' professional identity. With the understanding of the group values and norms that will motivate the teacher education students to acquire a positive social identity, teacher educators and the teacher training institution have to fully utilise their expertise and

facilities, as well as re-design the teaching and learning approach in higher education towards a deeper learning approach. The students need to be encouraged to get involved in different communities within higher education and become an active learner, as the professional identity should have to be learnt both in and outside the classroom.

6.3 Limitation of the study

The study has a number of limitations. To begin with, the research paradigm and approach. Being a qualitative research study, the small sample size of teacher education students (80 participants) from one of the largest teacher education institution may imply that the results cannot be readily generalised, although they are likely to have wider relevance and applicability especially in other teacher preparation institutions and other Muslim majority countries.

Furthermore, there is the intervention of admission tools described in this study. One of the tools, MEdSI, is used to raise the admission requirements and increase the value of the teaching profession for many undecided higher education applicants. This is the limitation for this study to go further on the test; however, it is worth to note the two consequences of the MEdSI test in this study. Firstly, when referred to the genuine reason to embark on the teacher education programme as expressed by the participants, one could relate to the reliability and validity of the test. The participants' genuine interests that do not favour the teaching profession at the time they decide to apply for the teacher preparation programme have an effect on the effectiveness of the test for admission purposes. Looking back at the participants' narratives, it is apparent that not all participants are aware that they have the positive qualities measured by the test. Although this issue is not the main concern of the study, the effectiveness of the test should be attended to; therefore, it becomes a limitation of this study.

Secondly, students might possess the minimum requirement to become a teacher in terms of personality, career interest, integrity value and emotional intelligence, yet they do not realise these qualities in themselves. It is clearly stated by the undecided participants that they do not have a clear direction of what they want to be, thus influences from significant others are very important. From the narrative, the

participants with extrinsic motives and external influences have moved towards becoming more passionate towards the teaching profession. Thus, it can be concluded that the participants who applied for the teacher education programme, consciously and unconsciously have qualities to be a teacher, and their behaviour might change according to the norms and values of the profession. Furthermore, the students who tend to agree with their parents denote that the structuralism factor is dominant in the Malaysian society. Participants also can be said as having less ability in career decision making because all the decisions are influenced by external factors, and not embedded in themselves.

6.4 Future studies

This study has identified the importance of societal factors, particularly young people at school as the main socialising agent who influences the teacher education student's motivation on teaching, and the development of their sense of professional identity, thus rejecting the assumption that professional socialisation in Malaysia's higher education is determined solely by profession. However, considerably more work will need to be done to determine the relationship between the admission tools and identity, as well as their professional identity negotiation during their commencement in the teacher education programme.

A number of possible future studies using the same paradigm and framework are apparent. It would be interesting to access the breadth and depth of insight into, and appreciation of the professional socialisation process of students from different academic specialisations in higher education institutions in Malaysia. Although the teacher education programme is dominated by Malay students, other academic programmes offered in higher education institutions have multiracial students, including the Chinese, Indians and other ethnics in Malaysia. Future research might explore the development of professional identity among the students undergoing different academic specialisation in order to identify whether they respond to the interaction with the environment, or rigidly place them in a position that the teacher or academic staff are the most influential socialising agents.

Further explorations are needed to identify whether there are intersections of different social identities such as gender and ethnic, having an impact on the socialisation process. This result will provide a perspective on the professional socialisation approach among higher education students in Malaysia, and their attitude towards their chosen profession. It is noteworthy that the result will enhance the landscape of the development of professional identity among higher education students particularly in Malaysia. In addition, the exploration on new academic programmes offered in Malaysia's higher education such as entrepreneurship, and more specialised academic programmes such as human resource management and quality management will contribute to the new knowledge of the higher education evaluation on the new programme, and their construction of professional identity related to their future careers in the related field.

As a concluding remark, I appreciate my experience of using the interpretive paradigm of research in conducting this study. By adopting this paradigm, I am able to obtain rich and in-depth data, thus providing the explanation of the subject being studied. However, future studies could fruitfully explore students' professional identity development further by using both quantitative and qualitative approach to allow both exploration and statistical analysis in the same study, and gather more comprehensive data.

REFERENCES

- 10th Malaysia Plan (2010). Retrieved from https://www.pmo.gov.my/dokumenattached/RMK/RMK10_Eds.pdf
- Abbott, A. (1998). Professionalism and the future of librarianship. *Library Trends*, 46(3), 430-444.
- Abrams, D., & Hogg, M. A. (1990). Social identification, self-categorization and social influence. *European review of social psychology*, *I*(1), 195-228.
- Abrams, D., & Hogg, M. A. (2017). Twenty years of group processes and intergroup relations research: A review of past progress and future prospects. *Group Processes & Intergroup Relations*, 20(5), 561-569.
- Abstract, N. J. (1989). Emotional labour: skill and work in the social regulation of feelings. *The sociological review*, *37*(1), 15-42.
- Admission and Registration Unit. Universiti Pendidikan Sultan Idris. Unpublished documents.
- Ahmad, I. (2003). Profession perguruan pilihan terakhir? (Teaching is the least preferred choice?). Retrieved from www.utm.my
- Ahmad, Z., Anantharaman, R.N. & Ismail, H. (2012). Students' motivation, perceived environment and professional commitment: An application of Astin's College Impact Model. *Accounting education: An International Journal*, 21(2), 187 208.
- Ajzen, I. (2001). Nature and operation of attitudes .*Annual review of psychology*, 52(1), 27-58.

- Ajzen, I. & Fishbein, M. (1980). Understanding attitudes and predicting social behaviour.
- Aldemir, J. & Sezer, O. (2009). Early childhood education pre-service teachers' images of teacher and belief about teaching. *Inonu University Journal of The Faculty of Education*, 10(3), 105-122.
- Allen, N., & Meyer, J. (1990). The measurement and antecedents of affective, continuance and normative commitment to the organization. *Journal of Occupational Psychology*, 63, 1-18.
- Allport, G. W. (1935). Attitudes. In *Handbook of social psychology*, (ed): C. Murchison, 789 844. MA: Clark University Press.
- Alsup, J. (2003). English education students and professional identity development: using narrative and metaphor to challenge pre-existing ideologies. *Pedagogy*, 3(2), 277-280.
- Andersen, L. B. (2009). What determines the behaviour and performance of health professionals? Public service motivation, professional norms and/or economic incentives. *International Review of Administrative Sciences*, 75(1), 79-97.
- Antonek, J. L., McCormick, D. E., & Donato, R. (2007). The student teacher portfolio as autobiography: Developing a professional identity. *The Modern Language Journal*, 81(1), 15-27.
- ArjaVirta (2002). Becoming a history teacher: observations on the beliefs and growth of student teachers, *Teaching and Teacher Education*, 18, 687 698.
- Arnett, J. J. (2000). Emerging adulthood: A theory of development from the late teens through the twenties. *American psychologist*, 55(5), 469.
- Atkinson, P., & Delamont, S. (1985). Socialisation into teaching: the research which lost its way. *British Journal of Sociology of Education*, 6(3), 307-322.

- Attride-Stirling, J. (2001). Thematic networks: an analytical tool for qualitative research. *Qualitative Research*. 1(3). SAGE Publication. 385-405.
- Ashforth, B., & Kreiner, G. (1999). How can you do it? Dirty work and the challenge of constructing a positive identity. *Academy of Management Review*, 24, 413-34.
- Ashforth, B., & Mael, F. (1989). Social identity theory and the organization. *Academy of Management Review*, 14(1), 20-39.
- Asikainen, H., & Gijbels, D. (2017). Do students develop towards more deep approaches to learning during studies? A systematic review on the development of students' deep and surface approaches to learning in higher education. *Educational Psychology Review*, 29(2), 205-234.
- Atadero, R. A., Paguyo, C. H., Rambo-Hernandez, K. E., & Henderson, H. L. (2018). Building inclusive engineering identities: implications for changing engineering culture. *European Journal of Engineering Education*, 43(3), 378-398.
- Azman, N. (2013). Choosing teaching as a career: Perspectives of male and female Malaysian student teachers in training. *European Journal of Teacher Education*, 36(1), 113-130.
- Badcock, P. B. T., Pattison, P. E. & Harris, Kerri-Lee (2010). Developing generic skills through university study: a study of arts, science and engineering in Australia. *Higher Education*, 60, 441 458.
- Baglin, M. R., & Rugg, S. (2010). Student nurses' experiences of community-based practice placement learning: A qualitative exploration. *Nurse Education in Practice*, 10(3), 144-152.
- Baik, C. & Greig, J. (2009). Improving the academic outcomes of undergraduate ESL students: the case for discipline-based academic skills programs. *Higher Education Research and Development*, 28(4), 401 416

- Bakar, A. R., Konting, M. M., Jamian, R., & Lyndon, N. (2008). Teaching efficacy of Universiti Putra Malaysia science student teachers. *College Student Journal*, 42(2), 493.
- Bandura, A. (1969). Social foundations of thought and action. Englewood Cliffs: NJ.
- Barber, B. (1963). Some problems in the sociology of the professions. *Daedalus*, 669-688.
- Baretti, M. (2004). What do we know about the professional socialisation of our students? *Journal of Social Work Education*, 40(2), 1-8.
- Barbour, R. (2014). *Introducing qualitative research: A student's guide*. London: Sage Publications.
- Baumeister, R. F. (1991). Meanings of life. Guilford Press.
- Bazeley, P. (2013). *Qualitative data analysis by NVivo*.2nd edition. London: Sage Publications.
- Beauchamp, C., & Thomas, L. (2009). Understanding teacher identity: An overview of issues in the literature and implications for teacher education. *Cambridge Journal of Education*, 39(2), 175-189.
- Beck, J. (2009). Appropriating professionalism: restructuring the official knowledge base of England's 'modernised' teaching profession. *British Journal of Sociology of Education*, 30(1), 3-14.
- Becker, H. S. (1964). Personal change in adult life. *Sociometry*, 40-53.
- Bennett, D., Rowley, J., Dunbar-Hall, P., Hitchcock, M., & Blom, D. (2016). Electronic portfolios and learner identity: An ePortfolio case study in music and writing. *Journal of Further and Higher Education*, 40(1), 107-124

- Berger, M. C. (1988). Predicted future earnings and choice of college major. *Industrial* and Labor Relations Review, 41(3),418 429
- Berings, D., Fruyt, F. D., & Bouwen, R. (2004). Work values and personality traits as predictors of enterprising and social vocational interests. *Personality and Individual Differences*, 36, 349 364
- Biggs, J., Kember, D., & Leung, D. Y. (2001). The revised two-factor study process questionnaire: R- SPQ- 2F. *British Journal of Educational Psychology*, 71(1), 133-149.
- Boeije, H. R. (2009). Analysis in qualitative research. Sage Publications.
- Bogler, R., & Somech, A. (2004). Influence of teacher empowerment on teachers' organizational commitment, professional commitment and organizational citizenship behavior in schools. *Teaching and teacher education*, 20(3), 277-28
- Borgen, W. A. & Young, R. A. (1982). Career perceptions of children and adolescents. *Journal of Vocational Behaviour*, 21, 37 49.
- Boyatzis, R. E. (1998). Transforming qualitative information: Thematic analysis and code development. Sage.
- Bragg, A. K. (1976). The socialization process in higher education. *Higher Education Research Report*,
- Bramming, P. (2007). An argument for strong learning in higher education. *Quality in Higher Education*, 13(1), 45-56.
- Brookhart, S. M., & Freeman, D. J. (1992). Characteristics of entering teacher candidates. *Review of educational research*, 62(1), 37-60.

- Braun, V. & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative* research in psychology, 3, 77 101.
- Brim, O. G. & Wheeler, S. (1966). Socialisation after childhood. Wiley: New York.
- Brooks, J.G., & Brooks, M.G. (1999). *In Search of Understanding: The Case for Constructivist Classrooms*. Alexandria, VA: Association for Supervision and Curriculum Development.
- Bruinsma, M., & Jansen, E. P. (2010). Is the motivation to become a teacher related to pre-service teachers' intentions to remain in the profession?. *European Journal of Teacher Education*, 33(2), 185-200.
- Bryman, A. (2012). Social research methods.4th edition. Oxford: Oxford University Press.
- Buchanan, M. T., & Stern, J. (2012). Pre-service teachers' perceptions of the benefits of peer review. *Journal of Education for Teaching*, 38(1), 37-49. Chicago.
- Bunce, L., Baird, A., & Jones, S. E. (2017). The student-as-consumer approach in higher education and its effects on academic performance. *Studies in Higher Education*, 42(11), 1958-1978.
- Burford, B. (2012). Group processes in medical education: learning from social identity theory. *Medical education*, 46(2), 143-152.
- Burrell, G., & Morgan, G. (1979). *Sociological paradigms and organisational analysis* (Vol. 248). London: Heinemann.
- Burr, V. (2003). Social constructionism.2nd edition. London, NY: Routledge.
- Byrne, M., & Flood, B. (2005). A study of accounting students' motives, expectations and preparedness for higher education. *Journal of Further and Higher Education*, 29(2), 111-124.

- Carrim, N. M. H., & Nkomo, S. M. (2016). Wedding intersectionality theory and identity work in organizations: South African Indian women negotiating managerial identity. *Gender, Work & Organization*, 23(3), 261-277.
- Chambers, G. N., Hobson, A. J. & Tracey, L. (2010). "Teaching could be a fantastic job but ...': Three stories of student teacher withdrawal from initial teacher preparation programmes in England. *Teacher and Teaching*, 16(1), 111 129.
- Chang, M. X. L., Jetten, J., Cruwys, T., & Haslam, C. (2017). Cultural identity and the expression of depression: A social identity perspective. *Journal of Community & Applied Social Psychology*, 27(1), 16-34.
- Charmaz, K. (2000). Grounded theory: Objectivist and constructivist methods. In N. Denzin & Y. Lincoln (Eds.), *Handbook of qualitative research* (2nd ed., pp. 509-535). Thousand Oaks, CA: Sage.
- Chen, H. F., & Chiu, Y. H. (2009). The influence of psychological contracts on the adjustment and organisational commitment among expatriates: An empirical study in Taiwan. *International Journal of Manpower*, 30(8), 797-814.
- Chiaburu, D. S., & Harrison, D. A. (2008). Do co-workers make the place? Conceptual synthesis and meta-analysis of lateral social influences in organizations. *Journal of Applied Psychology*, *93*(5), 1082-1103.
- Cheng, Y. C. (1994). Locus of control as an indicator of Hong Kong teachers' job attitudes and perceptions of organizational characteristics. *The Journal of Educational Research*, 87(3), 180-188.
- Cherry, N. L. (2005). Preparing for practice in the age of complexity. *Higher Education Research & Development*, 24(4), 309-320.
- Chinoy, E. (1955). Social mobility trends in the United States. *American Sociological Review*, 20(2), 180-186.

- Clandinin, D. J., Long, J., Schaefer, L., Downey, C. A., Steeves, P., Pinnegar, E., ... & Wnuk, S. (2015). Early career teacher attrition: Intentions of teachers beginning. *Teaching Education*, 26(1), 1-16.
- Clarke, M. & Otaky, D. (2006). Reflection 'on' and 'in' teacher education in the United Arab Emirates *International Journal of Educational Development*, 26, 111-122.
- Clausen, J. A. (1968). Perspectives on childhood socialization. *Socialization and society*, 131-81.
- Clouder, L. (2003). Becoming professional: exploring the complexities of professional socialization in health and social care. *Learning in Health and Social Care*, 2(4), 213-222.
- Codd, J. (2005). Teachers as 'managed professionals' in the global education industry: The New Zealand experience. *Educational review*, *57*(2), 193-206.
- Cohen, H. A. (1981). *The nurse's quest for a professional identity*. Addison-Wesley Publishing Company, Medical/Nursing Division.
- Colvin, R. M., Witt, G. B., & Lacey, J. (2015). The social identity approach to understanding socio-political conflict in environmental and natural resources management. *Global Environmental Change*, *34*, 237-246.
- Corbin, J., & Strauss, A. (Eds.). (2008). Basics of qualitative research: Techniques and procedures for developing grounded theory. CA, London: Sage.
- Cottrell, L. S. (1969). Interpersonal interaction and the development of the self. *Handook of Socialisation Theory and Research*, ed. D. Goslin. Chicago: Rand McNally pp. 543-70

- Crenshaw, K. (1991). Mapping the margins: Intersectionality, identity politics, and violence against women of color. *Stanford law review*, 1241-1299.
- Cresswell, J. W. (2013). *Qualitative inquiry and research design: choosing among five approaches*. 3rd edition. CA: Sage Publication Inc.
- Crosswell, L. J., & Elliott, R. G. (2004). Committed teachers, passionate teachers: The dimension of passion associated with teacher commitment and engagement. *QUT eprints*.
- Cresswell, J. W. & Miller, D. L. (2000). Determining validity in qualitative inquiry. *Theory into Practice*. 39(3), 124 130.
- Crotty, M. (1998). The foundations of Social Research: Meaning and perspective in social research. Australia: Allen & Unwin.
- Czarniawska-Joerges, B. (1998). A narrative approach to organization studies (Vol. 43). CA: Sage.
- Dahlgren, M. A., Hult, H., Dahlgren, L. O., af Segerstad, H. H., & Johansson, K. (2006). From senior student to novice worker: Learning trajectories in political science, psychology and mechanical engineering. *Studies in Higher Education*, 31(5), 569-586.
- Daniel, J. (2012). Sampling essentials: practical guidelines for making sampling choices. CA: Sage Publications.
- Dashtipour, P. (2012). Social identity in question: construction, subjectivity, and critique. Routledge.
- Day, C. (2002). School reform and transitions in teacher professionalism and identity. *International journal of educational research*, *37*(8), 677-692

- Decker, L. E., & Rimm-Kaufman, S. E. (2008). Personality characteristics and teacher beliefs among pre-service teachers. *Teacher Education Quarterly*, 45-64.
- De Neve, D., & Devos, G. (2017). Psychological states and working conditions buffer beginning teachers' intention to leave the job. *European Journal of Teacher Education*, 40(1), 6-27.
- Denzin, N. K. (1989). *Interpretive interactionism* (Vol. 16). London: Sage Publications Inc..
- Denzin, N. K., & Lincoln, Y. S. (Eds.).(2011). *The SAGE handbook of qualitative research*. London: Sage Publications.
- Deutsch, M., & Gerard, H. B. (1955). A study of normative and informational social influences upon individual judgment. *The journal of abnormal and social psychology*, 51(3), 629.
- Eagly, A. H., & Chaiken, S. (1993). *The psychology of attitudes*. Harcourt Brace Jovanovich College Publishers.
- Elliott, P. R. C. (1972). *The sociology of the professions*. New York: The Macmillan Press Ltd
- Eick, C. J., & Reed, C. J. (2002). What makes an inquiry- oriented science teacher? The influence of learning histories on student teacher role identity and practice. *Science Education*, 86(3), 401-416.
- Elias, R. Z. (2008). Auditing students' professional commitment and anticipatory socialisation and their relationship to whistle blowing. *Managerial Auditing Journal*, 23(3), 283 294.
- Ellemers, N., De Gilder, D., & Haslam, S. (2004). Motivating individuals and groups at work: A social identity perspective on leadership and group performance.

 Academy of Management Review, 29(3), 459-478.

- Eraut, M. (2000) .Non- formal learning and tacit knowledge in professional work. British journal of educational psychology, 70(1), 113-136.
- Eraut, M. (1994). Developing professional knowledge and competence. Psychology Press.
- Estola, E. (2003). Hope as Work--Student Teachers Constructing Their Narrative Identities. *Scandinavian Journal of Educational Research*, 47(2), 181-203.
- Etizioni, A. (1969). The Semi-Professions and their Organizations: Teachers, nurses and social workers. New York: The Free.
- Evans, L. (2010). Professionals or technicians? Teacher preparation programs and occupational understandings. *Teachers and Teaching: theory and practice*, 16(2), 183-205.
- Fazio, R. H. (1990). Multiple processes by which attitudes guide behavior: The MODE model as an integrative framework. *Advances in Experimental Social Psychology*, 23, 75-109.
- Freidson, E. (1970). *Profession of medicine*. New York: Dodd. Mead, and Company
- Fokkens-Bruinsma, M., & Jansen, E. P. (2010). Is the motivation to become a teacher related to pre-service teachers' intentions to remain in the profession? *European Journal of Teacher Education*, 33(2), 185-200.
- Fokkens-Bruinsma, M., & Canrinus, E. T. (2012). Adaptive and maladaptive motives for becoming a teacher. *Journal of Education for Teaching*, *38*(1), 3-19.
- Fokkens-Bruinsma, M., & Canrinus, E. T. (2014). Motivation for becoming a teacher and engagement with the profession: Evidence from different contexts. *International Journal of Educational Research*, 65, 65-74.

- Flores, M. A. & Day, C. (2006). Context that shape and reshape new teachers' identities: A multi-perspective study. *Teaching and Teacher Education*, 22, 219 232.
- Forte, J. A. (2007). Human behavior and the social environment. *Belmont: Thompson Higher Education*.
- Gao, X. & Trent (2009). Understanding mainland Chinese students' motivation for choosing teacher education in Hong Kong. Journal of Education for Teaching, 35(2), 145-159.
- Gawronski, B., & Bodenhausen, G. V. (2006). Associative and propositional processes in evaluation: an integrative review of implicit and explicit attitude change. *Psychological bulletin*, *132*(5), 692.
- Geer, B. (1966). Occupational commitment and the teaching profession. *The school review*, 74(1), 31-47.
- Gibney, A., Moore, N., Murphy, F., & O'Sullivan, S. (2011). The first semester of university life; 'will I be able to manage it at all?'. *Higher Education*, 62(3), 351-366.
- Giddens, A. (1991). Modernity and self-identity. Cambridge: Polity Press.
- Ginzberg, E., Ginsburg, S. W., Axelrad, S., & Herma, J. L. (1951). Occupational choice. *New York*.
- Glaser-Segura, D. A., Mudge, S., Brătianu, C., & Dumitru, I. (2010). Development of professional identity in Romanian business students. *Education+ Training*, 52(3), 198-213.
- Goh, K. C., & Atputhasamy, L. (2001, December). Factors motivating the young adults to choose teaching as a career in Singapore. In *International Educational*

- Research Conference, University of Notre Dame, Fremantle, Western Australia (pp. 2-6).
- Goh, P. S. C., Canrinus, E. T., & Wong, K. T. (2019). Preservice teachers' perspectives about coherence in their teacher education program. *Educational Studies*, 1-17.
- Gonzalez, L., Brown, M. S., & Slate, J. R. (2008). Teachers who left the teaching profession: A qualitative understanding. *The qualitative report*, *13*(1), 1-11.
- Gorman, E. H., & Sandefur, R. L. (2011). "Golden age," quiescence, and revival: how the sociology of professions became the study of knowledge-based work. *Work and Occupations*, 38(3), 275-302.
- Grolnick, W. S., & Ryan, R. M. (1989). Parent styles associated with children's self-regulation and competence in school. *Journal of educational psychology*, 81(2), 143.
- Gundlach, M., Zivnuska, S., & Stoner, J. (2006). Understanding the relationship between individualism–collectivism and team performance through an integration of social identity theory and the social relations model. *Human relations*, 59(12), 1603-1632.
- Hafferty, F. W. (2008). Professionalism and the socialization of medical students. *Teaching medical professionalism*, 53-70.
- Hagler, M. (2018). Processes of natural mentoring that promote underrepresented students' educational attainment: A theoretical model. *American journal of community psychology*, 62(1-2), 150-162.
- Hakanen, J. J., Bakker, A. B., &Schaufeli, W. B. (2006).Burnout and work engagement among teachers. *Journal of school psychology*, 43(6), 495-513.
- Hall, R. H. (1968). Professionalization and bureaucratization. *American sociological review*, 92-104.

- Hall, D. T., Schneider, B., & Nygren, H. T. (1970). Personal factors in organizational identification. *Administrative science quarterly*, 176-190.
- Hallier, J., & Summers, J. (2011). Dilemmas and outcomes of professional identity construction among students of human resource management. *Human Resource Management Journal*, 21(2), 204-219.
- Hamid, A., Zolkifli, M., Rajuddin, M. R., Zakaria, M., &Zahiri, M. A. (2007). Harapan dan keresahan bakal guru pelatih dalam praktikum mengajar.
- Hanushek, E. A. (2013). Economic growth in developing countries: The role of human capital. *Economics of Education Review*, *37*, 204-212.
- Hansen, J. T. (2010). Consequences of the postmodernist vision: Diversity as the guiding value for the counselling profession, *Journal of Counselling and Development*, 88, 101 107.
- Hargreaves, A. (1998). The emotional practice of teaching. *Teaching and teacher education*, 14(8), 835-854.
- Harun, H. (2006). Minat, motivasi dan kemahiran mengajar guru pelatih. *Jurnal Pendidikan*, 21, 83 96.
- Hashim, H., Damio, S. M., &Hussin, N. (2013). Malaysian Educators SelectionInventory (MEdSI) Certified Interviewers' Perspectives on its Implementation:A Preliminary Study. *Procedia-Social and Behavioral Sciences*, 90, 171-177.
- Haslam, S. A. (2004). *Psychology in organizations*. Sage.
- Haslam, S., Powell, C., & Turner, J. (2000). Social identity, self-categorization and work motivation: Rethinking the contribution of the group to positive and sustainable organizational outcomes. *Applied Psychology: An International Review*, 49, 319-339.

- Hassan Farooqi, A. (2006). Islamic social capital and networking. *Humanomics*, 22(2), 113-125.
- Haslam, S. A., Steffens, N. K., Peters, K., Boyce, R. A., Mallett, C. J., & Fransen, K. (2017). A social identity approach to leadership development. *Journal of Personnel Psychology*.
- Heggen, K. (2008). Social workers, teachers and nurses—from college to professional work. *Journal of Education and Work*, 21(3), 217-231.
- Herr, K., & Anderson, G. (2003). Violent youth or violent schools? A critical incident analysis of symbolic violence. *Int. J. Leadership in Education*, 6(4), 415-433.
- Herrmann, K. J., Bager-Elsborg, A., & McCune, V. (2017). Investigating the relationships between approaches to learning, learner identities and academic achievement in higher education. *Higher Education*, 74(3), 385-400.
- Hofstede, G., & Bond, M. H. (1984). Hofstede's Culture Dimensions An Independent Validation Using Rokeach's Value Survey. *Journal of cross-cultural psychology*, 15(4), 417-433.
- Hofstede, G., & Hofstede, G. J. (2005). *Cultures and organizations, software of the mind, intercultural cooperation and its importance for survival*. Revised and expanded 2nd edition, London: McGraw-Hill.
- Hogan, R. (1969). Development of an empathy scale. *Journal of consulting and clinical psychology*, *33*(3), 307.
- Hogg, M. A., & Abrams, D. (1988). Social identifications.
- Holland, J. L. (1985). *Making vocational choices: A theory of vocational personalities and work environments*. Psychological Assessment Resources.

- Hong, J. Y. (2010). Pre-service and beginning teachers' professional identity and its relation to dropping out of the profession. *Teaching and teacher Education*, 26(8), 1530-1543.
- Hong, K. S. (2002). Relationships between students' and instructional variables with satisfaction and learning from a Web-based course. *The Internet and Higher Education*, 5(3), 267-281.
- Hord, S. M. & Tobia, E. F. (2012). *Reclaiming Our Teaching Profession*. Teachers College Press.
- Hornsey, M. J. (2008). Social identity theory and self-categorization theory: A historical review. *Social and Personality Psychology Compass*, 2(1), 204-222.
- Hoyle, E. (1974). Professionality, professionalism and control in teaching. *London Educational Review*, *3*(2), 13-19.
- Hoyle, E. (1969). The role of the teacher. Routledge & Kegan Paul Books.
- Hudson, Judith N; Lethbridge, Alistair; Vella, Susan; Caputi, Peter (2016). Decline in medical students attitude into intraprofessinal. Medical Education. Vol. 50 Issue 5, 550-559.
- Huy, P. P. (2008). Predicting change in epistemological belief, reflecting thinking and learning styles: A longitudinal study. *British Journal of Educational Psychology*, 78, 75 93.
- Hwang, D., & Embi, M. A. (2007). Approaches employed by secondary school teachers to teaching the literature component in English. *Malaysian Journal of Educators and Education*, 22, 1-23.
- Hyland, N. E. (2006). De-tracking in the social studies: A path to a more democratic education?. *Theory into Practice*, 45(1), 64-71.

- Idris, N., Loh, S. C., Nor, N. M., Razak, A. Z. A., &Saad, R. M. (2007). The professional preparation of Malaysian teachers in the implementation of teaching and learning of mathematics and science in English. *Eurasia Journal of Mathematics, Science & Technology Education*, 3(2), 101-110.
- Ingersoll, R. M. (2003). The teacher shortage: Myth or reality?. *Educational Horizons*, 81(3), 146-152.
- Ingvarson, L., Schwille, J., Tatto, M. T., Rowley, G., Peck, R., & Senk, S. L. (2013). *An Analysis of Teacher Education Context, Structure, and Quality-Assurance Arrangements in TEDS-M Countries: Findings from the IEA Teacher Education and Development Study in Mathematics (TEDS-M)*. International Association for the Evaluation of Educational Achievement. Herengracht 487, Amsterdam, 1017 BT, The Netherlands.
- Izuma, K. (2013). The neural basis of social influence and attitude change. *Current opinion in neurobiology*, 23(3), 456-462.
- Jackling, B. & de Lange, P. (2009). Do accounting graduate' skills meet the expectations of employers? A matter of convergence and divergence. *Accounting Education: An International Journal*, 18(4-5), 369 385.
- Jackson, D. (2016). Re-conceptualising graduate employability: the importance of preprofessional identity. *Higher Education Research & Development*, 35(5), 925-939.
- Jackson, D., & Temperley, J. (2007). From professional learning community to networked learning community. Professional learning communities: Divergence, depth and dilemmas, 45-62.
- Jackson, L. A., Sullivan, L. A., Harnish, R., & Hodge, C. N. (1996). Achieving positive social identity: Social mobility, social creativity, and permeability of group boundaries. *Journal of Personality and Social Psychology*, 70(2), 241.

- Jackson, W. (2003). Methods: Doing Social Research. 3rd edition. Canada: Prentice Hall.
- Janesick, V. J. (1998). The dance of qualitative research design: metaphor, methodology and meaning. In N.K Denzin& Y. S. Lincoln (Eds). Strategies of Qualitative Inquiry. London: Sage Publications.
- Johansson, K., Segerstad, H. H., Hult, H., & Dahlgren, M. A. (2008). The two faces of political science studies junior and senior students' thoughts about their education and their future profession. *Higher Education*, 55, 623 636.
- Joireman, J., Kamdar, D., Daniels, D. & Duell, B. (2006). Good citizens to the end? It depends; Empathy and concern with future consequences moderate the impact of the short-term time horizon on organizational citizenship behaviours. *Journal of Applied Psychology*, 91(6), 1307-1320.
- Jones, G., & Ramchand, D. (2013). Education and human capital development in the giants of Asia. *Asian Pacific Economic Literature*, 27(1), 40-61.
- Jovchelovitch, S. & Bauer, M. W. (2000) Narrative interviewing.InBauer, M. W., & Gaskell, G. (Eds.). (2000). *Qualitative researching with text, image and sound: A practical handbook for social research*. London: Sage.
- Jugović, I., Marušić, I., PavinIvanec, T., &VizekVidović, V. (2012). Motivation and personality of preservice teachers in Croatia. *Asia-Pacific Journal of Teacher Education*, 40(3), 271-287.
- Kahveci, A. (2009). Exploring chemistry teacher candidates' profile characteristics, teaching attitudes and beliefs, and chemistry conceptions. *Chemistry Education Research and Practice*, 10, 109-120.
- Karpudewan, M., Ismail, Z. H., & Mohamed, N. (2009). The integration of green chemistry experiments with sustainable development concepts in pre-service

- teachers' curriculum: Experiences from Malaysia. *International Journal of Sustainability in Higher Education*, 10(2), 118-135.
- Kasim, T., Aini, T. S. (2012). Teaching and learning experiences in Malaysian higher education: a case study of a teacher education programme (Doctoral dissertation, AUT University).
- Kasim, T., Aini, T. S., & Furbish, D. (2010). Encouraging lifelong learning through student-centred learning approaches in a Malaysian teacher education programme. www.atea.edu.au
- Keat, O. Y., Selvarajah, C., & Meyer, D. (2011). Inclination towards entrepreneurship among university students: An empirical study of Malaysian university students. *International Journal of Business and Social Science*, 2(4), 206-220.
- Kelman, H. C. (1958). Compliance, identification, and internalization: Three processes of attitude change. *Journal of conflict resolution*, 51-60.
- Kerchner, C. T., & Caufman, K. D. (1995). Lurching toward professionalism: The saga of teacher unionism. *The Elementary School Journal*, *96*(1), 107-122.
- Kim, Y. M., & Hong, S. (2019). Intersectionality of cultural identity among immigrants in South Korea from a multidimensional acculturation perspective. *Journal of Ethnic & Cultural Diversity in Social Work*, 1-15.
- Klein, M. (2016). Educational expansion, occupational closure and the relation between educational attainment and prestige over time. *Sociology*, *50*(1), 3-23.
- Klegon, D. (1978). The sociology of professions: an emerging perspective. *Sociology of work and occupations*, *5*(3), 259-283.
- Klenowski, V., Askew, S., & Carnell, E. (2006). Portfolios for learning, assessment and professional development in higher education. *Assessment & Evaluation in Higher Education*, 31(3), 267-286.

- Knoblauch, D., & Chase, M. A. (2015). Rural, suburban, and urban schools: The impact of school setting on the efficacy beliefs and attributions of student teachers. *Teaching and Teacher Education*, 45, 104-114.
- Knowles, M. (1984). A Theory Of Adult Learning: Andragogy. *Agriculture and Rural Development* 202.
- Kolb, D. A. (1984). Experiential learning: Experience as the source of learning and development (Vol. 1). Englewood Cliffs, NJ: Prentice-Hall.
- Kontoghiorghes, C. (2016). Linking high performance organizational culture and talent management: satisfaction/motivation and organizational commitment as mediators. *The International Journal of Human Resource Management*, 27(16), 1833-1853.
- Krech, D., & Crutchfield, R. S. (1948). The measurement of beliefs and attitudes. *Theory and Problems of Social Psychology*, p. 205 272.
- Krejsler, J. (2005). Professions and their Identities: How to explore professional development among (semi-) professions. *Scandinavian Journal of Educational Research*, 49(4), 335-357.
- Kyratsis, Y., Atun, R., Phillips, N., Tracey, P., George, G. 2016. Health systems in transition: identity work in the context of shifting institutional logics. *Academy of Management Journal*, in press
- Kyriacou, C., & Kunc, R. (2007). Beginning teachers' expectations of teaching. *Teaching and Teacher Education*, 23(8), 1246-1257.
- Lamote, C., & Engels, N. (2010). The development of student teachers' professional identity. *European Journal of Teacher Education*, *33*(1), 3-18.

- Lasky, S. (2005). A sociocultural approach to understanding teacher identity, agency and professional vulnerability in a context of secondary school reform. *Teaching and teacher education*, 21(8), 899-916.
- Lee, M. N. (1996). Unity in diversity: Teacher education, in multicultural Malaysia. *Teacher education in plural societies: An international review*, 72-81.
- Levine-Rasky, C. (1998). Preservice teacher education and the negotiation of social difference. *British Journal of Sociology of Education*, 19(1), 89 112.
- Liau, A. K., Liau, A. W., Teoh, G. B., & Liau, M. T. (2003). The Case for Emotional Literacy: the influence of emotional intelligence on problem behaviours in Malaysian secondary school students. *Journal of Moral Education*, 32(1), 51-66.
- Liaw, S. Y., Siau, C., Zhou, W. T., & Lau, T. C. (2014). Interprofessional simulation-based education program: a promising approach for changing stereotypes and improving attitudes toward nurse–physician collaboration. *Applied Nursing Research*, 27(4), 258-260.
- Lieberman, J. (2009). Reinventing teacher professional norms and identities: The role of lesson study and learning communities. *Professional development in education*, 35(1), 83-99.
- Ligadu, C. P. (2012). The impact of the professional learning and psychological mentoring support for teacher trainees. *Journal of Social Sciences*, 8(3), 350.
- Lincoln, Y. S., & Guba, E. G. (1985). *Naturalistic inquiry*. London: Sage.
- Lincoln, Y. S., & Denzin, N. K. (Eds.). (2003). Turning points in qualitative research: Tying knots in a handkerchief. Rowman Altamira.
- Liu, D. (1998). Ethnocentrism in TESOL: Teacher education neglected needs of international TESOL students. *ELT Journal*, 52(1), 3 10.

- Liu, Y., Green, A., & Pensiero, N. (2016). Expansion of higher education and inequality of opportunities: a cross-national analysis. *Journal of Higher Education Policy and Management*, 38(3), 242-263.
- Luehmann, A. L. (2007). Identity development as a lens to science teacher preparation. *Science Education*, *91*(5), 822-839.
- Lui, M. H., Lam, L. W., Lee, I. F., Chien, W. T., Chau, J. P., &Ip, W. Y. (2008). Professional nursing values among baccalaureate nursing students in Hong Kong. *Nurse Education Today*, 28(1), 108-114.
- Lunenberg, M., Korthagen, F., &Swennen, A. (2007). The teacher educator as a role model. *Teaching and teacher education*, 23(5), 586-601.
- Mackie, D. M., & Smith, E. R. (2018). Intergroup emotions theory: Production, regulation, and modification of group-based emotions. In *Advances in Experimental Social Psychology* (Vol. 58, pp. 1-69). Academic Press.
- Malaysia Examination Council (n.d). Retrieved from http://portal.mpm.edu.my/medsi1
- Malaysia Labour Force Statistics, 2015. Retrieved from www.statictics.gov.my
- Malaysia Prime Minister Department (n.d). Retrieved from www.jpm.gov.my
- Malik, M. A., Zakaria, A. R., Mohamed, M. F., & Sulaiman, A. M. (2015). School Survival for Novice Teacher in Malaysia. *Advanced Science, Engineering and Medicine*, 7(7), 573-579.
- Malm, B. (2009). Towards a new professionalism: enhancing personal and professional development in teacher education. *Journal of education for teaching*, 35(1), 77-91.

- Mantrak, M. (1995). Kearah meningkatkan martabat pendidikan abad ke-21: Cabaran, hakikat dan harapan. *Prosiding Seminar JKLJAU K-9*.
- Mat Kasim, M., Ramli, R., Ibrahim, H., Ghazali, M. I., Mohmad Kamal, F., & Vikneswari, S. (2012). Prioritizing of criteria in teacher-candidate selection process by analytic hierarchy process. *International Journal of Management Studies* (*IJMS*), 20(2), 15-28.
- Maxwell-Stuart, R., & Huisman, J. (2018). An exploratory study of student engagement at transnational education initiatives: Proactive or apathetic?. *International Journal of Educational Management*, 32(2), 298-309.
- McPherson, D. Exploring Beliefs in Teacher Education. Retrieved online http://conference.crpp.nie.edu.sg/2007/paper/papers/TPD470.pdf
- Mead, G. H. (1934). *Mind, self and society* (Vol. 111). University of Chicago Press.: Chicago.
- Meca, A., Ritchie, R. A., Beyers, W., Schwartz, S. J., Picariello, S., Zamboanga, B. L., & Crocetti, E. (2015). Identity centrality and psychosocial functioning: A personcentered approach. *Emerging Adulthood*, *3*(5), 327-339.
- Mehra, A., Kilduff, M., & Brass, D. J. (1998). At the margins: A distinctiveness approach to the social identity and social networks of underrepresented groups. *Academy of Management Journal*, 41(4), 441-452.
- Merton, R. K. (1957). Social theory and social structure. Collier-Macmillan Limited: London
- Meyer, J., & Allen, N. (1991). A three-component conceptualization of organizational commitment. *Human Resource Management Review*, *1*(1), 61-89.
- Miles, M. B. &Huberman, A. M. (1994). *Qualitative data analysis*, 2nd edition. California: SAGE Publications.

- Ministry of Education Malaysia (n.d.- a). National Education Philosophy. Retrieved from www.moe.gov.my
- Ministry of Education Malaysia (n.d.-b). Higher education statistics. Retrieved from www.moe.gov.my/inclusive
- Ministry of Education Malaysia (n.d.-c). Higher education statistics. Retrieved from www.moe.gov.my/data
- Ministry of Education Malaysia (2012). Higher education document. Retrieved from www.moe.gov.my/standardgurumalaysia
- Ministry of Higher Education Malaysia (2006). The Report by the Committee to Study, Review and Make Recommendations Concerning the Development and Direction of Higher Education in Malaysia (2006). Putrajaya: Ministry of Education.
- Ministry of Higher Education Malaysia (n.d.-a). Higher education statistics. Retrieved from http://mohe.gov.my/en/download/awam/statistik
- Ministry of Higher Education Malaysia (n.d.-b). Malaysia Higher Education Blueprint 2015 2025. Retrieved from http://mohe.gov.my/muat-turun/awam/penerbitan/pppm-2015-2025-pt/5-malaysia-education-blueprint-2015-2025-higher-education
- Minnis, J. R. (1999). Is reflective practice compatible with Malay-Islamic values? Some thoughts on teacher education in Brunei Darussalam. *Australian Journal of Education*, 43,172 185.
- Mokshein, S. E., Ahmad, H. H., & Vongalis-Macrow, A. (2009). Secondary Teacher Policy Research in Asia: Towards providing quality secondary education: Training and retaining quality teachers in Malaysia.

- Mogra, I. (2010). Teachers and teaching: A contemporary Muslim understanding. Religious Education: The Official Journal of the Religious Education Association, 105(3), 317 – 329.
- Mohamad Dahan, H.& Ghazali, S. S. (2007). How does it feel to be a teacher? Personal journeys and lessons for a Malaysian university. *Asian Journal of University Education*, 3(1), 63 80
- Mohd Ilias, Siti Zafirah (2008). Faktor-Faktor yang Mempengaruhi Pemilihan Profession Perguruan di Kalangan Pelajar Tahun 4
- Mokshein, S. E., Ahmad, H. H., Vongalis-Macrow, A., & Asia, U. N. E. S. C. O. (2009). Towards providing quality secondary education: training and retaining quality teachers in Malaysia. UNESCO Bangkok Asia and Pacific Regional Bureau for Education.
- Monteiro, A. R. (2015). *The teaching profession: Present and future*. Springer International Publishing
- Morgan, D. L. (1997). *Focus groups as qualitative research*. California: Sage Publication Inc.
- Morris, M. W., & Leung, K. (2010). Creativity east and west: Perspectives and parallels. *Management and Organization Review*, 6(3), 313-327.
- Mortimer, J. T., & Simmons, R. G. (1978). Adult socialization. *Annual review of sociology*, 4, 421-454.
- Mylrea, M., Sen Gupta, T., & Glass, B. (2017). Developing professional identity in undergraduate pharmacy students: A role for self-determination theory. *Pharmacy*, 5(2), 16.

- Nason, R. S., Bacq, S., & Gras, D. (2018). A behavioral theory of social performance: Social identity and stakeholder expectations. *Academy of Management Review*, 43(2), 259-283.
- Nelson, D. L., & Quick, J. C. (2006). Organizational Behavior: Foundations. *Realities and Challenges*, 5.
- Neve, H., Lloyd, H., & Collett, T. (2017). Understanding students' experiences of professionalism learning: a 'threshold' approach. *Teaching in Higher Education*, 22(1), 92-108.
- Niemi, P. M. (1997). Medical students' professional identity: self-reflection during the preclinical years. *Medical Education*, 31, 408 415.
- Nilssen, V., & Solheim, R. (2015). 'I see what I see from the theory I have read'. Student teachers learning through theory in practice. *Journal of Education for Teaching*, 41(4), 404-416.
- Noah, S. M., Karim, A. M. A., & Othman, J. (2012). The Construction of the Malaysian Educators Selection Inventory (MEdSI) for the Selection of Bachelor of Education Students in Public University in Malaysia. *Pedagogi*, 9(1), 48-57.
- Oakes, P. J., Haslam, S. A. and Turner, J. C. (1994) *Stereotyping and social reality*. Oxford: Blackwell.
- O'Brien, K. M. (2001). The legacy of Parsons: Career counselors and vocational psychologists as agents of social change. *The Career Development Quarterly*, 50(1), 66-76.
- O'Connor, K. E. (2008). "You choose to care": Teachers, emotions and professional identity. *Teaching and teacher education*, 24(1), 117-126.

- Ormston, R., Spencer, L. Bernard, M. & Snape, D. (2013). The foundation of qualitative research. In Ritchie, J. et al. (Eds.). *Qualitative research practice: A guide for social science students and researchers*. 2nd edition. London: Sage Publications.
- Osunde, A. U., & Izevbigie, T. I. (2006). An Assessment of Teachers' Attitude Towards Teaching Profession In Midwestern Nigeria. *Education*, *126*(3), 462.
- Owens, T. J., Robinson, D. T., & Smith-Lovin, L. (2010). Three faces of identity. *Annual Review of Sociology*, 36.
- Paterson, M., Higgs, J., Wilcox, S., & Villeneuve, M. (2002). Clinical reasoning and self-directed learning: Key dimensions in professional education and professional socialisation. *Focus on Health Professional Education*, 4(2), 5-21.
- Patton, M. Q. (2002). Qualitative Research and Evaluation Methods. 3rd Edition. CA: Sage Publications.
- Pauline, S. C. G. (2013). Conception of competency: A phenomenographic investigation of beginning teachers in Malaysia. *The Qualitative Report*, 18(40), 1-16.
- Petty, R. E., Tormala, Z. L., Brinol, P., & Jarvis, W. B. G. (2006). Implicit ambivalence from attitude change: an exploration of the PAST model. *Journal of personality and social psychology*, 90(1), 21.
- Piaget, J. (1972). Development and learning. Reading in child behavior and development. New York: Hartcourt Brace Janovich.
- Pike, G. R. (2006). Students' personality types, intended majors, and college expectations: Further evidence concerning psychological and sociological interpretations of Holland's theory. *Research in Higher Education*, 47(7), 801-822.

- Platow, M. J., Mavor, K. I., & Grace, D. M. (2013). On the role of discipline-related self-concept in deep and surface approaches to learning among university students. *Instructional Science*, 41(2), 271-285.
- Radcliffe, C. & Lester, H. (2003). Perceived stress during undergraduate medical training: A qualitative study. Medical Education, 37, 32 38.
- Rashotte, L. (2007). Social influence. *The Blackwell encyclopedia of social psychology*, 9, 562-563.
- Rassool, G. H. (2000). The crescent and Islam: healing, nursing and spiritual dimension. Some considerations towards an understanding of the Islamic perspectives on caring. *Journal of Advanced Nursing*, 32(6), 1476 1484.
- Rather, R. A. (2018). Investigating the impact of customer brand identification on hospitality brand loyalty: A social identity perspective. *Journal of Hospitality Marketing & Management*, 27(5), 487-513.
- Reddy, J. P., Tan, S. M. K., Azmi, M. T., Shaharom, M. H., Rosdinom, R., Maniam, T., & Minas, I. H. (2005). The effect of a clinical posting in psychiatry on the attitudes of medical students towards psychiatry and mental illness in a Malaysian medical school. *Annals-Academy Of Medicine Singapore*, *34*(8), 505.
- Reid, A., Dahlgren, L. O., Petocz, P. & Dahlgren, M. A. (2008). Identity and engagement for professional formation. *Studies of Higher Education*. 33(6), 729 742.
- Reilly, A., & Spratt, C. (2007). The perceptions of undergraduate student nurses of high-fidelity simulation-based learning: A case report from the University of Tasmania. *Nurse Education Today*, 27(6), 542-550.
- Rice, G. (1999). Islamic ethics and the implications for business. *Journal of business ethics*, 18(4), 345-358.

- Richardson, P. M. (2004). Possible influences of Arabic-Islamic culture on the reflective practices proposed for an education degree in the Higher College of Technology in the United Arab Emirates. *International Journal of Educational Development*, 24, 429 436.
- Richardson, S. (2009). Undergraduates' perceptions of tourism and hospitality as a career choice. *International Journal of Hospitality Management*, 28(3), 382-388.
- Robbins, S. P., Judge, T. A., & Vohra, N. (2013). Organizational behavior. Pearson.
- Rogers, L. O., Scott, M. A., & Way, N. (2015). Racial and gender identity among Black adolescent males: An intersectionality perspective. *Child Development*, 86(2), 407-424.
- Roselina Shakir (2009). Soft skills at the Malaysian institute of higher learning. *Asia Pacific Education Review*, 10, 309 315.
- Rots, I., Aelterman, A., Vlerick, P., & Vermeulen, K. (2007). Teacher education, graduates' teaching commitment and entrance into the teaching profession. *Teaching and teacher education*, 23(5), 543-556.
- Rots, I., Aelterman, A., &Devos, G. (2014). Teacher education graduates' choice (not) to enter the teaching profession: does teacher education matter?. *European Journal of Teacher Education*, 37(3), 279-294.
- Rudman, L. A. (2004). Social justice in our minds, homes, and society: The nature, causes, and consequences of implicit bias. *Social Justice Research*, 17(2), 129-142.
- Sachs, J. (2003). The activist teaching profession. Open University Press
- Saks, M. (2010). Analyzing the professions: The case for the neo-Weberian approach. *Comparative Sociology*, 9(6), 887-915.

- Sarantakos, S. (2005). Social research. NY: Palgrave Macmillan.
- S. Bolster, Jr, A. (1983). Toward a more effective model of research on teaching. *Harvard Educational Review*, *53*(3), 294-308.
- Schepens, A, Alterman, A. & Van Keer, H. (2007). Studying learning processes of student teachers with stimulated recall interviews through changes in iterative cognitions. *Teaching and Teacher Education*, 23 (4), 457 472.
- School Division, Ministry of Education Malaysia. (n.d). Retrieved at www.moe.gov.my/bahagian sekolah
- Schwandt, T. A. (2003). Three epistemological stances for qualitative inquiry: Interpretivism, hermeneutics, and social constructionism. In N. K. Denzin& Y. Lincoln (Eds.), *The landscape of qualitative research: Theories and issues* (pp. 292-331). Thousand Oaks, CA: Sage Publications Inc.
- Schwarz, N. (2007). Attitude construction: Evaluation in context. *Social Cognition*, 25(5), 638-656.
- Seker, H., Deniz, S., & Gorgen, I. (2015). Pre-service teachers motivations toward teaching profession and their opinions about the pedagogic formation program. *Educational Research and Reviews*, 10(10), 1403-1414.
- Shanmugam, A. (1999). Information seeking behaviour of trainee teachers in selected teacher training colleges in Malaysia. *Malaysian Journal of Library & Information Science*, 4(1), 1-26.
- Shepherd, D. A., Patzelt, H., & Wolfe, M. (2011). Moving forward from project failure: Negative emotions, affective commitment, and learning from the experience. *Academy of Management Journal*, *54*(6), 1229-1259
- Siebert, D. C., & Siebert, C. F. (2005). The caregiver role identity scale: A validation study. *Research on Social Work Practice*, *15*(3), 204-212.

- Silverman, D. (Ed.). (2009). *Qualitative research*.3rd edition. London: Sage.
- Simmonds, A. H., & Dicks, A. P. (2018). Mentoring and professional identity formation for teaching stream faculty: A case study of a university Peer-to-Peer mentorship program. *International Journal of Mentoring and Coaching in Education*, 7(4), 282-295.
- Sinclair, C. (2008). Initial and changing student teacher motivation and commitment to teaching. *AsiaPasific Journal of Teacher Education*, 36(2), 79 104.
- Sinclair, C., Dowson, M., & McInerney, D. M. (2006). Motivations to teach: Psychometric perspectives across the first semester of teacher education. *Teachers College Record*, 108(6), 1132.
- Sirin, S. R. (2005). Socioeconomic status and academic achievement: A meta-analytic review of research. *Review of educational research*, 75(3), 417-453.
- Skaalvik, E. M., & Skaalvik, S. (2011). Teacher job satisfaction and motivation to leave the teaching profession: Relations with school context, feeling of belonging, and emotional exhaustion. *Teaching and Teacher Education*, 27(6), 1029-1038.
- Sleap, M. & Reed, H. (2006). View of sport science graduates regarding work skills developed at university. *Teaching in Higher Education*, 11(1), 47 61.
- Slocum, W. L. (1959). Some sociological aspects of occupational choice. *American Journal of Economics and Sociology*, 139-147.
- Smith, J. R., & Hogg, M. A. (2008). Social identity and attitudes. *Attitudes and attitude change*, 337-360.
- Smyth, L., Mavor, K. I., & Platow, M. J. (2018). Social identification and academic performance: integrating two existing models of tertiary student learning. *Educational Psychology*, 1-17.

- Smits, I., Soenens, B., Luyckx, K., Duriez, B., Berzonsky, M., & Goossens, L. (2008). Perceived parenting dimensions and identity styles: Exploring the socialization of adolescents' processing of identity-relevant information. *Journal of Adolescence*, 31(2), 151-164.
- Sollami, A., Caricati, L., & Mancini, T. (2015). Ambivalent stereotypes of nurses and physicians: impact on students' attitude toward interprofessional education. *Acta Bio Medica Atenei Parmensis*, 86(1S), 19-28.
- Strauss, A. & Corbin, J. (1990). *Basic of qualitative research: grounded theory procedures and techniques*, Newbury Park, CA: Sage Publications, Inc.
- Struyve, C., Daly, A., Vandecandelaere, M., Meredith, C., Hannes, K., & De Fraine, B. (2016). More than a mentor: The role of social connectedness in early career and experienced teachers' intention to leave. *Journal of professional capital and community*, *1*(3), 198-218.
- Struyven, K., Jacobs, K., &Dochy, F. (2013). Why do they want to teach? The multiple reasons of different groups of students for undertaking teacher education. *European journal of psychology of education*, 28(3), 1007-1022.
- Sufean Husein, (2004). *Pendidikan di Malaysia: Sejarah, sistem dan falsafah*, Kuala Lumpur: Dewan Bahasa dan Pustaka.
- Su, Y., & Chung, Y. H. (2015). Understanding the emotional reactions and exploring the professional development of college students based on reflections. *Teaching in Higher Education*, 20(3), 285-299.
- Taber, B. J., Hartung, P. J., & Borges, N. J. (2011). Personality and values as predictors of medical specialty choice. *Journal of Vocational Behavior*, 78(2), 202-209.

- Tajfel, H. & Turner, J. C. (1986). The social identity theory of intergroup behavior. S. Worchel & WG Austin (ed.), The Psychology of Intergroup Relations, 7-24.
- Tajfel, H. (1974). Social identity and intergroup behaviour. Social Science Information, 13, 65 93
- Tajfel, H. (1978). Social categorization, social identity and social comparison, In *Differentiation between social groups*, 61 76.
- Tajfel, H., & Turner, J. (1979). An integrative theory of intergroup conflict. In W. G. Austin & S. Worchel (Eds.), *The social psychology of intergroup relations* (pp. 33-48). Monterey, CA: Brooks/Cole.
- Tan, S., & Rey, J. (2005). Depression in the young, parental depression and parenting stress. *Australasian Psychiatry*, *13*(1), 76-79.
- Teng, C. (2008). The effect of personality traits and attitudes on student uptake in hospitality employment. *International Journal of Hospitality Management*, 27, 76 86.
- Teo, T., Lee, C. B., Chai, C. S., & Wong, S. L. (2009). Assessing the intention to use technology among pre-service teachers in Singapore and Malaysia: A multigroup invariance analysis of the Technology Acceptance Model (TAM). *Computers & Education*, 53(3), 1000-1009.
- ten Dam, G. T., & Blom, S. (2006). Learning through participation. The potential of school-based teacher education for developing a professional identity. *Teaching and teacher education*, 22(6), 647-660.
- Thurstone, L. L. (1931). The measurement of social attitudes. *The Journal of Abnormal and Social Psychology*, 26(3), 249.

- Tok, Ş. (2011). Effects of the know-want-learn strategy on students' mathematics achievement, anxiety and metacognitive skills. *Metacognition and Learning*, 1-20.
- Tomlinson, M. (2008). 'The degree is not enough': students' perceptions of the role of higher education credentials for graduate work and employability. *British journal of sociology of education*, 29(1), 49-61.
- Tomlinson, M. (2007). Graduate employability and students attitudes and orientations to the labour market. *Journal of Education and Work*, 20(4), 285 304.
- Toren, N. (1975). Deprofessionalization and its sources: a preliminary analysis. *Sociology of work and occupations*, 2(4), 323-337.
- Trede, F., Macklin, R., & Bridges, D. (2012). Professional identity development: a review of the higher education literature. *Studies in Higher Education*, *37*(3), 365-384.
- Tseng, H. C., Wang, H. H., & Weng, W. C. (2013). Nursing students' perceptions toward the nursing profession from clinical practicum in a baccalaureate nursing program—A qualitative study. *The Kaohsiung journal of medical sciences*, 29(3), 161-168.
- Tschannen-Moran, M., & Hoy, A. W. (2001). Teacher efficacy: Capturing an elusive construct. *Teaching and teacher education*, *17*(7), 783-805.
- Turner, J. C. (1991). Social influence. Thomson Brooks/Cole Publishing Co.
- Turner, J. C. (1975). Social comparison and social identity: Some prospects for intergroup behaviour. *European journal of social psychology*, *5*(1), 1-34.
- Turner, J.C. (1985). 'Social categorization and the self concept: a social cognitive theory of group behaviour', in E.J. Lawler (ed.), *Advances in Group Processes*, 2, Greenwich, CT: JAI Press.

- Turner, J. C. (2010). Towards a cognitive redefinition of the social group. In Postmes,T. & Branscombe, N. R. (ed): *Rediscovering Social Identity: Key Reading*. UK: Psychology Press.
- Turner, J. C., Hogg, M. A., Oakes, P. J., Reicher, S. D., &Wetherell, M. S. (1987). *Rediscovering the social group: A self-categorization theory*. Basil Blackwell.
- UNESCO Institute of Statistics (n.d). Retrieved from www. uis.unesco.org
- Universiti Pendidikan Sultan Idris (n.d). Retrieved from www. upsi.edu.my
- Van Maanen, J. and Barley, S.R. (1984). Occupational communities: Culture and control in organizations. *Research in Organizational Behavior*, 6, 287-365.
- Van Scotter, R. D., Hass, J. D., Kraft, R. J., and Schott, J. C. (1985). *Social foundations of education*. New York: Prentice- Hall, Inc.
- Vaughn, S., Schumm, J. S., & Sinagub, J. M. (1996). Focus group interviews in education and psychology. Sage.
- Virta, A. (2002). Becoming a history teacher: observations on the beliefs and growth of student teachers, *Teaching and Teacher Education*, 18, 687 698.
- Verkuyl, H. J. (1998). Stereotyping, prototyping and figurative use: towards a proper semantic analysis. Utrecht University. Utrecht Institute of Linguistics OTS.
- Walling, B., & Lewis, M. (2000). Development of professional identity among professional development school preservice teachers: Longitudinal and comparative analysis. *Action in Teacher Education*, 22(2A), 65-72.

- Wang, M., Shen, R., Novak, D., & Pan, X. (2009). The impact of mobile learning on students' learning behaviours and performance: Report from a large blended classroom. *British Journal of Educational Technology*, 40(4), 673-695.
- Wellington, (2012). A Study to Investigate reasons for early change in career to teaching and challenges experienced in a one-year post graduate music education programme at a teachers' training institution in Kuala Lumpur- 4th Music Education Conference (Seminar Pendidikan Muzik Kebangsaan) MusEd'12 di Universiti Pendidikan Sultan Idris, Tanjong Malim 16-17 February 2012
- Whitehead, K. (2007). Addressing social difference with prospective teachers who want "to make a difference". *Asia- Pacific Journal of Teacher Education*, 35(4), 367-385.
- Yuval-Davis, N. (2006). Intersectionality and feminist politics. *European Journal of Women's Studies*, 13(3), 193-209.
- Zamri bin Khairani, A. (2011). Student Teachers' Expectations of Teaching as a Profession in Malaysia. *Online Submission*.
- Zhu, J., Tatachari, S., & Chattopadhyay, P. (2017). Newcomer Identification: Trends, Antecedents, Moderators, and Consequences. *Academy of Management Journal*, 60(3), 855-879.
- Zevenbergen, A. A., Whitehurst, G. J., & Zevenbergen, J. A. (2003). Effects of a shared-reading intervention on the inclusion of evaluative devices in narratives of children from low-income families. *Journal of Applied Developmental Psychology*, 24(1), 1-15.
- Zubir, R. (1988). Descriptions of teaching and learning: A Malaysian perspective. *Studies in Higher Education*, *13*(2), 139-149.